BUBBLE TREE CONVERGENCE OF CONFORMALLY CROSS PRODUCT PRESERVING MAPS*

DA RONG CHENG[†], SPIRO KARIGIANNIS[‡], AND JESSE MADNICK[§]

Abstract. We study a class of weakly conformal 3-harmonic maps, called associative Smith maps, from 3-manifolds into 7-manifolds that parametrize associative 3-folds in Riemannian 7-manifolds equipped with G_2 -structures. Associative Smith maps are solutions of a conformally invariant nonlinear first order PDE system, called the Smith equation, that may be viewed as a G_2 -analogue of the Cauchy–Riemann system for J-holomorphic curves.

In this paper, we show that associative Smith maps enjoy many of the same analytic properties as J-holomorphic curves in symplectic geometry. In particular, we prove: (i) an interior regularity theorem, (ii) a removable singularity result, (iii) an energy gap result, and (iv) a mean-value inequality. While our approach is informed by the holomorphic curve case, a number of nontrivial extensions are involved, primarily due to the degeneracy of the Smith equation.

At the heart of above results is an ε -regularity theorem that gives quantitative $C^{1,\beta}$ -regularity of $W^{1,3}$ associative Smith maps under a smallness assumption on the 3-energy. The proof combines previous work on weakly 3-harmonic maps and the observation that the associative Smith equation demonstrates a certain "compensation phenomenon" that shows up in many other geometric PDEs.

Combining these analytical properties and the conformal invariance of the Smith equation, we explain how sequences of associative Smith maps with bounded 3-energy may be conformally rescaled to yield bubble trees of such maps. When the G₂-structure is closed, we prove that both the 3-energy and the homotopy are preserved in the bubble tree limit. This result may be regarded as an associative analogue of part of Gromov's Compactness Theorem in symplectic geometry.

Key words. calibrated submanifolds, associative submanifolds, Cayley submanifolds, vector cross products, *p*-harmonic maps, removable singularities, bubbling, bubble trees.

Mathematics Subject Classification. 53C15, 53C38, 53C43, 57R57, 58E20, 58J60.

1. Introduction.

1.1. Motivation. In symplectic geometry, moduli spaces of holomorphic curves play an important role [20, 35]. Starting with a symplectic manifold (M,ω) , one chooses an almost complex structure J_M that is compatible with ω in some sense. A map $u : (\Sigma, J_{\Sigma}) \to (M, \omega, J_M)$ from a Riemann surface (Σ, J_{Σ}) is then called a (J_{Σ}, J_M) -holomorphic map (or simply a holomorphic curve) if it solves the Cauchy–Riemann system

$$J_M \circ du = du \circ J_{\Sigma}. \tag{1}$$

Various moduli spaces of solutions to (1) can then be studied. A priori, these moduli spaces are almost never compact, largely due to their conformal invariance (see property (i) below); their compactifications are described by Gromov's Compactness Theorem. From the compactified moduli spaces, one can derive powerful invariants of the original symplectic manifold (M,ω) , independent of the original choice of J_M . Here we emphasize that in its full generality, Gromov's Compactness Theorem allows the complex structure J_{Σ} to vary in such way that the Riemann surface (Σ, J_{Σ}) degenerates. However, for the purposes of the present paper we are primarily interested in the case where J_{Σ} is fixed.

^{*}Received December 15, 2019; accepted for publication February 10, 2020.

[†]Department of Mathematics, University of Chicago (chengdr@uchicago.edu).

[‡]Department of Pure Mathematics, University of Waterloo (karigiannis@uwaterloo.ca).

[§]Department of Mathematics and Statistics, McMaster University (madnickj@mcmaster.ca).

Underpinning Gromov's Compactness Theorem in this case are a litany of crucial algebraic and analytic properties enjoyed by solutions of the Cauchy–Riemann system (1). To recall these, let $u \colon \Sigma^2 \to M^{2n}$ be a smooth map, and equip both (Σ, J_{Σ}) and (M, J_M, ω) with compatible Riemannian metrics. Define the 2-energy of u on a measurable set $A \subset \Sigma$ by

$$E(u;A) = \frac{1}{2} \int_A |du|^2 \operatorname{vol}_{\Sigma}$$

where $\operatorname{vol}_{\Sigma}$ is the volume form on Σ , and write $\overline{\partial}u := \frac{1}{2} (du + J_M \circ du \circ J_{\Sigma})$. Then we have

- (i) (Conformal Invariance.) If u satisfies (1) and $f: \Sigma \to \Sigma$ is a conformal diffeomorphism, then $u \circ f$ satisfies (1).
- (ii) (Weak Conformality.) Every solution of (1) is weakly conformal.
- (iii) (Calibrated Image.) If u is an immersion satisfying (1), then $u(\Sigma) \subset M$ is calibrated by ω .
- (iv) (Energy Identity.) Every smooth map $u: \Sigma \to M$ satisfies $\frac{1}{2}|du|^2 \operatorname{vol}_{\Sigma} = |\overline{\partial} u|^2 \operatorname{vol}_{\Sigma} + u^*\omega$. Therefore

$$E(u;\Sigma) = \int_{\Sigma} |\overline{\partial} u|^2 \operatorname{vol}_{\Sigma} + \int_{\Sigma} u^* \omega.$$

In particular, $E(u; \Sigma) \geq \int_{\Sigma} u^* \omega$, with equality if and only if $\overline{\partial} u = 0$. Thus, by Stokes's theorem, holomorphic curves minimize 2-energy in their homology class. Moreover, if $\partial \Sigma = \emptyset$, then the minimum 2-energy attained is $\int_{\Sigma} u^* \omega = [\omega] \cdot u_*[\Sigma]$, which depends only on the topological data $[\omega] \in H^2(M; \mathbb{R})$ and $u_*[\Sigma] \in H_2(M; \mathbb{R})$.

Note that Properties (i)–(iv) may all be proved using linear algebra. Significantly less trivial are the following five analytic properties of holomorphic curves. In the sequel, we let $B(r) \subset \mathbb{R}^2$ denote the open ball of radius r > 0 centered at the origin.

(v) (Mean value inequality.) There exist constants $C, \varepsilon_0 > 0$ such that every holomorphic curve $u: B(2r) \to M$ with $E(u; B(2r)) < \varepsilon_0$ satisfies

$$\sup_{B(r)} |du|^2 \le \frac{C}{r^2} E(u; B(2r)).$$

- (vi) (Interior Regularity.) If $u \in W^{1,2}(\Sigma; M)$ satisfies (1) almost everywhere, then u is C^{∞} .
- (vii) (Removable Singularities.) If $u: B(1) \setminus \{0\} \to M$ is a smooth holomorphic curve with

$$E(u; B(1) \setminus \{0\}) < \infty$$
,

then u extends to a smooth holomorphic curve $u: B(1) \to M$.

- (viii) (Energy Gap.) There exists a constant $\varepsilon_0 > 0$ such that every holomorphic curve $u \colon S^2 \to M$ with $E(u; S^2) < \varepsilon_0$ is constant.
 - (ix) (Compactness Modulo Bubbling.) Let $u_n : (\Sigma, J_{\Sigma}) \to (M, J_M)$ be a sequence of holomorphic curves with $E(u_n; \Sigma) \leq C$. Then there exists a holomorphic curve $u_{\infty} : \Sigma \to M$ and a (possibly empty) finite set $\mathcal{S} = \{x_1, \dots, x_q\} \subset \Sigma$ such that, after passing to a subsequence of $\{u_n\}$:
 - (a) We have $u_n \to u_\infty$ in C^1_{loc} on $\Sigma \setminus \mathcal{S}$.

- (b) We have $|du_n|^2 d\mu \to |du_\infty|^2 d\mu + \sum_{i=1}^q m_i \delta(x_i)$ as Radon measures on Σ , where $\delta(x_i)$ is the Dirac delta measure at x_i . Also, each $m_i \ge \frac{1}{2}\varepsilon_0$ with ε_0 as in (viii).
- (c) If $||du_n||_{L^p} \leq C$ for some $p \in (2, \infty]$, then $S = \emptyset$.

Note that by (iv), any sequence of holomorphic curves representing the same homology class will satisfy a uniform 2-energy bound, and hence property (ix) may be applied to such a sequence. For a more detailed discussion of (i)–(ix), we refer the reader to [35, 40, 53].

In fact, several important conformally-invariant geometric PDE systems, such as the Yang-Mills equation on 4-manifolds [7, 13] and the harmonic map equation on surfaces [39], satisfy analogues of properties (i)–(ix) with respect to an appropriate conformally invariant energy functional. In each of these cases, properties (i)–(ix) can be used to construct "bubble trees" of such objects, which may be regarded as comprising the boundary of the corresponding compactified moduli space. In favorable cases, such as that of holomorphic curves [40] or harmonic maps from surfaces [39], the bubble tree enjoys two additional properties, which can be loosely stated as follows:

- (x) The 2-energy is preserved in the bubble tree limit.
- (xi) The bubble tree has no necks. Consequently, homotopy is preserved in the bubble tree limit.

Precise statements of (x) and (xi) may be found in [35, 40, 53].

In this paper, we demonstrate that analogues of Properties (i)–(xi) hold for a class of weakly conformal maps that parametrize associative 3-folds in Riemannian 7-manifolds equipped G_2 -structures. We are motivated in part by the larger project of "counting" associative submanifolds in order to obtain invariants of G_2 -manifolds, which we briefly discuss.

Recall that a G₂-structure on a 7-manifold M is a G₂-subbundle of the frame bundle of M. Equivalently, it is a choice of a 3-form $\varphi \in \Omega^3(M)$ with the property that the bilinear form

$$B_{\varphi} \in \Gamma(\operatorname{Sym}^2(T^*M) \otimes \Lambda^7(T^*M))$$

given by $B_{\varphi}(X,Y) = \iota_X \varphi \wedge \iota_Y \varphi \wedge \varphi$ is definite. Since $G_2 \leq SO(7)$, a G_2 -structure on M naturally induces a Riemannian metric h_{φ} on M, although the correspondence $\varphi \mapsto h_{\varphi}$ is not injective.

Two classes of submanifolds are of particular interest in G_2 geometry: the associative 3-folds, which are semi-calibrated by φ , and the coassociative 4-folds, which are semi-calibrated by $*\varphi$. It is often useful to regard associative 3-folds and coassociative 4-folds, respectively, as G_2 analogues of holomorphic curves and special Lagrangian submanifolds. One may also consider gauge-theoretic objects on (bundles over) 7-manifolds with G_2 -structures, such as G_2 -monopoles and G_2 -instantons, by analogy with familiar objects in 3 dimensions and 4 dimensions. For background on G_2 geometry, we refer the reader to [28].

In [9, §3], Donaldson–Thomas described an analogy between G_2 -instantons on 7-manifolds and flat connections on 3-manifolds, suggesting that G_2 -instantons may be used to define 7-dimensional analogues of the Casson invariant and Floer homology. Then, Tian [47] showed that sequences of G_2 -instantons may bubble along associative submanifolds, complicating the compactification of the moduli space. To account for this, Donaldson–Segal [8, §6.2] conjectured that a count of G_2 -instantons, appropriately weighted by a count of associative submanifolds, ought to yield a deformation invariant of G_2 -manifolds.

In [25] and [24], Haydys and Walpuski suggested that these weights might be described by considering generalized Seiberg–Witten monopoles on the associatives. Doan–Walpuski [6] explored this idea further, describing a Floer homology group generated by both associatives and ADHM Seiberg–Witten monopoles that could lead to a deformation invariant. A rather different perspective is offered by Joyce in [29], who puts forth a conjectural notion of G_2 quantum cohomology. (Some other earlier work on the analysis of counting associative submanifolds in a particular special case ("thin") was done by Leung–Wang–Zhu [32, 33].)

One of the main challenges confronting each of these proposed deformation invariants is that 1-parameter families of associative immersions may degenerate to a map of the form $u \circ \pi$, where $u \colon \Sigma^3 \to M^7$ is an associative immersion and $\pi \colon \widetilde{\Sigma}^3 \to \Sigma^3$ is a branched cover. Thus, we are led to seek a class of maps whose immersions parametrize associative submanifolds, but is sufficiently large to allow for nonempty critical loci.

Indeed, in [33, page 90, item (i)], Leung–Wang–Zhu point out that the conformality of the Cauchy–Riemann equation plays a key role for holomorphic curves, and suggest that an analogous conformally invariant PDE system for associative submanifolds would be desirable.

This is where the work of Smith [45] enters the picture. In his 2011 PhD thesis, Smith introduced a class of maps that generalize holomorphic maps of Riemann surfaces into almost Hermitian manifolds. Namely, a map $u: (\Sigma^n, g, P_{\Sigma}) \to (M^m, h, P_M)$ between Riemannian manifolds (Σ^n, g) and (M^m, h) endowed with compatible k-fold vector cross products $P_{\Sigma}: \Gamma(\Lambda^k T\Sigma) \to \Gamma(T\Sigma)$ and $P_M: \Gamma(\Lambda^k TM) \to \Gamma(TM)$ is said to be k-Smith (or Smith when k is clear from context) if du preserves the vector cross products up to a scale factor $\lambda = \lambda(du)$ depending only on du. That is,

$$P_M \circ \Lambda^k du = \lambda(du) \, du \circ P_{\Sigma}. \tag{2}$$

Equation (2) is henceforth referred to as the Smith equation. In brief, Smith maps are "conformally cross product preserving". Smith's original terminology [45] for such maps was "multiholomorphic", which we do not use.

Vector cross products are rather special geometric structures. Indeed, a k-fold vector cross product $P \colon \Gamma(\Lambda^k TX) \to \Gamma(TX)$ on a Riemannian manifold (X, g) induces a semi-calibration $\alpha \in \Omega^{k+1}(X)$ by raising an index:

$$\alpha(v_1, \dots, v_{k+1}) = g(P(v_1, \dots, v_k), v_{k+1}). \tag{3}$$

Moreover, by the work of Brown–Gray [3], k-fold vector cross products are classified into four families, and each is equivalent to particular reduction of the structure group of X. We summarize their classification in the following table:

k	$\dim(X)$	Equivalent Structure
k	k+1	Orientation
1	2r	Almost Hermitian Structure
2	7	G_2 -Structure
3	8	Spin(7)-Structure

In particular, 2-fold vector cross products can only occur in dimensions 3 and 7, while 3-fold vector cross products can only occur in dimensions 4 and 8.

Smith primarily considered (as do we) the case in which the domain Σ^n and target M^m carry (n-1)-fold vector cross products. In that case, it can be shown that

nonconstant solutions of (2) must satisfy $\lambda(du) = \frac{1}{(\sqrt{n})^{n-2}} |du|^{n-2}$, so that (2) reads

$$P_M \circ \Lambda^{n-1} du = \frac{1}{(\sqrt{n})^{n-2}} |du|^{n-2} du \circ P_{\Sigma}.$$

$$\tag{4}$$

In particular, when n = 2, equation (4) is the Cauchy–Riemann system (1) for holomorphic curves.

Note that (4) is a nonlinear first order PDE system involving m equations on m unknown functions of n variables. When n = 2, the system is elliptic, but when n > 2, it degenerates at points of the critical set

$$\operatorname{crit}_{u} = \{ x \in \Sigma \mid (du)_{x} = 0 \}.$$

Smith proved [45] that solutions of (4) satisfy the analogues of properties (i)–(iv). To state his result, let $\alpha \in \Omega^n(M)$ denote the semi-calibration induced by P_M as in (3), and denote the *n*-energy of a map $u \in W^{1,n}(\Sigma;M)$ on a measurable set $A \subset \Sigma$ by

$$E(u;A) = \frac{1}{(\sqrt{n})^n} \int_A |du|^n \operatorname{vol}_{\Sigma}.$$

THEOREM 1.1 (Smith [45]). Let $u: \Sigma^n \to M^m$ be a smooth map, where n < m.

- (i) If u satisfies (4) and $f: \Sigma \to \Sigma$ is a conformal diffeomorphism, then $u \circ f$ satisfies (4).
- (ii) Every solution of (4) is weakly conformal.
- (iii) If u is an immersion satisfying (4), then $u(\Sigma) \subset M$ is semi-calibrated by α .
- (iv) We have

$$E(u; \Sigma) \ge \int_{\Sigma} u^* \alpha.$$

Moreover, equality holds if and only if u satisfies (4). Thus, if $d\alpha = 0$, then Smith maps minimize n-energy in their homology class. If, in addition, $\partial \Sigma = \emptyset$, then the minimum n-energy attained can simply be expressed as $\int_{\Sigma} u^* \alpha = [\alpha] \cdot u_*[\Sigma]$, which depends only on topological data.

Our interest is in the exceptional cases (n, m) = (3, 7) and (n, m) = (4, 8). We shall refer to 2-Smith maps from a 3-manifold into a 7-manifold as associative Smith maps, and refer to 3-Smith maps from a 4-manifold into an 8-manifold as Cayley Smith maps. The corresponding cases of (4) are called the associative Smith equation and Cayley Smith equation, respectively.

In light of Smith's result, it is natural to ask whether analogues of (v)–(xi) for holomorphic curves also hold true for associative and Cayley Smith maps. **The purpose of the present paper is to establish all seven of these properties.** In this present paper, our main focus is on the associative Smith equation, although analogous results hold for the Cayley Smith equation.

REMARK 1.2. Smith's theorem indicates that solutions of (4) are attractive parametrizations of calibrated submanifolds. Fortunately, associative (respectively, Cayley) Smith maps are at least as abundant as associative (respectively, Cayley) submanifolds. Indeed, given an associative immersion $u \colon \Sigma^3 \to (M^7, h)$ into a 7-manifold M with a G₂-structure and a smooth function $F \colon \Sigma \to \mathbb{R}$, equipping the domain Σ with the Riemannian metric $g = e^F u^* h$ makes u into an associative Smith map.

1.2. Methods and main results.

Methods. We emphasize that many of the techniques used in establishing properties of holomorphic curves (n = 2) do not carry over mutatis mutandis to our situation (n = 3 or 4). This is primarily due to the following two reasons, which are themselves related:

- (i) The second order elliptic system obtained by differentiating (4) is of "n-harmonic type" and thus degenerate when $n \neq 2$, as opposed to being uniformly elliptic when n = 2. Consequently, the methods used in proving regularity and compactness results for holomorphic curves do not directly apply to our case.
- (ii) Solutions to (4) for $n \neq 2$ need not be "immersions except for a discrete set" as they are when n = 2. In fact, it is not known whether a unique continuation theorem holds when $n \neq 2$ (see [35, Theorem 2.3.2] for the case n = 2). Thus, in extending properties (x) and (xi), one runs into complications when trying to apply results about minimal submanifolds that prove useful in the study of holomorphic curves, such as the isoperimetric inequality or the monotonicity formula.

Of the two issues above, the second is less of a problem as we simply avoid it by working directly with the *n*-energy of our maps, as in [35], instead of applying minimal surface results (or more generally geometric measure theory) to their images, as in [40]. The reader interested in the distinction between these two approaches is suggested to compare, for example, [35, Theorem 4.4.1(i)] and [40, Lemma 3.1].

As for the first issue, one expects previous work on n-harmonic maps to help in dealing with the degeneracy of the equation, and indeed we are aided greatly by the results of Uhlenbeck [49], Giaquinta–Modica [17], Duzaar–Fuchs [10] and Duzaar–Mingione [11]. However, there is one crucial aspect that existing literature on n-harmonic maps does not fully cover, namely the continuity of $W^{1,n}$ -weak solutions (on n-dimensional domains), which has only been established in full generality for the case n=2 [26]. Analogous results for $n\neq 2$ all require additional assumptions (see for example [22, 46, 48]) that do not generally fit with our setting. This is where the special structure of the Smith equation (4) comes in. Specifically, we are able to show that equation (4) demonstrates a "compensation phenomenon" which has been observed in many other geometric PDEs, including n-harmonic maps into special targets. See §4.2, in particular Remarks 4.2 and 4.5, for details. Finally, we point out that results on minimizers of the n-energy, such as those obtained by Hardt–Lin [22], do not apply to our situation, as Smith maps are only minimizing within a fixed homology class.

Main results. We summarize the main results of the paper. In this section, the statements of the theorems are slightly simplified from their precise forms.

Let $(\Sigma^3, g, *)$ denote an oriented Riemannian 3-manifold equipped with the 2-fold vector cross product $* = P_{\Sigma} \colon \Gamma(\Lambda^2 T\Sigma) \to \Gamma(T\Sigma)$ given by the Hodge operator.

Let (M^7,h,J) denote a closed oriented Riemannian 7-manifold equipped with a 2-fold vector cross product $J = P_M \colon \Gamma(\Lambda^2 TM) \to \Gamma(TM)$. Note that the data (h,J) induces a definite 3-form $\varphi \in \Omega^3(M)$, and hence a G_2 -structure on M, by raising an index:

$$\varphi(x, y, z) = h(J(x, y), z).$$

For the moment, we do not assume that φ is closed or coclosed.

For technical convenience, we isometrically embed M in a Euclidean space \mathbb{R}^d .

The Sobolev space $W^{1,p}(\Sigma; M)$ is then by definition

$$W^{1,p}(\Sigma;M) = \{ u \in W^{1,p}(\Sigma;\mathbb{R}^d) \mid u(x) \in M \text{ for a.e. } x \in \Sigma \}.$$

Throughout, the manifolds Σ and M, the embedding $M \to \mathbb{R}^d$, and all of the relevant geometric data g, h, *, J, are assumed to be smooth.

The associative Smith equation (4) for maps $u \colon (\Sigma^3, g, *) \to (M^7, h, J)$ reads

$$J \circ \Lambda^2 du = \frac{1}{\sqrt{3}} |du| du \circ *.$$

In the case $u \in W^{1,3}(\Sigma; M)$, the weak derivative $(du)_x$ is a linear map $T_x\Sigma \to T_{u(x)}M$ for almost every $x \in \Sigma$, and at such points both sides of the above equation make sense. Hence we say that a map $u \in W^{1,3}(\Sigma; M)$ is an associative Smith map if the weak derivative du satisfies the above equation almost everywhere on Σ .

As above, we let $E(u;A) = \int_A |du|^3 \operatorname{vol}_{\Sigma}$ denote the 3-energy of u on a measurable set $A \subset \Sigma$.

Our first result is an ε -regularity theorem for weak solutions of (4), which gives $C^{1,\beta}$ -regularity with a priori estimates under a smallness assumption on the 3-energy. Two of the main ingredients for the proof are the regularity of n-harmonic functions due to Uhlenbeck [49], and a deep result from harmonic analysis due to Fefferman–Stein [14]. Note that because of the degeneracy of (4), in general we do not expect the solution to be better than $C^{1,\beta}$. However, on the set where du is nonzero we do get smoothness. For convenience we state the ε -regularity theorem in local terms and assume the domain is $B(2) \subset \mathbb{R}^3$. In this case we may identify the (weak) derivative du of a map $u \in W^{1,3}(B(2); \mathbb{R}^d)$ with an element of $L^3(B(2); \mathbb{R}^{3\times d})$, which we denote by Du.

THEOREM 4.7 (ε -regularity). There exist $\varepsilon_0, \beta > 0$ depending on M, J and the embedding $M \to \mathbb{R}^d$ such that if g is a Riemannian metric on B(2) with $|g - g_{euc}|_{0;B(2)} + |Dg|_{0;B(2)} < \varepsilon_0$ and $u \in W^{1,3}(B(2);M)$ is an associative Smith map with respect to g satisfying $E(u;B(2)) < \varepsilon_0$, then the following hold:

- (a) The map u belongs to $C^{1,\beta}(B(1);M)$ and the norm $|u|_{1,\beta;B(1)}$ can be estimated through M,J, the embedding $M\to\mathbb{R}^d$, and E(u;B(2)).
- (b) In addition, u is smooth on the open set $\{x \in B(1) \mid Du(x) \neq 0\}$.

On a number of occasions, especially in §5.5, we need a more explicit gradient estimate than the one obtained in part (a) of Theorem 4.7. The following result, based on an adaptation of [10], provides such an estimate, again under a smallness assumption on the energy. As in Theorem 4.7 we suppose that the domain Σ is the ball $B(2) \subset \mathbb{R}^3$.

Theorem 4.9 (Mean value inequality). Suppose that the Ricci curvature of the domain metric g is bounded, in the sense that

$$\left| Ric_{g(x)}(v,v) \right| \le K \left| v \right|_{g(x)}^2 \quad \text{ for all } x \in B(2), \, v \in \mathbb{R}^3.$$

Then there exist C > 0 and $\varepsilon_1 > 0$ such that if $|g - g_{euc}|_{0;B(2)} + |Dg|_{0;B(2)} \le \varepsilon_1$ and $u \in W^{1,3}(B(2);M)$ is an associative Smith map with $E(u;B(2)) < \varepsilon_1$, then

$$\sup_{B(\frac{1}{2})} |Du(x)|^3 \le CE(u; B(2))$$

where both C and ε_1 depend only on (M, J), the embedding $M \to \mathbb{R}^d$, and the constant K.

Covering the domain with small enough balls to which Theorem 4.7 is applicable, we obtain the following interior regularity result without a smallness assumption on the energy. We remark that, as opposed to Theorem 4.7, the $C^{1,\beta}$ -regularity given below is only qualitative as Theorem 4.10 does not provide an a priori estimate of the $C^{1,\beta}$ -norm and thus leaves open the possibility of a sequence of Smith maps bubbling.

THEOREM 4.10 (Interior regularity). Suppose that g is a smooth Riemannain metric on B(2), and that u is a $W^{1,3}$ -Smith map on B(2) with respect to g. Then u has Hölder continuous first derivatives on B(1). Moreover, u is C^{∞} on the open set $\{x \in B(1): Du(x) \neq 0\}$.

As another corollary of Theorem 4.7, we get that isolated singularities of associative Smith maps are removable.

Theorem 4.11 (Removable singularity). Suppose that g is a smooth Riemannian metric on B(2), and that $u \in C^1_{loc}(B(2) \setminus \{0\}; M)$ is a Smith map with respect to g, satisfying

$$\int_{B(2)} |Du|^3 dx < \infty.$$

Then in fact u extends to a C^1 -Smith map on all of B(2).

The next result can be viewed as a global version of Theorem 4.9. Specifically, as in [35, §4.1], a short argument using Theorem 4.9 and the conformal invariance of the Smith equation (4) establishes the following.

PROPOSITION 4.17 (Energy gap). There exists a constant $\varepsilon_0 > 0$, depending only on (M, J) and the embedding $M \to \mathbb{R}^d$, such that every C^1 associative Smith map $u: (S^3, g_{round}) \to (M^7, h)$ with $E(u; S^3) < \varepsilon_0$ is constant.

We then turn to sequences of associative Smith maps with uniformly bounded 3-energy. The following theorem is central to the construction of the bubble tree limit. Specifically, from Theorem 4.7 and Theorem 4.11, together with an argument using the weak-* compactness of Radon measures which is by now standard, we deduce C^1 -convergence away from a finite set of points to a $C^{1,\beta}$ -limit which is again a Smith map.

PROPOSITION 4.12 (Compactness modulo bubbling). Let (Σ^3, g) be a closed Riemannian 3-manifold. Let $\Omega \subset \Sigma$ be an open set, and let $\{\Omega_n\}$ be a sequence of open sets that exhaust Ω . Let g_n be a Riemannian metric on Ω_n such that $g_n \to g$ smoothly on compact subsets of Ω .

Let $u: (\Omega_n, g_n) \to (M^7, h)$ be a sequence of C^1 associative Smith maps satisfying a uniform 3-energy bound

$$E(u_n; \Omega_n) \le E_0.$$

Then there exists a finite set $S = \{x_1, \dots, x_q\} \subset \Sigma$ and an associative Smith map $u \in C^1_{loc}(\Omega; M)$ with $\int_{\Omega} |du|_g^3 d\mu_g \leq E_0$ such that, after passing to a subsequence:

- (a) We have $u_n \to u$ in $C^1_{loc}(\Omega \setminus \mathcal{S})$.
- (b) We have

$$|du_n|_{g_n}^3 d\mu_{g_n} \to |du|_g^3 d\mu_g + \sum_{i=1}^q m_i \delta(x_i)$$
 as Radon measures on Ω ,

where each $\delta(x_i)$ is a Dirac measure. Moreover, each $m_i \geq \frac{1}{2}\varepsilon_0$, where ε_0 is as in Theorem 4.7.

(c) If $||du_n||_{p;\Omega_n} \leq C$ for some $p \in (3,\infty]$, then $S = \varnothing$.

Remark 1.3. One can construct associative submanifolds of the form $\Sigma^2 \times S^1$ in $Y^6 \times S^1$, where Y^6 is a Calabi-Yau 3-fold. Because of this, one might think that, since holomorphic curves bubble at points, associative Smith maps should generically bubble along curves. The issue is that bubbling of holomorphic curves corresponds to concentration of 2-energy, whereas bubbling of associative Smith maps corresponds to concentration of 3-energy. This is all very carefully explained in §4.5 using relations between different classes of Smith maps we establish in §3.3.

In §5, we study the behavior of sequences of associative Smith maps with uniformly bounded 3-energy near the bubble points $x_i \in \mathcal{S}$. Indeed, using Theorem 4.12, the conformal invariance of the Smith equation, and the conformal invariance of the 3-energy functional E, we will see that such a sequence gives rise to a bubble tree of associative Smith maps.

In outline, bubble trees arise in the following way. Beginning with a sequence $u_n: (\Sigma^3, g) \to (M^7, h)$ of associative Smith maps with $E(u_n; \Sigma) \leq E_0$, Theorem 4.12 shows that a subsequence of u_n converges in C^1_{loc} to an associative Smith map $u_\infty: \Sigma \to M$ off of a finite set \mathcal{S} of bubble points in Σ .

If $S \neq \emptyset$, then for each bubble point $x_i \in S$, we may conformally rescale the sequence u_n in such a way that a subsequence of the rescaled maps converges in C^1_{loc} to an associative Smith map $\widetilde{u}_{\infty,i} \colon S^3 \to M^7$ off of a finite set $S_i \subset S^3 \setminus \{p^-\}$, where p^- is the south pole. If any of the sets $S_i \neq \emptyset$, then for each first-level bubble point $x_{ij} \in S_i$, we may again conformally rescale u_n so that a subsequence of the rescaled maps converge in C^1_{loc} to an associative Smith map $\widetilde{u}_{\infty,ij} \colon S^3 \to M$ off of a finite set $S_{ij} \subset S^3 \setminus \{p^-\}$, and so on.

In §5.2, we use the energy gap of Proposition 4.17 to show that this process does, in fact, terminate after a finite number of iterations. The result is a tree of associative Smith maps whose base vertex corresponds to the base map u_{∞} , whose higher vertices correspond to the bubble maps $\widetilde{u}_{\infty,I}$, and whose edges correspond to the bubble points x_I , where here $I=(i_1,\ldots,i_k)$ is a multi-index.

Having constructed the bubble tree, we turn to the analogues of Properties (x) and (xi). For this, we need to assume that the G_2 -structure φ on M is closed. In §5.4 and §5.5, respectively, we show that both 3-energy and homotopy are preserved in the "bubble tree limit".

THEOREM 5.1 (No energy loss). Suppose $d\varphi = 0$. If $u_n : \Sigma \to M$ is a sequence of associative Smith maps with uniformly bounded 3-energy, then

$$\lim_{n \to \infty} E(u_n) = E(u_\infty) + \sum_I E(\widetilde{u}_{\infty,I}).$$

THEOREM 5.2 (Zero neck length). Suppose $d\varphi = 0$. If $u_n : \Sigma \to M$ is a sequence of associative Smith maps with uniformly bounded 3-energy, then its bubble tree has no necks. Therefore, for each multi-index I, we have

$$\widetilde{u}_{\infty,I}(x_{Ij}) = \widetilde{u}_{\infty,Ij}(p^-).$$

In particular, the set

$$u_{\infty}(\Sigma) \cup \bigcup_{I} \widetilde{u}_{\infty,I}(S^{3})$$

is connected.

The precise meaning of "no necks" is explained in both §5.1 and §5.5.

Our proofs of Theorems 5.1 and 5.2 both depend on a more detailed understanding of the 3-energies of associative Smith maps on 3-dimensional annuli, which we study in $\S5.3$. In turn, this requires an energy gap result of Brian White [52] as well as the homologically energy-minimizing property of Smith maps (see Smith's result (iv) above), the latter of which requires that the G_2 -structure be closed.

Remark 1.4. Here is a simple explicit example of bubbling, adapted from [34, Example 1.1]. Let $\iota: (S^3, g_{\text{round}}) \to (M^7, h)$ be an isometric associative immersion of a round 3-sphere into a G₂-manifold (M^7, h) . For example, we could take (M^7, h) to be the spinor bundle $\mathcal{S}(S^3)$ equipped with the Bryant–Salamon metric [5], and take ι to be the inclusion map of the zero section. Let

$$\sigma \colon (S^3 \setminus \{p^-\}, g_{\text{round}}) \to (\mathbb{R}^3, g_{\text{euc}})$$

denote stereographic projection. Then the maps $u_n: (\mathbb{R}^3, g_{\text{euc}}) \to (M^7, h)$ given by $u_n(x) := \iota(\sigma^{-1}(nx))$ form a sequence of associative Smith maps with bounded 3-energy. Moreover, $u_n \to u_\infty$ in $C^1_{\text{loc}}(\mathbb{R}^3 \setminus \{0\})$, where $u_\infty(x) \equiv \iota(p^-)$ is a constant map, while

$$|du_n|^3 d\mu \to m\delta(\{0\})$$

as Radon measures on \mathbb{R}^3 for some m > 0.

1.3. Organization and notation. The paper is organized as follows. In $\S 2$, we present a comprehensive treatment of the linear algebra of Smith maps, including vector cross products and calibrations. In $\S 3$, we discuss Smith maps between manifolds, including the energy identity and the relation to n-harmonic maps. In $\S 4$, we establish many of the analytical results, including interior regularity, the mean value inequality, removable singularities, compactness modulo bubbling, and the energy gap. In $\S 5$, we give a detailed explanation of bubbling and the bubble tree, and prove that there is no energy loss and zero neck length when the G_2 -structure is closed. Finally, Appendix A collects some needed results from harmonic analysis, and Appendix B presents the proofs of two of the results from $\S 4$.

We employ the following notation throughout the paper:

- We use g_{euc} to denote the Euclidean metric on \mathbb{R}^n for any n.
- On a Riemannian manifold we use $|\cdot|$ for the pointwise norm on tensors induced from the metric. We use \inf_M to denote the injectivity radius of M and $\operatorname{diam}(A)$ to denote the diameter of a set A. In most cases the metric is clear from the context, but it is indicated by subscripts when necessary.
- When A is a set, we use the following notation for norms:

$$|\cdot|_{k;A}$$
 on $C^k(A)$, $|\cdot|_{k,\alpha;A}$ on $C^{k,\alpha}(A)$, $\|\cdot\|_{p;A}$ on $L^p(A)$, $\|\cdot\|_{1,p;A}$ on $W^{1,p}(A)$.

Moreover, $[\cdot]_{\alpha;A}$ denotes the Hölder semi-norm with exponent α on A.

- In a Riemannian manifold, we use B(x;r) to denote the open ball at x of radius r > 0. When local coordinates are chosen, we write B(r) := B(0;r) for brevity.
- When $u: \Sigma \to M$ is a C^1 map, then $du \in \Gamma(T^*\Sigma \otimes u^*TM)$ denotes the differential, so $(du)_x: T_x\Sigma \to T_{u(x)}M$ for all $x \in \Sigma$. By the embedding $M \to \mathbb{R}^d$, we may also view $(du)_x$ as a linear map $T_x\Sigma \to \mathbb{R}^d$ with image contained in $T_{u(x)}M$. The latter viewpoint extends to maps in $W^{1,3}(\Sigma; M)$, but only holds for almost every $x \in \Sigma$.
- If $\Sigma \subset \mathbb{R}^3$ and $u \in W^{1,3}(\Sigma; \mathbb{R}^d)$, the derivative du may be identified with a map in $L^3(\Sigma; \mathbb{R}^{3\times d})$, which we denote by Du.
- Given a function $F: A \to \mathbb{R}$, we use $(f)_A$ to denote its average value over A, namely

$$(f)_A = \frac{1}{\int_A \operatorname{vol}} \int_A F \operatorname{vol} = \oint_A F \operatorname{vol}.$$

Other notation is introduced and defined when it is needed.

Acknowledgements. The authors gratefully acknowledge useful conversations with Ali Aleyasin, Gavin Ball, Benoit Charbonneau, Aleksander Doan, Andriy Haydys, Jason Lotay, André Neves, Rick Schoen, Kyler Siegel, and Thomas Walpuski. The second author is supported by an NSERC Discovery Grant. The third author also thanks McKenzie Wang for his support and encouragement.

Remark. After this work was completed, the authors learned that Mou-Wang [37] had studied a conformally invariant n-harmonic type system which the Smith maps satisfy when n=3 or 4, and obtained a no energy loss result for bubble trees of solutions using methods quite different from ours. Indeed our proof of Theorem 5.1 relies on an isoperimetric-type estimate coming from the geometric properties of Smith maps (see Section 5.3), whereas in [37] the authors exploit again the compensation phenomenon mentioned in Section 1.2 (see [37, Lemma 3.3 and pages 363–364]). We refer the reader to the introduction of our Section 4.2 for comments on other similarities and contrasts between our work and [37].

2. The linear algebra of Smith maps. Vector cross products were introduced by Brown–Gray [3] and were further studied much later by Lee–Leung [31]. The notion of a vector cross product preserving map was introduced by Gray in [19]. We refer to such maps as *Gray maps*. A generalized notion, which can reasonably be called conformally vector cross product preserving, was introduced by Smith in [45], where they were called *multiholomorphic maps*. We refer to such maps as *Smith maps*. Calibrations were introduced in the seminal paper of Harvey–Lawson [23]. In this section we discuss the linear algebraic aspects of vector cross products, calibrations, and Smith maps.

A few of the results in §2.2 and in §2.4 are at least implicit in [19] for the case of Gray maps. We adapt the proofs to the more general case of Smith maps and flesh out several details that are missing from [19], which is in any case somewhat difficult to access. We also present many more results in §2.4 that do not appear to be in the literature.

The main results in this section are that any Smith map is weakly conformal, proved in Theorem 2.22, the relations between Smith maps and calibrations in §2.5, and the *generalized calibration inequality* established in Theorem 2.40. These results are also implicit in the unpublished preprint [45], although their precise statements

are somewhat obscured. In the present paper we significantly clarify both of these results, first by distilling them to their minimal hypotheses (which is entirely linear algebraic), and second by parcelling out the various required components into separate lemmas and propositions. We hope the careful exposition in this section will be useful to a wider audience.

2.1. Preliminaries. We use the term *Euclidean space* to denote a finite-dimensional real vector space equipped with a positive definite inner product. Here we collect several preliminary results on linear maps between Euclidean spaces and the exterior powers of such maps. In particular we require multiple versions of *Hadamard's inequality*. We collect them here with proofs for completeness.

Let $(V, \langle \cdot, \rangle)$ be an *n*-dimensional Euclidean space and let $(W, \langle \cdot, \rangle)$ be an *m*-dimensional Euclidean space. Let $A: V \to W$ be a linear map. Let $A^*: W \to V$ be the adjoint map. We define the *matrix norm*, also called the Frobenius norm or Hilbert–Schmidt norm, of A by $|A|^2 = \operatorname{tr}(A^*A)$. With respect to orthonormal bases $\{v_1, \ldots, v_n\}$ of $V \cong \mathbb{R}^n$ and $\{w_1, \ldots, w_m\}$ of $W \cong \mathbb{R}^m$, this norm is given by

$$|A|^{2} = \operatorname{tr}(A^{*}A) = \sum_{l=1}^{n} \sum_{i=1}^{m} A_{l}^{i} A_{l}^{i}$$

$$= \sum_{l=1}^{n} \sum_{i=1}^{m} \langle A_{l}^{i} w_{i}, A_{l}^{j} w_{j} \rangle = \sum_{l=1}^{n} \langle A v_{l}, A v_{l} \rangle = \sum_{l=1}^{n} |A v_{l}|^{2}.$$
(5)

We need to use both the $\operatorname{tr}(A^*A)$ and the $\sum_{l=1}^n \langle Av_l, Av_l \rangle$ expressions for $|A|^2$.

Lemma 2.1. The map $A: V \to W$ is called a conformal injection if $\langle Av_1, Av_2 \rangle = \lambda^2 \langle v_1, v_2 \rangle$ for some $\lambda > 0$. This is equivalent to $A^*A = \lambda^2 I$ and also equivalent to $A = \lambda \hat{A}$ where \hat{A} is an isometric injection. That is, $\langle \hat{A}v_1, \hat{A}v_2 \rangle = \langle v_1, v_2 \rangle$. When $\lambda = 1$ then $A = \hat{A}$ is an isometric injection. Moreover, we necessarily have $\lambda^2 = \frac{1}{n} |A|^2$ where $n = \dim V$.

Proof. From $\langle A^*Av_1, v_2 \rangle = \langle Av_1, Av_2 \rangle$, we have that $\langle Av_1, Av_2 \rangle = \lambda^2 \langle v_1, v_2 \rangle$ if and only if $A^*A = \lambda^2 I$. Let $\hat{A} = \lambda^{-1}A$. Then $A^*A = \lambda^2 I$ is equivalent to $\hat{A}^*\hat{A} = I$. The last statement follows from (5), since $|A|^2 = \operatorname{tr}(A^*A) = \operatorname{tr}(\lambda^2 I) = n\lambda^2$. \square

COROLLARY 2.2. Let $n = \dim V$. Suppose $A: V \to W$ is a conformal injection, and let $B: V \to V$ be a conformal isomorphism. Then $|AB| = \frac{1}{\sqrt{n}} |A| |B|$.

Proof. By Lemma 2.1 we have $A^*A = \lambda^2 I$ and $B^*B = \mu^2 I$, where $n\lambda^2 = |A|^2$ and $n\mu^2 = |B|^2$. Moreover since B is invertible we have $B^*B = BB^*$. Thus we have

$$|AB|^{2} = \operatorname{tr}((AB)^{*}(AB)) = \operatorname{tr}(B^{*}A^{*}AB) = \operatorname{tr}(A^{*}ABB^{*})$$
$$= \operatorname{tr}(\lambda^{2}\mu^{2}I) = n\lambda^{2}\mu^{2} = \frac{1}{n}|A|^{2}|B|^{2}$$

as claimed. \square

For $1 \le r \le \dim V$, let $\Lambda^r A : \Lambda^r V \to \Lambda^r W$ be the rth exterior power of A, defined by

$$(\Lambda^r A)(v_1 \wedge \cdots \wedge v_r) = (Av_1) \wedge \cdots \wedge (Av_r)$$

on decomposable elements and extended linearly to all of $\Lambda^r V$. It is immediate that $(\Lambda^r A^*) = (\Lambda^r A)^*$ and that $\Lambda^r (A_1 A_2) = (\Lambda^r A_1)(\Lambda^r A_2)$.

LEMMA 2.3. Let $B: V \to V$ be a positive self-adjoint linear map. This means that $\langle Bv, w \rangle = \langle v, Bw \rangle$ and $\langle Bv, v \rangle \geq 0$ with equality if and only if v = 0. Let $0 < r < n = \dim V$.

Suppose that $\Lambda^r B: \Lambda^r V \to \Lambda^r V$ is an isometry on decomposable elements. This means that

$$\langle Bv_1 \wedge \dots \wedge Bv_r, v_1 \wedge \dots \wedge v_r \rangle = |v_1 \wedge \dots \wedge v_r|^2 \quad \text{for all } v_1, \dots, v_r \in V.$$
 (6)

Then B is is the identity map on V.

Proof. By the spectral theorem, there is an orthonormal basis $\{v_1,\ldots,v_n\}$ of V consisting of eigenvectors of B, where the eigenvalues μ_1,\ldots,μ_n are strictly positive real numbers. Let $1 \leq i_1 < \cdots < i_r \leq n$. From $Bv_j = \mu_j v_j$, we have that $(Bv_{i_1}) \wedge \cdots \wedge (Bv_{i_r}) = (\mu_{i_1} \cdots \mu_{i_r}) v_{i_1} \wedge \cdots \wedge v_{i_r}$. By hypothesis (6) it follows that $\mu_{i_1} \cdots \mu_{i_r} = 1$ for all such strictly increasing multi-indices. The positivity of the μ_j 's now implies that $\mu_j = 1$ for all j, so Bv = v for all $v \in V$. Note that we definitely need 0 < r < n to obtain the conclusion. \square

LEMMA 2.4 (Hadamard's inequality on $\Lambda^r A$). Let $1 < r \le n = \dim V$. Let $A: V \to W$ be nonzero. Then we have

$$|\Lambda^r A|^2 \le n^{-r} \binom{n}{r} |A|^{2r},\tag{7}$$

with equality if and only if A is a conformal injection in the sense of Lemma 2.1.

Proof. Let $B=A^*A$, which is a nonnegative self-adjoint linear map. By the spectral theorem, there is an orthonormal basis $\{v_1,\ldots,v_n\}$ of V consisting of eigenvectors of B, where the eigenvalues μ_1,\ldots,μ_n are nonnegative. Then $\{v_{i_1}\wedge\cdots\wedge v_{i_r}:1\leq i_1<\cdots< i_r\leq n\}$ is an orthonormal basis of Λ^rV and $(\Lambda^rB)(v_{i_1}\wedge\cdots\wedge v_{i_r})=(\mu_{i_1}\cdots\mu_{i_r})v_{i_1}\wedge\cdots\wedge v_{i_r}$. Applying (5) to Λ^rA gives

$$|\Lambda^r A|^2 = \operatorname{tr}\left((\Lambda^r A)^* (\Lambda^r A)\right) = \operatorname{tr}\left(\Lambda^r (A^* A)\right) = \operatorname{tr}(\Lambda^r B)$$
$$= \sum_{1 \le i_1 < \dots < i_r \le n} (\mu_{i_1} \cdots \mu_{i_r}).$$

McLaurin's inequality, which is a generalization of the arithmetic-geometric mean inequality to other symmetric polynomials, says that if μ_1, \ldots, μ_n are nonnegative, then

$$\left(\frac{1}{\binom{n}{r}} \sum_{1 \le i_1 < \dots < i_r \le n} (\mu_{i_1} \cdots \mu_{i_r})\right)^{\frac{1}{r}} \le \left(\frac{1}{n} \sum_{j=1}^n \mu_j\right) \quad \text{with equality iff } \mu_1 = \dots = \mu_n.$$

Using this inequality and $\operatorname{tr} B = \operatorname{tr}(A^*A) = |A|^2$, we get that

$$|\Lambda^r A|^2 \le \binom{n}{r} \left(\frac{1}{n} \sum_{j=1}^n \mu_j\right)^r = \binom{n}{r} \left(\frac{1}{n} \operatorname{tr} B\right)^r = \binom{n}{r} n^{-r} |A|^{2r}$$

with equality if and only if $B = A^*A = \mu I$ for some $\mu \ge 0$, where $n\mu = |A|^2$. Since A is nonzero, we deduce that $\mu > 0$. The result now follows by Lemma 2.1. \square

COROLLARY 2.5. Let $A: V \to W$ be nonzero, and let $n = \dim V$. Then we have

$$|\Lambda^n A| \le \frac{1}{(\sqrt{n})^n} |A|^n,$$

with equality if and only if A is a conformal injection in the sense of Lemma 2.1.

Proof. This is obtained by taking square roots of both sides of (7) in the special case r = n. \square

COROLLARY 2.6. Let $n = \dim V$ and let v_1, \ldots, v_n be nonzero vectors in V. Then we have

$$|v_1 \wedge \dots \wedge v_n| \le |v_1| \dots |v_n|$$
 with equality iff v_1, \dots, v_n are orthogonal. (8)

Proof. By dividing both sides by $|v_1| \cdots |v_n|$, we can assume that v_1, \ldots, v_n are all *unit vectors*, and we need to prove that

$$|v_1 \wedge \cdots \wedge v_n|^2 \leq 1$$
 with equality iff v_1, \dots, v_n are orthonormal.

Let $\{u_1, \ldots, u_n\}$ be an orthonormal basis of V and define a linear map $A: V \to V$ by $Au_i = v_i$. By (5) we have

$$|A|^2 = \sum_{i=1}^n |Au_i|^2 = \sum_{i=1}^n |v_i|^2 = n.$$

Then we have

$$|v_1 \wedge \cdots \wedge v_n|^2 = |(Au_1) \wedge \cdots (Au_n)|^2 = |(\Lambda^n A)(u_1 \wedge \cdots \wedge u_n)|^2$$
.

Since the single element $u_1 \wedge \cdots \wedge u_n$ is an orthonormal basis for $\Lambda^n V$, equation (5) applied to $\Lambda^n A$ gives $|(\Lambda^n A)(u_1 \wedge \cdots \wedge u_n)|^2 = |\Lambda^n A|^2$. Using this and Corollary 2.5 we have

$$|v_1 \wedge \dots \wedge v_n|^2 = |\Lambda^n A|^2 \le \frac{1}{n^n} |A|^{2n} = \frac{1}{n^n} n^n = 1,$$

with equality if and only if A is a conformal injection. Note that by Lemma 2.1, since $|A|^2 = n$, the conformal factor must be 1 and thus A is an isometric injection. But by the definition of A this means equality occurs if and only if v_1, \ldots, v_n are orthonormal. \square

2.2. Vector cross products. Let $(V, \langle \cdot, \cdot \rangle)$ be an *n*-dimensional Euclidean space.

DEFINITION 2.7. Let $1 \le k \le n-1$. A k-fold vector cross product P on V is an element $P \in \Lambda^k(V^*) \otimes V$ that satisfies the following two properties:

$$P(v_1 \wedge \cdots \wedge v_k)$$
 is orthogonal to v_1, \dots, v_k , (9)

and

$$|P(v_1 \wedge \dots \wedge v_k)|^2 = |v_1 \wedge \dots \wedge v_k|^2.$$
(10)

Note that equation (9) is equivalent to the statement that the covariant (k+1)-tensor α_P defined by

$$\alpha_P(v_1, \dots, v_k, v_{k+1}) = \langle P(v_1 \wedge \dots \wedge v_k), v_{k+1} \rangle$$
 is totally skew-symmetric. (11)

The (k+1)-form α_P is called the *calibration form* associated to P. It is discussed in §2.3.

Example 2.8. Consider the case k=1. Conditions (9) and (10) say that $P:V\to V$ satisfies $\langle Pv,v\rangle=0$ and $|Pv|^2=|v|^2$. Polarizing these two equations gives $\langle Pv,w\rangle=-\langle v,Pw\rangle$, so P is skew-adjoint, and $\langle Pv,Pw\rangle=\langle v,w\rangle$, so P is an isometry. Thus we have

$$\langle P^2v, w \rangle = -\langle Pv, Pw \rangle = -\langle v, w \rangle.$$

Since this holds for all v, w, we have $P^2 = -I$ and thus P is an orthogonal complex structure on V. In particular then $n = \dim V$ must be even.

EXAMPLE 2.9. Consider the case k=n-1. Let u_1,\ldots,u_n be an orthonormal basis for V. Conditions (9) and (10) say that $P:\Lambda^{n-1}V\to V$ satisfies $\langle P(u_1\wedge\cdots\wedge u_{n-1}),u_i\rangle=0$ for $1\leq i\leq n-1$, so $P(u_1\wedge\cdots\wedge u_{n-1})$ must be a multiple of u_n , and $|P(u_1\wedge\cdots\wedge u_{n-1})|^2=|u_1\wedge\cdots\wedge u_{n-1}|^2=1$, so $P(u_1\wedge\cdots\wedge u_{n-1})=\pm u_n$. Let $\text{vol}=\pm u_1\wedge\cdots\wedge u_n$ be the orientation for V induced by the ordered orthonormal basis $\{u_1,\ldots,u_{n-1},\pm u_n\}$. Then P=* is the Hodge star operator on $\Lambda^{n-1}V$ corresponding to the inner product and this orientation. The Hodge star is an isometry.

In general, condition (10) says that $P: \Lambda^k(V) \to V$ is length-preserving on the *decomposable* elements of $\Lambda^k(V)$. When k=1 or k=n-1, any k-vector is decomposable, so the vector cross P is an honest isometry in the cases of Examples 2.8 and 2.9.

The fundamental identities (9) and (10) relating a vector cross product to the inner product are very strong constraints. In fact, there are only four possible types of vector cross products, the two we have already discussed and two exceptional types. This is the Brown–Gray classification [3] given in Table 1. The two exceptional vector cross products are the most interesting, as they are related to associative submanifolds of G₂-manifolds and to Cayley submanifolds of Spin(7)-manifolds, respectively.

Type	n	k	P	α	Name of structure
I	n	n-1	*	vol	Orientation
II	2m	1	J	ω	Orthogonal complex structure
III	7	2	×	φ	G_2 -structure
IV	8	3	P	Φ	Spin(7)-structure

 $TABLE\ 1 \\ The\ Brown-Gray\ classification\ of\ vector\ cross\ products.$

Despite this simple classification, almost all the properties of vector cross products and of Smith maps that we establish in the rest of this section and in §2.4 are consequences of the defining properties (9) and (10), so the statements and proofs are identical for all four types of vector cross products.

PROPOSITION 2.10. Let P be a k-fold vector cross product on $(V, \langle \cdot, \cdot \rangle)$. Let u_1, \ldots, u_{k-1} be linearly independent vectors in V and let $w \in V$ be arbitrary. Let U be the (k-1)-dimensional subspace of V spanned by u_1, \ldots, u_{k-1} . Then the following identity holds:

$$P(u_1 \wedge \dots \wedge u_{k-1} \wedge P(u_1 \wedge \dots \wedge u_{k-1} \wedge w)) = -|u_1 \wedge \dots \wedge u_{k-1}|^2 \pi_{U^{\perp}} w, \qquad (12)$$

where $\pi_{U^{\perp}}$ is orthogonal projection onto the subspace U^{\perp} .

Proof. We first polarize equation (10) in v_k to obtain

$$\langle P(u_1 \wedge \dots \wedge u_{k-1} \wedge v), P(u_1 \wedge \dots \wedge u_{k-1} \wedge w) \rangle$$

$$= \langle u_1 \wedge \dots \wedge u_{k-1} \wedge v, u_1 \wedge \dots \wedge u_{k-1} \wedge w \rangle. \tag{13}$$

Let $U = \text{span}\{u_1, \dots, u_{k-1}\}$. Since $w = \pi_U w + \pi_{U^{\perp}} w$ and $u_1 \wedge \dots \wedge u_{k-1} \wedge (\pi_U w) = 0$, we can replace w with $\pi_{U^{\perp}} w$ in the right hand side of equation (13). Thus the right hand side is

$$\langle u_1 \wedge \dots \wedge u_{k-1} \wedge v, u_1 \wedge \dots \wedge u_{k-1} \wedge (\pi_{U^{\perp}} w) \rangle = \det \begin{pmatrix} \langle u_i, u_j \rangle & \langle u_i, \pi_{U^{\perp}} w \rangle \\ \langle v, u_i \rangle & \langle v, \pi_{U^{\perp}} w \rangle \end{pmatrix}$$

$$= \det \begin{pmatrix} \langle u_i, u_j \rangle & 0 \\ \langle v, u_i \rangle & \langle v, \pi_{U^{\perp}} w \rangle \end{pmatrix}$$

$$= |u_1 \wedge \dots \wedge u_{k-1}|^2 \langle \pi_{U^{\perp}} w, v \rangle. \quad (14)$$

Using the definition and skew-symmetry of the calibration form α_P from (11) the left hand side of (13) becomes

$$\alpha_{P}(u_{1}, \dots, u_{k-1}, v, P(u_{1} \wedge \dots \wedge u_{k-1} \wedge w))$$

$$= -\alpha_{P}(u_{1}, \dots, u_{k-1}, P(u_{1} \wedge \dots \wedge u_{k-1} \wedge w), v)$$

$$= -\langle P(u_{1} \wedge \dots \wedge u_{k-1} \wedge P(u_{1} \wedge \dots \wedge u_{k-1} \wedge w)), v \rangle.$$
(15)

Since (15) and (14) agree for all $v \in V$, the identity (12) holds. \square

REMARK 2.11. In the particular case when k = 1, Proposition 2.10 simply says $P^2(u) = -u$ for all $u \in V$, which we already knew from the discussion in Example 2.8. But for k > 1 it is nontrivial.

Proposition 2.10 has two important corollaries.

COROLLARY 2.12. Let $v, u_1 \in V$ be orthogonal vectors. Then there exists a decomposable element of $\Lambda^k V$ of the form $u_1 \wedge \cdots \wedge u_k$ such that $v = P(u_1 \wedge \cdots \wedge u_k)$.

Proof of Corollary 2.12. By the linearity of P, we can without loss of generality assume that v and u_1 both have unit length. Choose orthonormal vectors u_2, \ldots, u_{k-1} in V that are orthogonal to both v and u_1 . We can always do this because $(k-2)+2=k < n = \dim V$. Let $U = \operatorname{span}\{u_1, \ldots, u_{k-1}\}$. By construction we have $\pi_{U^{\perp}}v = v$. Thus the fundamental identity (12) gives

$$P(u_1 \wedge \cdots \wedge u_{k-1} \wedge P(u_1 \wedge \cdots \wedge u_{k-1} \wedge v)) = -v.$$

Taking $u_k = -P(u_1 \wedge \cdots \wedge u_{k-1} \wedge v)$ completes the proof. \square

Remark 2.13. Corollary 2.12 says that not only is the linear map $P: \Lambda^k V \to V$ always surjective, but that we can in fact always choose a preimage of v that is of the decomposable form $u_1 \wedge \cdots \wedge u_k$, where u_1 is any nonzero vector orthogonal to v.

COROLLARY 2.14. Consider the same hypotheses as in Proposition 2.10. Then the following identity holds:

$$P(u_1 \wedge \dots \wedge u_{k-1} \wedge P(u_1 \wedge \dots \wedge u_{k-1} \wedge P(u_1 \wedge \dots \wedge u_{k-1} \wedge w)))$$

$$= -|u_1 \wedge \dots \wedge u_{k-1}|^2 P(u_1 \wedge \dots \wedge u_{k-1} \wedge w).$$
(16)

Proof. Let $y = \pi_{U^{\perp}}w$. Then $v - y = \pi_{U}w \in U$. Thus, by multinearity and the property (9) of vector cross products, we have

$$P(u_1 \wedge \dots \wedge u_{k-1} \wedge y) = P(u_1 \wedge \dots \wedge u_{k-1} \wedge w). \tag{17}$$

Take the wedge product of both sides of (12) with $u_1 \wedge \cdots \wedge u_{k-1}$ and then apply the vector cross product P to both sides. The observation (17) therefore establishes (16). \square

2.3. Calibration forms. Before we define Smith maps and Gray maps we review the closely related but somewhat more general notion of a *calibration* on V. This is used crucially in the generalized calibration inequality of Theorem 2.40. The theory of calibrations was introduced in [23] by Harvey–Lawson. We emphasize that in this section we are only considering the linear algebraic (pointwise) aspects of the theory.

DEFINITION 2.15. Let α be a (k+1)-form on V, where $k+1 \leq n = \dim V$. We say that α is a *calibration* if

$$\alpha(v_1, \dots v_{k+1}) \le |v_1 \wedge \dots \wedge v_{k+1}| \quad \text{for all } v_1, \dots, v_{k+1} \in V. \tag{18}$$

We say that the (k+1)-form α has comass one.

Lemma 2.16. The inequality (18) is equivalent to the inequality

$$\alpha(u_1, \dots u_{k+1}) \le 1$$
 for all orthonormal vectors $u_1, \dots, u_{k+1} \in V$. (19)

Proof. Both sides of (18) vanish if $\{v_1,\ldots,v_{k+1}\}$ is linearly dependent so we may assume it is linearly independent. Moreover, recall that $\Lambda^k V^* \cong (\Lambda^k V)^*$. That is, $\alpha(v_1,\ldots,v_{k+1})=\alpha(v_1\wedge\cdots\wedge v_{k+1})$. Hence, both sides of (18) depend only on the *oriented* (k+1)-plane $u_1\wedge\cdots\wedge u_{k+1}\in\Lambda^k V$ where $\{u_1,\ldots,u_{k+1}\}$ is an ordered orthonormal basis for span $\{v_1,\ldots,v_{k+1}\}$ inducing the same orientation. That is, $u_1\wedge\cdots\wedge u_k=tv_1\wedge\cdots\wedge v_{k+1}$ for some t>0. Thus we have

$$\alpha(u_1, \dots, u_{k+1}) = t\alpha(v_1, \dots, v_{k+1}) \le t|v_1 \wedge \dots \wedge v_{k+1}|$$

= $|u_1 \wedge \dots \wedge u_{k+1}| = 1$,

which is what we wanted to show. \square

DEFINITION 2.17. A (k+1)-dimensional oriented subspace L of V is called calibrated with respect to α if the maximum in (19) is achieved on L. That is, if $\alpha(u_1, \ldots, u_{k+1}) = 1$ for any oriented orthonormal basis $\{u_1, \ldots, u_{k+1}\}$ of L. Equivalently, if L is an oriented k-dimensional subspace of V with volume form vol_L , then L is calibrated with respect to α if and only if $\alpha|_L = \operatorname{vol}_L$.

The set $\{u_1 \wedge \cdots \wedge u_k \in \Lambda^k V : u_1, \ldots, u_k \text{ are orthonormal}\}$ of unit-length decomposable k-vectors in V is compact, as it is precisely the Grassmanian $G^+(k,V)$ of oriented k-planes in V. Hence, any nonzero (k+1)-form can be suitably rescaled to satisfy the comass one condition (18) to obtain a calibration. However, not all calibrations admit a rich class of calibrated k-planes. One of the most important classes of such calibrations arise from vector cross products. In fact, any vector cross product P on V induces a calibration α_P , originally introduced in (11), as the following result demonstrates.

PROPOSITION 2.18. Let P be a k-fold vector cross product on V. The element $\alpha_P \in \otimes^{k+1} V^*$ defined by

$$\alpha_P(v_1, \dots, v_k, v_{k+1}) = \langle P(v_1 \wedge \dots \wedge v_k), v_{k+1} \rangle \tag{20}$$

is a calibration on V. An oriented (k+1)-dimensional subspace L of V is calibrated with respect to α_P if and only if $P(u_1 \wedge \cdots \wedge u_k) = u_{k+1}$ whenever $\{u_1, \ldots, u_{k+1}\}$ is an oriented orthonormal basis for L.

Proof. We first observe that by (9), the covariant (k+1)-tensor α_P is totally skew-symmetric and hence a (k+1)-form. Let u_1, \ldots, u_{k+1} be unit vectors in V. By property (10), the Cauchy–Schwarz inequality, and Hadamard's inequality (8), we have

$$\alpha_P(u_1, \dots, u_k, u_{k+1}) \le |P(u_1 \wedge \dots \wedge u_k)| |u_{k+1}|$$

= $|u_1 \wedge \dots \wedge u_k| |u_{k+1}|$
 $\le |u_1| \dots |u_k| |u_{k+1}| = 1.$

Thus α_P is indeed a calibration. Equality is achieved if and only if we have equality in *both* the Cauchy–Schwarz and the Hadamard inequalities. This means that $\{u_1,\ldots,u_k\}$ must be *orthonormal* and that $u_{k+1}=P(u_1\wedge\cdots\wedge u_k)$. In particular by (9) the vector u_{k+1} is also orthogonal to u_1,\ldots,u_k and thus $\{u_1,\ldots,u_{k+1}\}$ is an orthonormal basis for the (k+1)-dimensional subspace L that it spans. Since $\alpha(u_1,\ldots,u_k,u_{k+1})=1$, the ordered orthonormal basis $\{u_1,\ldots,u_{k+1}\}$ induces the given orientation on L. The converse is clear. \square

The form α_P in (20) is called the calibration form associated to to the vector cross product P.

Despite Proposition 2.18, there are many more interesting calibrations, admitting a rich class of calibrated subspaces, that do not arise from vector cross products. The most well-studied calibrations are those introduced by Harvey–Lawson [23] that include all those associated to vector cross products plus several others. They are summarized in Table 2.

n	k+1	α	Calibrated subspaces	Associated VCP
n	n	vol	Entire space	Hodge star
2m	2	ω	Complex lines	Orthogonal complex structure
2m	2r	$\frac{1}{r!}\omega^r$	Complex r -planes	NONE
2m	m	$\operatorname{Re}(e^{i\theta}\Omega)$	Special Lagrangian m -planes	NONE
7	3	φ	Associative 3planes	2-fold VCP of a G ₂ -structure
7	4	$\psi = *\varphi$	Coassociative 4-planes	NONE
8	4	Φ	Cayley 4-planes	3-fold VCP of a Spin(7)-structure

 ${\it TABLE~2} \\ The~calibrations~discussed~in~Harvey-Lawson~[23].$

Another very interesting class of calibrations [4] is related to the quaternions, but appeared after [23].

It is interesting to note from Table 2 the particular case when V is equipped with an orthogonal complex structure J and the associated 2-form ω is defined by $\omega(v,w) = \langle Jv,w\rangle$. In this case, the complex r-planes in V are calibrated by $\frac{1}{r!}\omega^r$, which is the classical Wirtinger inequality, but only the case r=1 is associated to a vector cross product.

2.4. Smith maps and Gray maps. Let P be a k-fold vector cross product on the n-dimensional Euclidean space V with associated calibration (k+1)-form α_P . Further, let Q be a k-fold vector cross product on the m-dimensional Euclidean space W with associated calibration (k+1)-form α_Q , for the same k. We do not assume that m=n, although this is a special case.

Let $A:V\to W$ be a linear map. We consider a special type of such a linear map, which we call a Smith map. This is a linear map that is in a particular sense "compatible" with the vector cross products P,Q on V,W respectively.

DEFINITION 2.19. Let $\Lambda^k A : \Lambda^k V \to \Lambda^k W$ be the kth exterior power A. We say that A is a Smith map if is satisfies the equation

$$Q(\Lambda^k A) = \lambda^{k-1} A P \tag{21}$$

for some positive constant λ . Note that both sides of (21) are linear maps from $\Lambda^k V$ to W. Explicitly, A is a Smith map iff for all $v_1, \ldots, v_k \in V$, we have $Q(Av_1 \wedge \cdots \wedge Av_k) = \lambda^{k-1} A(P(v_1 \wedge \cdots \wedge v_k))$. Thus we also say that a Smith map is conformally vector cross product preserving.

REMARK 2.20. We show in Theorem 2.22 and Proposition 2.24 that if k > 1 or (k = 1, n = 2) then the constant λ in (21) is not arbitrary if A is nonzero.

DEFINITION 2.21. A special case of Smith maps corresponds to $\lambda = 1$. Then equation (21) becomes $Q(\Lambda^k A) = AP$. Such a map is called a *Gray map* and is *vector cross product preserving*.

We now establish the fundamental properties of Smith maps.

THEOREM 2.22. Let k > 1 and let $A : V \to W$ be a Smith map. Then either A = 0 or A is a conformal injection in the sense of Lemma 2.1. Thus if $A \neq 0$ then λ is necessarily given by $\lambda = (\dim V)^{-\frac{1}{2}}|A|$.

Proof. Suppose that A is not injective. Then there exists nonzero $u_1 \in V$ with $Au_1 = 0$. We want to show that Av = 0 for all $v \in V$. We can assume v is nonzero and orthogonal to u_1 . By Corollary 2.12 we can find $u_2, \ldots, u_k \in V$ such that $v = P(u_1 \wedge \cdots \wedge u_k)$. Applying $\lambda^{k-1}A$ to both sides and using the fact that A is a Smith map, we find that

$$\lambda^{k-1}Av = \lambda^{k-1}A(P(u_1 \wedge \dots \wedge u_k)) = Q(Au_1 \wedge \dots \wedge Au_k) = 0.$$

Since $\lambda > 0$, we deduce that Av = 0. Because v was arbitrary we conclude that A = 0 whenever A is not injective.

From now on assume A is nonzero and thus injective. Let u_1, \ldots, u_{k-1}, v be linearly independent vectors in V. From Corollary 2.14 for (V, P), we have that

$$P\Big(u_1 \wedge \dots \wedge u_{k-1} \wedge P\big(u_1 \wedge \dots \wedge u_{k-1} \wedge P\big(u_1 \wedge \dots \wedge u_{k-1} \wedge v\big)\Big)\Big)$$

= $-|u_1 \wedge \dots \wedge u_{k-1}|^2 P(u_1 \wedge \dots \wedge u_{k-1} \wedge v).$

We apply $\lambda^{3(k-1)}A$ to both sides of the above and use the Smith equation (21) repeatedly. This gives

$$Q\Big(Au_1 \wedge \cdots \wedge Au_{k-1} \wedge Q\big(Au_1 \wedge \cdots \wedge Au_{k-1} \wedge Q(Au_1 \wedge \cdots \wedge Au_{k-1} \wedge Av)\big)\Big)$$

$$= -|u_1 \wedge \cdots \wedge u_{k-1}|^2 \lambda^{2(k-1)} Q(Au_1 \wedge \cdots \wedge Au_{k-1} \wedge Av).$$
(22)

Since A is injective, the vectors $Au_1, \dots Au_{k-1}, Av$ are linearly independent in W. From Corollary 2.14 for (W, Q), we have that

$$Q\Big(Au_1 \wedge \dots \wedge Au_{k-1} \wedge Q\big(Au_1 \wedge \dots \wedge Au_{k-1} \wedge Q(Au_1 \wedge \dots \wedge Au_{k-1} \wedge Av)\big)\Big)$$

$$= -|Au_1 \wedge \dots \wedge Au_{k-1}|^2 Q(Au_1 \wedge \dots \wedge Au_{k-1} \wedge Av).$$
(23)

From (10) and the linear independence of $Au_1, \ldots, Au_{k-1}, Av$ we have $Q(Au_1 \wedge \cdots \wedge Au_{k-1} \wedge Av) \neq 0$. Comparing equations (22) and (23) therefore gives

$$|Au_1 \wedge \dots \wedge Au_{k-1}|^2 = \lambda^{2(k-1)} |u_1 \wedge \dots \wedge u_{k-1}|^2$$
 (24)

whenever u_1, \ldots, u_{k-1} are linearly independent, and hence by multilinearity for all u_1, \ldots, u_{k-1} . Let $A^* : W \to V$ be the adjoint map, and let $B = \lambda^{-2} A^* A : V \to V$. The map B is self-adjoint and positive. Equation (24) says

$$|u_1 \wedge \dots \wedge u_{k-1}|^2 = \lambda^{-2(k-1)} \det \langle Au_i, Au_j \rangle$$

= \det\langle \langle^2 A^* Au_i, \, u_i \rangle = \langle Bu_1 \langle \dots \langle Bu_{k-1}, \, u_1 \langle \dots u_{k-1} \rangle.

We can thus apply Lemma 2.3 (this is where we need the hypothesis that k>1) to conclude that B is the identity, so $A^*A=\lambda^2I$. From Lemma 2.1 we conclude that A is conformal with conformal factor $\lambda=(\dim V)^{-\frac{1}{2}}|A|$ as claimed. \square

COROLLARY 2.23. For k > 1, the nonzero Gray maps are precisely the nonzero Smith maps that are also isometric injections.

Proof. This is immediate from Theorem 2.22. \square

The case when k=1 is special, as it requires a restriction on the dimension of the domain V. It is well known but we include it for completeness. The precise statement is as follows.

PROPOSITION 2.24. Let k=1. Let $A:V\to W$ be a Smith map. Then the conclusion of Theorem 2.22 always holds if and only if dim V=2.

Proof. Let dim V=n and dim W=m. By Example 2.8, the maps $P=V\to V$ and $Q:W\to W$ are both orthogonal complex structures, and moreover $n=\dim V=2r$ and $m=\dim W=2s$. We can choose orthonormal bases of the form $\{u_1,\ldots,u_p,Pu_1,\ldots,Pu_r\}$ and $\{e_1,\ldots,e_s,Qe_1,\ldots,Qe_s\}$ of V and W, respectively. With respect to such bases, the matrices for P and Q are

$$P = \begin{pmatrix} 0_{r \times r} & -I_{r \times r} \\ I_{r \times r} & 0_{r \times r} \end{pmatrix}, \qquad Q = \begin{pmatrix} 0_{s \times s} & -I_{s \times s} \\ I_{s \times s} & 0_{s \times s} \end{pmatrix}.$$

The Smith equation (21) in this case is AP = QA. It follows easily from this equation that with respect to these bases, the $2s \times 2r$ matrix for A must be of the block diagonal form

$$A = \begin{pmatrix} B & -C \\ C & B \end{pmatrix}$$

where B, C are $s \times r$ matrices. Because we are using orthonormal bases, the matrix of the adjoint A^* is just the transpose. Hence we have

$$A^*A = \begin{pmatrix} B^T & C^T \\ -C^T & B^T \end{pmatrix} \begin{pmatrix} B & -C \\ C & B \end{pmatrix} = \begin{pmatrix} B^TB + C^TC & C^TB - B^TC \\ -(C^TB - B^TC) & B^TB + C^TC \end{pmatrix}.$$

By Lemma 2.1, if A is nonzero then it is a conformal injection if and only if $A^*A = \lambda^2 I$ for some $\lambda > 0$. If dim V = 2, so r = 1, then B, C are $s \times 1$ column vectors, and B^TC is 1×1 . Thus in this case

$$A^{T}A = \begin{pmatrix} |B|^{2} + |C|^{2} & 0\\ 0 & |B|^{2} + |C|^{2} \end{pmatrix} = (|B|^{2} + |C|^{2})I.$$

Thus a nonzero Smith map A is indeed a conformal injection in this case. However, if r > 1 it is clear that there exist choices of B, C which do not yield conformal injections. \square

Next we investigate the properties of Smith maps under composition and inversion.

Proposition 2.25. The composition of Smith maps is a Smith map. If a Smith map is invertible, then its inverse is a Smith map.

Proof. Let (V, P), (W, Q), and (U, R) be Euclidean spaces equipped with k-fold vector cross products P, Q, R, respectively. Let $A: V \to W$ and $B: W \to U$ be Smith maps. Then we have

$$Q(\Lambda^k A) = \lambda^{k-1} A P, \qquad R(\Lambda^k B) = \mu^{k-1} B Q \tag{25}$$

for some $\lambda, \mu > 0$. Define $C = BA : V \to W$. Since $\Lambda^k(BA) = (\Lambda^k B)(\Lambda^k A)$, from (25) we have

$$\begin{split} R(\Lambda^k C) &= R(\Lambda^k B)(\Lambda^k A) = \mu^{k-1} B Q(\Lambda^k A) \\ &= \mu^{k-1} \lambda^{k-1} B A P = (\mu \lambda)^{k-1} C P = \rho^{k-1} C P \end{split}$$

where $\rho = \lambda \mu > 0$. Thus C is a Smith map.

Now suppose that $A: V \to V$ is an invertible Smith map. The scaling of A^{-1} is obviously inverse to the scaling of A. But we need to show that the inverse is still conformally cross-product preserving. From $(\Lambda^k A)^{-1} = (\Lambda^k A^{-1})$ we have

$$\begin{split} P(\Lambda^k A) &= \lambda^{k-1} A P &\iff P &= \lambda^{k-1} A P(\Lambda^k A^{-1}) \\ &\iff (\lambda^{-1})^{k-1} A^{-1} P = P(\Lambda^k A^{-1}). \end{split}$$

Thus A^{-1} is a Smith map. \square

COROLLARY 2.26. The composition of Gray maps is a Gray map. If a Gray map is invertible, then its inverse is a Gray map.

Proof. This is immediate from Proposition 2.25 and Corollary 2.23. \square

2.5. Smith maps and calibrations. In this section we investigate relations between Smith maps and calibrations. In a precise sense that we explain, the image of a Smith map is calibrated and conversely, any calibrated subspace is the image of a Smith map in many different ways. The consequences for manifolds are discussed in §3.

LEMMA 2.27. Let $A: V \to W$ be a Smith map. Then we have

$$A^* \alpha_Q = \lambda^{k+1} \alpha_P. \tag{26}$$

Proof. Let $v_1, \ldots, v_{k+1} \in V$ be arbitrary. We compute

$$A^* \alpha_Q(v_1, \dots, v_{k+1}) = \alpha_Q(Av_1, \dots, Av_{k+1}) = \langle Q(\Lambda^k A)(v_1 \wedge \dots \wedge v_k), Av_{k+1} \rangle$$

$$= \langle \lambda^{k-1} A P(v_1 \wedge \dots \wedge v_k), Av_{k+1} \rangle$$

$$= \lambda^{k-1} \lambda^2 \langle P(v_1 \wedge \dots \wedge v_k), v_{k+1} \rangle$$

$$= \lambda^{k+1} \alpha_P(v_1, \dots, v_{k+1})$$

and hence (26) holds. \square

We have seen that if A is a nonzero Smith map and either k > 1 or (k = 1, n = 2) so that either Theorem 2.22 or Proposition 2.24 holds, then A is a conformal injection with conformal factor $\lambda > 0$ where $\lambda^2 = \frac{1}{n}|A|^2$ and $n = \dim V$. Moreover, by Lemma 2.27, any Smith map satisfies (26). In fact, when the vector cross product P on the domain V is of Type I from Table 1, then these two conditions together are equivalent to the Smith map equation (21). The precise statement is proved in Proposition 2.32 below. First we need some definitions.

DEFINITION 2.28. Let V, W be Euclidean spaces with calibration (k+1)-forms α_V, α_W , respectively. Let $A: V \to W$ be a nonzero linear map. Suppose that the following two conditions both hold:

- (i) A is a conformal injection, necessarily with conformal factor $\lambda = \frac{1}{\sqrt{\dim V}} |A|$. Equivalently we write $A^*g_W = \lambda^2 g_V$ where g_V, g_W are the Euclidean inner products on V, W, respectively;
- (ii) $A^*\alpha_W = \lambda^{k+1}\alpha_V$.

Then we say that A is conformally calibrating.

Remark 2.29. In two particular cases, condition (ii) of Definition 2.28 automatically implies condition (i). These are the cases when V=W and $\alpha_P=\alpha_Q$ is a calibration induced from a vector cross product P=Q of either Type III or Type IV from Table 1. The fact that $A^*\alpha_P=\lambda^{k+1}\alpha_P$ implies $A^*g_V=\lambda^2g_V$ in these two cases is well-known. See [30, Sections 3.1 and 5.1], for example. The reason this happens is that a G₂-structure or a Spin(7)-structure determines the Euclidean inner product uniquely.

Assumption 2.30. From now on, we restrict to the case when the vector cross product P on the domain V is of Type I from Table 1.

DEFINITION 2.31. An oriented Euclidean n-space $(V, g_V, *_V, \mathsf{vol}_V)$ is an n-dimensional Euclidean space (V, g_V) equipped with an orientation which induces a Hodge star operator $*_V$. The associated calibration n-form is the volume form vol_V . By Example 2.9 this is the same as saying (V, g_V) is equipped with a vector cross product of Type I from Table 1. In [45] such a structure $(V, g_V, *_V, \mathsf{vol}_V)$ is called a 'conformal n-triad' but we do not use this terminology.

An (n-1)-fold VCP space (W, g_W, Q, α_Q) is a Euclidean space (W, g_W) equipped with an (n-1)-fold vector cross product Q on W and its associated calibration n-form α_Q . Note that if dim W=n then an (n-1)-fold VCP space is an oriented Euclidean n-space by Table 1.

PROPOSITION 2.32. Let $(V, g_V, *, \mathsf{vol}_V)$ be an oriented Euclidean n-space. Let (W, g_W, Q, α_Q) be an (n-1)-fold VCP space. Let $A: V \to W$ be a nonzero linear

map. Then A is a Smith map in the sense of equation (21) if and only if A is conformally calibrating in the sense of Definition 2.28.

Proof. We first note that necessarily we have either k = n - 1 > 1 or else (k = 1, n = 2). In these cases we have already shown that any Smith map is conformally calibrating. Thus assume that A is conformally calibrating. That is,

$$A^*g_W = \lambda^2 g_V, \qquad A^*\alpha_O = \lambda^{k+1}\alpha_P = \lambda^n \text{vol}_V, \tag{27}$$

and we need to prove that

$$Q(\Lambda^{n-1}A)(v_1 \wedge \dots \wedge v_{n-1}) = \lambda^{n-2}AP(v_1 \wedge \dots \wedge v_{n-1})$$
(28)

for all $v_1, \ldots, v_{n-1} \in V$. Both sides of (28) vanish if $\{v_1, \ldots, v_{n-1}\}$ is linearly dependent so we may assume it is linearly independent, so both sides of (28) depend only on the (n-1)-plane $u_1 \wedge \cdots \wedge u_{n-1} \in \Lambda^{n-1}V$ where $\{u_1, \ldots, u_{n-1}\}$ is an orthonormal basis for span $\{v_1, \ldots, v_{n-1}\}$. That is, $u_1 \wedge \cdots \wedge u_{n-1} = tv_1 \wedge \cdots \wedge v_{n-1}$ for some nonzero $t \in \mathbb{R}$. Thus (28) holds for all $v_1, \ldots, v_{n-1} \in V$ if and only if

$$Q(\Lambda^{n-1}A)(u_1 \wedge \dots \wedge u_{n-1}) = \lambda^{n-2}AP(u_1 \wedge \dots \wedge u_{n-1})$$
(29)

holds for all orthonormal $u_1, \ldots, u_{n-1} \in V$. By Lemma 2.1 we know $A = \lambda \hat{A}$ where $\lambda > 0$ and $\hat{A}: V \to W$ is an isometric injection. Therefore (27) becomes

$$\hat{A}^* g_W = g_V, \qquad \hat{A}^* \alpha_Q = \text{vol}_V. \tag{30}$$

Let u_1, \ldots, u_{n-1} be orthonormal. Let $u_n = P(u_1 \wedge \cdots \wedge u_{n-1})$. Then $\{u_1, \ldots, u_n\}$ is an oriented orthonormal basis for V since P = * is the Hodge star operator. Using (30) we compute

$$1 = \text{vol}_{V}(u_{1}, \dots u_{n}) = (\hat{A}^{*} \alpha_{Q})(u_{1}, \dots, u_{n})$$

$$= \alpha_{Q}(\hat{A}u_{1}, \dots, \hat{A}u_{n-1}, \hat{A}u_{n})$$

$$= g_{W}(Q(\Lambda^{n-1}\hat{A})(u_{1} \wedge \dots \wedge u_{n-1}), \hat{A}u_{n}).$$
(31)

Since \hat{A} is an isometric injection, $\hat{A}u_n$ and $(\Lambda^{n-1}\hat{A})(u_1 \wedge \cdots \wedge u_{n-1})$ are unit vectors in W and $\Lambda^{n-1}W$, respectively. Since Q is a vector cross product, property (10) says that $Q(\Lambda^{n-1}\hat{A})(u_1 \wedge \cdots \wedge u_{n-1})$ is also a unit vector in W. Then Cauchy–Schwarz applied to the equality (31) gives

$$Q(\Lambda^{n-1}\hat{A})(u_1 \wedge \dots \wedge u_{n-1}) = \hat{A}u_n = \hat{A}P(u_1 \wedge \dots \wedge u_{n-1}).$$
(32)

Multiplying both sides by λ^{n-1} gives equation (29) as required. \square

The above result has a number of important corollaries.

COROLLARY 2.33. Let $(V, g_V, *, \text{vol}_V)$ be an oriented Euclidean n-space. Let (W, g_W, Q, α_Q) be an (n-1)-fold VCP space. Let $A: V \to W$ be a nonzero Smith map. Then A induces an orientation on its image L = A(V) such that L is calibrated with respect to α_Q .

Proof. Since $A = \lambda \hat{A}$ where \hat{A} is an isometric injection, the image $L = A(V) = \hat{A}(V)$ is isometric to V and thus inherits an induced orientation by declaring that $\{\hat{A}u_1, \ldots, \hat{A}u_n\}$ is oriented (and necessarily orthonomal) whenever $\{u_1, \ldots, u_n\}$ is an

oriented orthonormal basis for V. Now equation (32) in the proof of Proposition 2.32 says that

$$Q((\hat{A}u_1) \wedge \cdots \wedge (\hat{A}u_{n-1})) = \hat{A}u_n.$$

Then Proposition 2.18 applied with P replaced by Q and u_i replaced by $\hat{A}u_i$ says that L is calibrated with respect to α_Q . \square

Corollary 2.34. Let $(V, g_V, *, \mathsf{vol}_V)$ be an oriented Euclidean n-space. A nonzero Smith map $A: V \to V$ is precisely an orientation preserving conformal isomorphism.

Proof. Apply Proposition 2.32 with W=V and Q=P=*. Then $A:V\to V$ is a nonzero Smith map if and only if it satisfies $A^*g_V=\lambda^2g_V$ and $A^*\mathrm{vol}_V=\lambda^n\mathrm{vol}_V$, with $n\lambda=|A|>0$. This precisely means that A is an orientation preserving conformal isomorphism. \square

COROLLARY 2.35. Let $(V, g_V, *, \text{vol}_V)$ be an oriented Euclidean n-space. Let (W, g_W, Q, α_Q) be an (n-1)-fold VCP space. Let $A: V \to W$ be a Smith map. Then for any orientation preserving conformal isomorphism $B: V \to V$, the composition $AB: V \to W$ is a Smith map.

Proof. This is immediate from Corollary 2.34 and Proposition 2.25. \square

Remark 2.36. Corollary 2.35 is very important, as it implies that the notion of a Smith map between appropriate Riemannian manifolds is a *conformally invariant* notion. This is discussed in §3.1.

In Corollary 2.33 we showed that the image of a nonzero Smith map is calibrated. In fact, a kind of converse holds, which is Proposition 2.38 below.

LEMMA 2.37. Let (W, g_W, Q, α_Q) be an (n-1)-fold VCP space. Let L be an n-dimensional oriented subspace, and let $\iota: L \to W$ be the linear inclusion. Equip L with the induced inner product $g_L = \iota^* g_W$ and volume form vol_L . Then L is calibrated with respect to α_Q if and only if $\iota: L \to W$ is a Gray map.

Proof. The map $\iota: L \to W$ is a Gray map if and only if it is a Smith map with $\lambda = 1$. By Proposition 2.32 we deduce that ι is a Gray map if and only if $\iota^* \alpha_Q = \mathsf{vol}_L$, which is precisely the condition that L is calibrated with respect to α_Q . \square

More generally, we have the following result.

PROPOSITION 2.38. Let (W, g_W, Q, α_Q) be an (n-1)-fold VCP space. Let V be a n-dimensional real vector space. Let $A: V \to W$ be a linear injection. Equip V with the inner product $g_V = A^*g_W$. Then the following are equivalent:

- (i) The image L = A(V) admits an orientation for which L is calibrated with respect to α_O .
- (ii) The space V admits an orientation such that $A: V \to W$ is a Gray map, with respect to $*_{g_V}$.

Proof. Let L=A(V) and give L the induced inner product $g_L=\iota^*g_W$ where $\iota:L\to W$ is the inclusion. Let A_1 denote the map A with codomain L=A(V). That is, $A_1:V\to L$ is given by $A_1(v)=A(v)$. Then $A_1:V\to L$ is a linear isomorphism, and $A=\iota A_1$, so $A^*=A_1^*\iota^*$. Hence $g_V=A^*g_W=A_1^*g_L$, so $A_1:V\to L$ is an isometry, and any orientation vol_L on L compatible with g_L corresponds to a unique orientation

 $A_1^* \text{vol}_L$ on V compatible with $g_V = A_1^* g_L$. Thus $A_1 : V \to L$ is an invertible Gray map with respect to the Hodge star operators $*_{g_V}, *_{g_L}$ on V, L, respectively.

By Lemma 2.37, condition (i) is equivalent to the statement that $\iota: L \to W$ is a Gray map with respect to some orientation vol_L on L compatible with g_L . Using $A = \iota A_1$ and the invertibility of A_1 , we deduce from Corollary 2.26 that ι is a Gray map if and only if A is a Gray map. \square

Remark 2.39. Suppose V is an n-dimensional real vector space and $A:V\to W$ is a linear injection where (W,g_W,Q,α_Q) is an (n-1)-fold VCP space, and that the image L=A(V) admits an orientation for which L is calibrated with respect to α_Q . Then Proposition 2.38 gives the structure of an oriented Euclidean n-space on V such that $A:V\to W$ is a Gray map. But then Corollaries 2.34 and 2.35 say that by precomposing A with any orientation preserving conformal isomorphism B of $(V,g_V,*,\mathrm{vol}_V)$, we obtain another Smith map $AB:V\to W$ whose image is calibrated.

2.6. A generalized calibration inequality. We now establish the fundamental generalized calibration inequality that is a crucial ingredient for the energy identity in §3.4.

THEOREM 2.40 (Generalized calibration inequality). Let $(V, g_V, *, \mathsf{vol}_V)$ be an oriented Euclidean n-space. Let (W, g_W, Q, α_Q) be an (n-1)-fold VCP space. Let $A: V \to W$ be a linear map. Let $\{u_1, \ldots, u_n\}$ be an oriented orthonormal basis for V. Then we have

$$\left(\frac{1}{\sqrt{n}}\right)^n |A|^n - (A^*\alpha_Q)(u_1, \dots, u_n) \ge 0, \tag{33}$$

with equality if and only if $A: V \to W$ is a Smith map.

Proof. If A is zero then equality holds in (33) trivially, and the zero map trivially satisfies (21). Thus assume that A is nonzero. Let u_1, \ldots, u_n be an oriented orthonormal basis of V. We compute

$$(A^*\alpha_Q)(u_1,\ldots,u_n)$$

$$=\alpha_Q(Au_1,\ldots,Au_n)$$

$$\leq |(Au_1)\wedge\cdots\wedge(Au_n)| \quad \text{by (18), since } \alpha_Q \text{ is a calibration} \qquad (34)$$

$$=|(\Lambda^nA)(u_1\wedge\cdots\wedge u_n)|$$

$$=|\Lambda^nA| \quad \text{by (5), since } \{u_1\wedge\cdots\wedge u_n\} \text{ is o.n. basis for } \Lambda^nV$$

$$\leq \left(\frac{1}{\sqrt{n}}\right)^n|A|^n \quad \text{by Hadamard's inequality (7) with } r=n \qquad (35)$$

Thus the inequality (33) has been established. Equality holds if and only if equality holds in both (34) and (35). By Lemma 2.4, equality holds in (35) if and only if condition (i) of Definition 2.28 holds. But then $A = \lambda \hat{A}$ where $\hat{A}: V \to W$ is an isometric injection. Thus $\{\hat{A}u_1, \ldots, \hat{A}u_n\}$ is an orthonormal basis for the *n*-dimensional subspace im A of W. By Corollary 2.6 applied to im A we have

$$|(Au_1) \wedge \cdots \wedge (Au_n)| = \lambda^n |(\hat{A}u_1) \wedge \cdots \wedge (\hat{A}u_n)| = \lambda^n.$$

Thus, given equality in (35), equality also holds in (34) if and only if $(A^*\alpha_Q)(u_1,\ldots,u_n)=\lambda^n$, and this holds if and only if $A^*\alpha_Q=\lambda^n \text{vol}_V$, which is

condition (ii) of Definition 2.28. Thus $A:V\to W$ is conformally calibrating. Using Proposition 2.32, we conclude that equality holds in (33) if and only if A is a Smith map. \square

Remark 2.41. We remark that Theorem 2.40 generalizes Proposition 2.18 in two important ways:

- It allows general injective linear maps $A:V\to W$. This is the passage from subspaces to maps.
- It allows the freedom that g_V is only conformal to A^*g_W , not necessarily isometric. This is the passage from Gray maps to Smith maps.

The consequences for manifolds are discussed in §3.

- 3. Smith maps between manifolds. In this section we apply the linear algebraic results of §2 to maps between manifolds equipped with the appropriate geometric structures. The most important result of this section is that, if $d\alpha = 0$, then a Smith map $u:(\Sigma^n,g,*,\mathsf{vol})\to (M^m,h,Q,\alpha)$ is a minimizer of the n-energy in its homology class, and thus is also an n-harmonic map.
- 3.1. Definition and basic properties of Smith maps between manifolds. Let (Σ, q) be an oriented Riemannian n-manifold, with Riemannian volume form vol. At each $x \in \Sigma$, the associated Hodge star operator $*_x$ is an (n-1)-fold vector cross product on $T_x\Sigma$ of Type I from Table 1 and vol_x is its associated calibration n-form. Thus $(T_x \Sigma, g_x, *_x, \mathsf{vol}_x)$ is an oriented Euclidean n-space for each $x \in \Sigma$, as in Definition 2.31.

Let (M,h) be a Riemannian m-manifold equipped with an (n-1)-fold vector cross product Q and its associated calibration n-form α . This means that Q is a smooth section of $\Lambda^{n-1}T^*M\otimes TM$ and α is a smooth n-form on M such that, at each $y \in M$, the map $Q_y : \Lambda^{n-1}T_yM \to T_yM$ is a vector cross product with associated calibration n-form α_y . Thus $(T_yM, h_y, Q_y, (\alpha_Q)_y)$ is an (n-1)-fold VCP space, as in Definition 2.31.

Definition 3.1. Let $u: \Sigma \to M$ be a smooth map. We say that u is a Smith map if the differential $(du)_x: T_x\Sigma \to T_{u(x)}M$ is a Smith map for all $x\in \Sigma$ in the sense of Definition 2.19. Explicitly, a Smith map satisfies the equation

$$Q(\Lambda^{n-1}du) = \frac{1}{(\sqrt{n})^{n-2}} |du|^{n-2} (du) *$$
(36)

where * is the Hodge star operator on Σ . Both sides of (36) are smooth sections of $\Lambda^{n-1}T^*\Sigma\otimes u^*TM$. In [45] the Smith maps are called multiholomorphic maps, which is reasonable given Example 3.9 in §3.2.

A point $x \in \Sigma$ where $(du)_x = 0$ is called a *critical point* of u. We define

$$\operatorname{crit}_u = \{x \in \Sigma \mid (du)_x = 0\},\$$

the set of all critical points of u, also called the *critical locus* of u.

PROPOSITION 3.2. A smooth map $u: \Sigma \to M$ is a Smith map if and only if both of the following conditions hold:

- u is weakly conformal. (This means that $u^*h = \frac{1}{n}|du|^2g$.) $u^*\alpha = \frac{1}{(\sqrt{n})^n}|du|^n \text{vol}$.

We call such maps "conformally calibrating". Thus a map is Smith if and only if it is conformally calibrating.

Proof. This is immediate from Proposition 2.32. \square

Remark 3.3. The alternate characterization of Smith maps in Proposition 3.2 makes it easy to see that *precomposition* of a Smith map $u: \Sigma \to M$ by an orientation preserving weakly conformal map $F: \Sigma \to \Sigma$ yields another Smith map $u \circ F: \Sigma \to M$. See Proposition 3.6.

Note that both of the above conditions are trivially satisfied at any critical point of u. The first condition expresses the fact that $(du)_x: T_x\Sigma \to T_{u(x)}M$ is a conformal injection at any noncritical point.

Smith maps have several basic properties which are immediate from the results of $\S 2.5$.

PROPOSITION 3.4. Let $u: \Sigma \to M$ be a Smith map. Let $\Sigma^{\circ} = \Sigma \backslash \operatorname{crit}_u$ be the open set on which u has no critical points. Then for any $x \in \Sigma^{\circ}$, the image $(du)_x(T_x\Sigma)$ is an n-dimensional subspace of $T_{u(x)}M$ that is calibrated with respect to α_Q . That is, $u(\Sigma^{\circ})$ is an immersed calibrated submanifold of M. Here the orientation of $u(\Sigma^{\circ})$ is the one naturally induced by the injection $(du)_x$ for each $x \in \Sigma^{\circ}$.

Proof. This is immediate from Corollary 2.33. \square

PROPOSITION 3.5. Let Z be a smooth oriented n-manifold, and let (M, h, Q, α_Q) be as above. Let $u: Z \to M$ be a smooth immersion. Give the image u(Z) the orientation induced by the injection $(du)_z$ for each $z \in Z$. If u(Z) is calibrated with respect to α_Q , then let $g = u^*h$, which is a Riemannian metric on Z. With respect to g and the orientation on Z, the map $u: Z \to M$ is a Smith map.

Proof. This is immediate from Proposition 2.38. \square

PROPOSITION 3.6. Let $u: \Sigma \to M$ be a Smith map. Let $F: \Sigma \to \Sigma$ be an orientation preserving conformal diffeomorphism. Then the composition $u \circ F: \Sigma \to M$ is a Smith map.

Proof. This is immediate from Corollary 2.35. \square

Thus the image of any Smith map is a calibrated submanifold (away from the critical points), and any calibrated submanifold is the image of a Smith map, in many different ways, since the Smith equation (36) is conformally invariant in the sense of Proposition 3.6.

Therefore, for those calibrations that correspond to a vector cross product, calibrated submanifolds are in some sense equivalent to Smith maps. We observe that Gray maps also have these two properties with respect to calibrated submanifolds. However, the crucial difference is that the Smith equation is conformally invariant, whereas the Gray equation is not. Conformal invariance is a fundamental feature of those geometric partial differential equations that exhibit phenomena of removable singularities, compactness, and bubbling. This is why Smith maps are much more preferable than Gray maps.

We close this section with another demonstration of this conformal invariance, which is instructive. Let $u: \Sigma \to M$ be a smooth map. Define the *Smith operator* \eth to be the operator that takes u to

$$\eth u = Q(\Lambda^{n-1}du) - \frac{1}{(\sqrt{n})^{n-2}}|du|^{n-2}(du) *.$$

(In [45] this operator is called the multi-Cauchy-Riemann operator.) We observe that $\eth u$ is a section of $\Lambda^{n-1}T^*\Sigma\otimes u^*TM$ and that $\eth u=0$ if and only if u is a Smith map. Let $F:\Sigma\to\Sigma$ be a smooth map. This corresponds to the special case when $(M,h,Q,\alpha_Q)=(\Sigma,g,*,\text{vol})$. In this case the Smith operator applied to F is

$$\eth F = *(\Lambda^{n-1}dF) - \frac{1}{(\sqrt{n})^{n-2}}|dF|^{n-2}(dF) *.$$

Now consider the composition $u \circ F : \Sigma \to M$. Using $d(u \circ F) = (du)(dF)$ and Corollary 2.2 (which holds trivially at critical points) we compute

$$\begin{split} \eth(u \circ F) &= Q \big(\Lambda^{n-1} \big((du)(dF) \big) \big) - \frac{1}{(\sqrt{n})^{n-2}} |(du)(dF)|^{n-2} (du)(dF) * \\ &= Q (\Lambda^{n-1} du) (\Lambda^{n-1} dF) - \frac{1}{(\sqrt{n})^{n-2} (\sqrt{n})^{n-2}} |du|^{n-2} |dF|^{n-2} (du)(dF) * \\ &= Q (\Lambda^{n-1} du) (\Lambda^{n-1} dF) - \frac{1}{(\sqrt{n})^{n-2}} |du|^{n-2} (du) * (\Lambda^{n-1} dF) \\ &\quad + \frac{1}{(\sqrt{n})^{n-2}} |du|^{n-2} (du) * (\Lambda^{n-1} dF) \\ &\quad - \frac{1}{(\sqrt{n})^{n-2} (\sqrt{n})^{n-2}} |du|^{n-2} |dF|^{n-2} (du)(dF) * \\ &= (\eth u) (\Lambda^{n-1} dF) + \frac{1}{(\sqrt{n})^{n-2}} |du|^{n-2} (du)(\eth F). \end{split}$$

Thus we see explicitly that if u is a Smith map and F is also a Smith map (which for F means it is an orientation preserving conformal diffeomorphism) then $u \circ F$ is also a Smith map.

REMARK 3.7. Here is yet another way to see the conformal invariance of the Smith equation (36), which shows explicitly the necessity of the particular power of |du| on the right hand side. Let g be a metric on Σ and let $\widetilde{g}=f^2g$ be another metric in the same conformal class, where f is a positive function. Let $\mathcal{V}_g \in \Lambda^n(T\Sigma)$ denote the volume form on tangent vectors with respect to g. Then $\mathcal{V}_{\widetilde{g}}=f^{-n}\mathcal{V}_g$, and consequently the Hodge star on $\Lambda^k(T\Sigma)$ with respect to \widetilde{g} is $*_{\widetilde{g}}=f^{2k-n}*_g$. Moreover, we also have $|du|_{\widetilde{g}}^2=f^{-2}|du|_g^2$. Putting these all together with k=n-1, we deduce that $|du|_{\widetilde{g}}^{n-2}*_{\widetilde{g}}=|du|_g^{n-2}*_g$ and thus the right hand side of (36) is independent of the conformal class of g.

3.2. Four classes of Smith maps. In this section we consider the four classes of Smith maps, corresponding to the four types of vector cross product Q on M from Table 1. We also consider the Smith equation (36) explicitly in local coordinates for the four classes. In all these cases we have a smooth Smith map $u: (\Sigma, g, *, \mathsf{vol}) \to (M, h, Q, \alpha)$. Define $\Sigma^0 = \Sigma \setminus \mathrm{crit}_u$, where crit_u is the critical locus of u.

EXAMPLE 3.8. Suppose that Q is of type I from Table 1. Then $Q = *_M$ is the Hodge star operator corresponding to the metric h and an orientation on M. Furthermore, $\alpha_Q = \mathsf{vol}_M$ is the associated volume form. Since n-1 = m-1, we have n = m. The Smith equation (36) in this case becomes

$$*_M(\Lambda^{n-1}du) = \frac{1}{(\sqrt{n})^{n-2}}|du|^{n-2}(du) *_{\Sigma}.$$

This equation is equivalent to the pair of equations $u^*h = \frac{1}{n}|du|^2g$ and $u^*\text{vol}_M = \frac{1}{(\sqrt{n})^n}|du|^n\text{vol}_{\Sigma}$. Thus in this case a Smith map $u: \Sigma \to M$ is an orientation preserving weakly conformal diffeomorphism. The image $u(\Sigma^{\circ})$ is an open submanifold of M.

EXAMPLE 3.9. Suppose that Q is of type II from Table 1. Then necessarily m is even and $Q=J_M$ is an orthogonal almost complex structure on (M,h), which need not be integrable. Furthermore, $\alpha_Q=\omega_M$ is the associated Kähler form, which need not be parallel nor even closed. Since n-1=1, we have n=2. Thus $*=J_\Sigma$ is simultaneously of both Type I and Type II, where J_Σ is an orthogonal almost complex structure on (Σ,g) . In particular $(\Sigma,g,*,\mathrm{vol})$ is a complex 1-dimensional Kähler manifold, as a U(1)-structure is always torsion-free. The Smith equation (36) in this case becomes

$$J_M(du) = (du)J_{\Sigma}. (37)$$

Note that since the Smith equation depends only on the conformal class of the metric g on Σ , in this case it really depends only on the underlying Riemann surface (Σ, J_{Σ}) . In particular, in this case a Smith map $u: \Sigma \to M$ is a J-holomorphic map. The image $u(\Sigma^{\circ})$ is a J-holomorphic curve in M, also called a 1-dimensional almost complex submanifold. This Smith equation is the classical Cauchy–Riemann equation.

We remark that if n=2 then Example 3.8 is a special case of Example 3.9 corresponding to m=2.

For our purposes, the two most important Smith maps are the exceptional cases corresponding to the vector cross products of type III and IV from Table 1.

EXAMPLE 3.10. Suppose that Q is of type III from Table 1. Then necessarily n=3, m=7 and Q is a 2-fold vector cross product corresponding to a G₂-structure $\varphi=\alpha_Q$, which is the associated calibration 3-form. This G₂-structure need not be torsion-free. The Smith equation (36) in this case becomes

$$Q(\Lambda^2 du) = \frac{1}{\sqrt{3}} |du|(du) *. \tag{38}$$

The image $u(\Sigma^{\circ})$ is an associative submanifold of M and this Smith equation is called the associative Smith equation.

EXAMPLE 3.11. Suppose that Q is of type IV from Table 1. Then necessarily $n=4,\ m=8$ and Q is a 3-fold vector cross product corresponding to a Spin(7)-structure $\Phi=\alpha_Q$, which is the associated calibration 4-form. This Spin(7)-structure need not be torsion-free. The Smith equation (36) in this case becomes

$$Q(\Lambda^3 du) = \frac{1}{2} |du|^2 (du) *.$$

The image $u(\Sigma^{\circ})$ is a Cayley submanifold of M and this Smith equation is called the Cayley Smith equation.

In Section 3.3 we discuss relations between different classes of Smith maps, which is nontrivial.

We now express these equations in local coordinates. Let (x^1, \ldots, x^n) be local coordinates for Σ and let (y^1, \ldots, y^m) be local coordinates for M. Let $u: \Sigma \to M$

be a smooth map. In terms of these coordinates we have $y^a = y^a(x^1, ..., x^n)$ for $1 \le a \le m$. The section du of $T^*\Sigma \otimes u^*TM$ becomes

$$(du)_x = \frac{\partial u^a}{\partial x^i}\Big|_x dx^i|_x \otimes \frac{\partial}{\partial y^a}\Big|_{u(x)}.$$

This means that $(du)_x \frac{\partial}{\partial x^i}\Big|_x = \frac{\partial u^a}{\partial x^i}\Big|_x \frac{\partial}{\partial y^a}\Big|_{u(x)}$. Write

$$\mathsf{vol}_{\Sigma} = \frac{1}{n!} \mu_{i_1 \cdots i_n} dx^{i_1} \wedge \cdots \wedge dx^{i_n} \qquad \text{and} \qquad \alpha_Q = \frac{1}{n!} \alpha_{b_1 \cdots b_n} dy^{b_1} \wedge \cdots \wedge dy^{b_n}$$

where $1 \leq i_k \leq n$ and $1 \leq b_k \leq m$ for all k. Then by (20) the Hodge star * on $\Lambda^{n-1}T\Sigma$ and the vector cross product Q on $\Lambda^{n-1}TM$ are given by

$$*\left(\frac{\partial}{\partial x^{i_1}}\wedge\cdots\wedge\frac{\partial}{\partial x^{i_{n-1}}}\right) = \mu_{i_1\cdots i_{n-1}j}g^{jl}\frac{\partial}{\partial x^l},$$
$$Q\left(\frac{\partial}{\partial y^{b_1}}\wedge\cdots\wedge\frac{\partial}{\partial y^{b_{n-1}}}\right) = \alpha_{b_1\cdots b_{n-1}c}h^{ca}\frac{\partial}{\partial y^a}.$$

Then the equation (36) becomes

$$\alpha_{b_1 \cdots b_{n-1} c} \frac{\partial u^{b_1}}{\partial x^{i_1}} \cdots \frac{\partial u^{b_{n-1}}}{\partial x^{i_{n-1}}} h^{ca} = \frac{1}{(\sqrt{n})^{n-2}} |du|^{n-2} \frac{\partial u^a}{\partial x^l} \mu_{i_1 \cdots i_{n-1} j} g^{jl}.$$

which simplifies to

$$\alpha_{b_1 \cdots b_{n-1} c} \frac{\partial u^{b_1}}{\partial x^{i_1}} \cdots \frac{\partial u^{b_{n-1}}}{\partial x^{i_{n-1}}} = \frac{1}{(\sqrt{n})^{n-2}} |du|^{n-2} \frac{\partial u^b}{\partial x^l} \mu_{i_1 \cdots i_{n-1} j} g^{jl} h_{bc}, \tag{39}$$

where |du| is given in the coordinates by

$$|du|^2 = \frac{\partial u^a}{\partial r^i} \frac{\partial u^b}{\partial r^j} h_{ab} g^{ij}. \tag{40}$$

For the cases of type II, III, IV in Examples 3.9, 3.10, 3.11, respectively, equation (39) becomes

$$\omega_{ab} \frac{\partial u^a}{\partial x^i} = \frac{\partial u^a}{\partial x^l} \mu_{ij} g^{jl} h_{ab}, \qquad \text{(classical Cauchy-Riemann equation)},$$

$$\varphi_{abc} \frac{\partial u^a}{\partial x^i} \frac{\partial u^b}{\partial x^j} = \frac{1}{\sqrt{3}} |du| \frac{\partial u^a}{\partial x^l} \mu_{ijk} g^{kl} h_{ac} \qquad \text{(associative Smith equation)},$$

$$\Phi_{abcd} \frac{\partial u^a}{\partial x^i} \frac{\partial u^b}{\partial x^j} \frac{\partial u^c}{\partial x^k} = \frac{1}{2} |du|^2 \frac{\partial u^a}{\partial x^p} \mu_{ijkl} g^{lp} h_{ad} \qquad \text{(Cayley Smith equation)},$$

where μ is the Riemannian volume form on (Σ, g) and |du| is given by (40).

- **3.3. Relations between classes of Smith maps.** Let us recall the following two product constructions of associative and Cayley submanifolds.
- (A.) Let $(Y^6, J, \omega, \Upsilon)$ be a Calabi-Yau 3-fold with Kähler form ω and holomorphic volume form Υ . Define $M^7 = Y^6 \times S^1$ and

$$\varphi = \operatorname{Re}(\Upsilon) + d\theta \wedge \omega \tag{41}$$

where θ is the standard periodic "coordinate" on S^1 . Then φ is a torsion-free G₂-structure on M^7 . Moreover, if Σ^2 is a complex submanifold of Y^6 then $\Sigma^2 \times S^1$ is an associative submanifold of M^7 .

(B.) Let (Y^7, φ) be a torsion-free G_2 manifold with associative 3-form φ and Hodge dual coassociative 4-form $\psi = *_{\varphi} \varphi$. Define $M^8 = Y^7 \times S^1$ and $\Phi = d\theta \wedge \varphi + \psi$ where θ is as before. Then Φ is a torsion-free Spin(7)-structure on M^8 . Moreover, if Σ^3 is an associative submanifold of Y^7 , then $\Sigma^3 \times S^1$ is a Cayley submanifold of M^8 .

In light of the relationships (A.) and (B.) between complex, associative, and Cayley submanifolds, it is natural to ask whether there exist analogous relationships between Smith maps of types II, III, and IV. In this section, we focus on the relationship between Smith maps of types II and III. We use this discussion in §4.5 to clarify the (non-)relation between bubbling of J-holomorphic curves and bubbling of associative Smith maps.

To begin, let $v: \Sigma^2 \to Y^6$ be a Smith map of type II as in Example 3.9, where Σ^2 is a Riemann surface equipped with a conformal class of metrics $[g_2]$ and volume form vol_2 , and where Y^6 is a Calabi-Yau 3-fold with Riemannian metric h_6 and data (J, ω, Υ) as above. By Proposition 3.2, we have

$$v^*h_6 = \mu^2 g_2, \qquad v^*\omega = \mu^2 \text{vol}_2.$$
 (42)

where we are writing

$$\mu^2 = \frac{1}{2} |dv|^2 \tag{43}$$

for ease of notation, and where $|dv|^2$ is computed with respect to g_2 on Σ and h_6 on Y. We remark that by the first equation in (42), or by direct computation, the expression $|dv|^2g_2$ is independent of the choice of representative metric in the conformal class $[g_2]$.

Next, equip $M^7 = Y^6 \times S^1$ with the G₂-structure $\varphi = \text{Re}(\Upsilon) + d\theta \wedge \omega$ as above. The induced Riemannian metric h_7 on M^7 is simply the product metric

$$h_7 = h_6 + (d\theta)^2 \tag{44}$$

where θ denotes the angle coordinate in $Y^6 \times S^1$. Since the image $v(\Sigma)$ is a complex curve in Y, it follows that $v(\Sigma) \times S^1$ is an associative submanifold of $M^7 = Y^6 \times S^1$.

Finally, we equip the 3-manifold $\Sigma \times S^1$ with a Riemannian metric g_3 , and let vol_3 denote the corresponding volume form. We let ϕ denote the angle coordinate for the S^1 factor of $\Sigma \times S^1$.

We may now consider maps $u: \Sigma \times S^1 \to Y \times S^1$ of the form

$$u(x,\phi) = (v(x), f(x,\phi)) \tag{45}$$

where $f: \Sigma \times S^1 \to S^1$ is a smooth function $\theta = f(x, \phi)$. For simplicity, we suppose that

$$f' := \frac{\partial f}{\partial \phi}$$

nonvanishing, so that u as defined in (45) is an immersion whenever v is an immersion. The particular form (45) of u is motivated by the desire to have the image $u(\Sigma \times S^1)$ be an open subset of the associative submanifold $v(\Sigma) \times S^1$. Note that the choice $f(z, \phi) = \phi$ is permissible, which corresponds to $u = v \times \operatorname{Id}_{S^1}$, but for later use in §4.5 we need to consider the general form in (45). See Remark 4.16.

We aim to understand when such maps u are associative Smith. To this end, note that using (44), followed by the first part of (42), we see that

$$u^*h_7 = u^*(h_6 + (d\theta)^2) = v^*h_6 + (u^*d\theta)^2 = \mu^2g_2 + (d(u^*\theta))^2 = \mu^2g_2 + (df)^2.$$
 (46)

Similarly, using (41), followed by the second part of (42), we see that

$$u^*\varphi = u^*(\operatorname{Re}(\Upsilon) + d\theta \wedge \omega) = v^*\operatorname{Re}(\Upsilon) + (d(u^*\theta)) \wedge v^*\omega$$
$$= 0 + df \wedge (\mu^2\operatorname{vol}_2)$$
$$= \mu^2 f' d\phi \wedge \operatorname{vol}_2$$
(47)

using that $df = f'd\phi + d_{\Sigma}f$ and $(d_{\Sigma}f) \wedge \mathsf{vol}_2 = 0$. Finally, to streamline notation, we write

$$\lambda^2 = \frac{1}{3}|du|^2\tag{48}$$

where $|du|^2$ is computed with respect to g_3 on $\Sigma \times S^1$ and h_7 on $Y \times S^1$.

PROPOSITION 3.12. The map u of (45) is an associative Smith map if and only if the metric g_3 and volume form vol_3 on $\Sigma \times S^1$ satisfy

$$\lambda^2 g_3 = \mu^2 g_2 + (df)^2 \tag{49}$$

and

$$\lambda^3 \text{vol}_3 = \mu^2 f' d\phi \wedge \text{vol}_2. \tag{50}$$

Proof. By Proposition 3.2, the map u is associative Smith if and only if it satisfies

$$u^*h_7 = \lambda^2 g_3, \qquad u^*\varphi = \lambda^3 \text{vol}_3. \tag{51}$$

The result now follows from (46) and (47). \square

3.4. The fundamental energy identity for Smith maps. In this section we establish the fundamental energy identity for Smith maps, when Σ is compact and the calibration form $\alpha = \alpha_Q$ on M is closed. We explain why $d\alpha = 0$ is necessary for the energy identity, which says that the L^n -energy of a Smith map is in some sense a topological invariant when $\partial \Sigma = \emptyset$.

The energy identity, which is our Theorem 3.14 below, originally appeared in the unpublished preprint [45, Proposition 6.5]. The way we present it here, the proof is almost immediate due to our parcelling out the preliminary linear algebraic results in §2.

Let Σ^n and M^m be as in §3.1, and let u be any smooth map. Further, throughout this section we suppose that Σ is *compact*, so that we may integrate over Σ .

DEFINITION 3.13. The *n*-energy of u, denoted $E_n(u)$, is defined to be

$$E_n(u) = \frac{1}{(\sqrt{n})^n} \int_{\Sigma} |du|^n \text{vol.}$$

Thus, up to a factor which is chosen for later convenience, we observe that $E_n(u)$ is essentially $||du||_n^n$, where $||du||_n$ is the L^n -norm of u.

Theorem 3.14 (The energy identity for Smith maps). When the n-energy E_n is written as

$$E_n(u) = \int_{\Sigma} \left(\frac{1}{(\sqrt{n})^n} |du|^n \text{vol} - u^* \alpha \right) + \int_{\Sigma} u^* \alpha,$$

then the first term is always nonegative and vanishes if and only if u is a Smith map.

Moreover, if $d\alpha = 0$ and $\partial \Sigma = \emptyset$, then the second term is a topological invariant, depending only on the cohomology class $[\alpha] \in H^n(M,\mathbb{R})$ of α in M and on the homology class $u_*[\Sigma] \in H_n(M)$, which is the image of the fundamental class $[\Sigma] \in H_n(\Sigma)$ by the pushforward homomorphism $u_* : H_n(\Sigma) \to H_n(M)$ induced by the map $u : \Sigma \to M$.

Consequently, when $d\alpha = 0$, a smooth map $u : \Sigma \to M$ is a Smith map if and only if its n-energy is given by

$$E_n(u) = \int_{\Sigma} u^* \alpha = [\alpha] \cdot u_*[\Sigma].$$

Proof. The generalized calibration inequality Theorem 2.40 applied to A=du says that

$$\frac{1}{(\sqrt{n})^n}|du|^n \text{vol} - u^*\alpha \ge 0 \tag{52}$$

with equality if and only if $u: \Sigma \to M$ is a Smith map. If $d\alpha = 0$, then by Stokes's Theorem, $\int_{\Sigma} u^* \alpha = [\alpha] \cdot u_*[\Sigma]$ depends only on the cohomology class $[\alpha] \in H^n(M, \mathbb{R})$. \square

Remark 3.15. The first part of Theorem 3.14 actually holds for maps $u \in W^{1,n}(\Sigma;M)$ since in this case the inequality (52) holds almost everywhere on Σ and vanishes almost everywhere if it integrates to zero. Recall that a map $u \in W^{1,n}(\Sigma;M)$ is said to be a Smith map if its weak derivative du satisfies the Smith equation almost everywhere on Σ .

COROLLARY 3.16. Let $d\alpha = 0$. If a Smith map $u : \Sigma \to M$ exists, then it is an absolute minimizer of the n-energy E_n amongst all smooth maps $v : \Sigma \to M$ representing the same homology class in $H_n(M)$ as u.

Proof. This is immediate from Theorem 3.14. \square

Of course, when n = 2, so that we are in the case of Type II from Table 1, then this result is classical in the theory of J-holomorphic maps, and can be found, for example, in McDuff–Salamon [35].

REMARK 3.17. A consequence of Theorem 3.14 and Proposition 2.32 is that if $u: \Sigma \to M$ is a Smith map with no critical points, then the n-energy $E_n(u)$ equals the volume of Σ with respect to the metric $u^*h = \frac{1}{n}|du|^2g$ on Σ . In particular, if u is a Gray map with no critical points, then $E_n(u)$ equals the volume of (Σ, g) . (See also Proposition 3.24 and Remark 3.25 for the geometric measure theory analogue of this in the setting of currents.)

3.5. Smith maps and *n*-harmonic maps. The theory of *p*-harmonic maps has been extensively studied. Some possible references for this theory (certainly not exhaustive) include [10, 15, 22, 36, 38, 46]. Let (Σ, g) and (M, h) be Riemannian manifolds. A smooth map $u: \Sigma \to M$ is *p*-harmonic if is satisfies

$$\operatorname{div}(|du|^{p-2}du) = 0, (53)$$

where div is the Riemannian divergence on (M,g) taking a section of $T^*\Sigma \otimes u^*TM$ to a section of u^*TM . When Σ is compact, the *p*-harmonic maps are the critical points of the *p*-energy $\int_{\Sigma} |du|^p \text{vol}$.

Suppose that $(\Sigma^n, g, *, \mathsf{vol})$ and (M^m, h, Q, α_Q) are as in §3.1 and suppose that $d\alpha = 0$, where we write α for α_Q . Then by Corollary 3.16, since a Smith map $u: \Sigma \to M$ is an absolute minimizer of the n-energy in its homology class, it is certainly a critical point, and thus it is an n-harmonic map. (Here $p = n = \dim \Sigma$.)

In this section we show explicitly that Smith maps satisfy (53) when $d\alpha = 0$. This argument is interesting because we cannot just differentiate the Smith equation to obtain the *n*-harmonic map equation. Rather, we also need to use the fact that a Smith map is conformally calibrating.

LEMMA 3.18. Let Q be a k-fold vector cross product on (M,h) with associated calibration (k+1)-form α . Let ∇ be the Levi-Civita connection of h. Then we have

$$(\nabla_V \alpha)(W_1, \dots, W_{k+1}) = h((\nabla_V Q)(W_1, \dots, W_k), W_{k+1}),$$
 (54)

$$h((\nabla_V Q)(W_1, \dots, W_k), Q(W_1, \dots, W_k)) = 0,$$
 (55)

for all smooth vector fields V, W_1, \ldots, W_{k+1} on M.

Proof. The metric h is parallel with respect to ∇ . Equation (54) is a consequence of applying ∇_V to both sides of

$$\alpha(W_1,\ldots,W_{k+1}) = h(Q(W_1,\ldots,W_k),W_{k+1}),$$

which is the definition of α from Q as in (20). Similarly (55) is obtained by applying ∇_V to both sides of

$$h(Q(W_1,\ldots,W_k),Q(W_1,\ldots,W_k)) = h(W_1 \wedge \cdots \wedge W_k,W_1 \wedge \cdots \wedge W_k),$$

which is the fundamental property (10) of a vector cross product. \square

COROLLARY 3.19. Let $u: \Sigma \to M$ be a smooth Smith map as in §3.1. Then we have $u^*(\nabla_V \alpha) = 0$ for any smooth vector field V on M.

Proof. Fix $x \in \Sigma$ and let e_1, \ldots, e_n be an oriented orthonormal frame for $T_x\Sigma$. Since u is a Smith map, it follows from Proposition 2.32 that u_*e_1, \ldots, u_*e_n are all orthogonal and of the same length $\lambda = \frac{1}{\sqrt{n}} |(du)_x|$, and that they span a calibrated subspace $L = u_*(T_x\Sigma)$ of $T_{u(x)}M$. Then the vectors $f_i = \frac{1}{\lambda}u_*e_i$ for $i = 1, \ldots, n$ form an oriented orthonormal basis of L. By Proposition 2.18 we deduce that

$$f_n = Q(f_1, \dots, f_{n-1}).$$

Using the above equation as well as (54) and (55), we compute

$$\lambda^{-n}(u^*\nabla_V \alpha)(e_1, \dots, e_n) = \lambda^{-n}(\nabla_V \alpha)(u_*e_1, \dots, u_*e_n)$$

$$= (\nabla_V \alpha)(f_1, \dots, f_{n-1}, f_n)$$

$$= h((\nabla_V Q)(f_1, \dots, f_{n-1}), f_n)$$

$$= h((\nabla_V Q)(f_1, \dots, f_{n-1}), Q(f_1, \dots, f_{n-1}))$$

$$= 0,$$

which is equivalent to $u^*(\nabla_V \alpha) = 0$. \square

Now we show that a Smith map is *n*-harmonic if $d\alpha = 0$. Because the Hodge star * on $\Lambda^{n-1}T\Sigma$ is invertible, with $*^{-1} = (-1)^{n-1}*$, the Smith equation (36) is equivalent to

$$Q(\Lambda^{n-1}du)* = (-1)^{n-1} \frac{1}{(\sqrt{n})^{n-2}} |du|^{n-2} (du).$$

Here both sides of the above equation are sections of $T^*\Sigma \otimes u^*TM$. Thus, in order to show that a Smith map satisfies the *n*-harmonic map equation (53), we need to show that $\operatorname{div}(Q(\Lambda^{n-1}du)*) = 0$.

Proposition 3.20. If $d\alpha = 0$, then $\operatorname{div}(Q(\Lambda^{n-1}du) *) = 0$, so any Smith map is an n-harmonic map.

Proof. First consider an arbitrary section B of $T^*\Sigma \otimes u^*TM$. Using local coordinates as in the end of §3.2 we can write $B = B_j^a dx^j \otimes \left(\frac{\partial}{\partial y^a} \circ u\right)$, which means that

$$B_x = B_j^a(x) dx^j|_x \otimes \frac{\partial}{\partial y^a}\Big|_{u(x)}.$$

Now we consider the particular case $B = Q(\Lambda^{n-1}du) *$, where $*: T\Sigma \to \Lambda^{n-1}T\Sigma$. Writing $\text{vol} = \frac{1}{n!}\mu_{i_1\cdots\mu_{i_n}}dx^{i_1}\wedge\cdots\wedge dx^{i_n}$ and $\alpha = \frac{1}{n!}\alpha_{b_1\cdots b_n}dy^{b_1}\wedge\cdots\wedge dy^{b_n}$, we can compute that

$$B_j^a = \frac{1}{(n-1)!} \mu_{i_1 \cdots i_{n-1} j} g^{i_1 l_1} \cdots g^{i_{n-1} l_{n-1}} \frac{\partial u^{b_1}}{\partial x^{l_1}} \cdots \frac{\partial u^{b_{n-1}}}{\partial x^{l_{n-1}}} \alpha_{b_1 \cdots b_{n-1} c} h^{ca}.$$
 (56)

The connection ∇ on $T^*\Sigma \otimes u^*TM$ is the tensor product of the Levi-Civita connection of g on $T^*\Sigma$ and the pullback by u of the Levi-Civita connection of h on TM. Using the fact that g, h, and vol are all parallel, applying $g^{ij}\nabla_i$ to (56) gives

$$(\operatorname{div} B)^{a} = g^{ij}(\nabla_{i}B)_{j}^{a}$$

$$= \frac{1}{(n-1)!} \mu_{i_{1}\cdots i_{n-1}j} g^{ij} g^{i_{1}l_{1}} \cdots g^{i_{n-1}l_{n-1}} h^{ca} \nabla_{i} \left(\frac{\partial u^{b_{1}}}{\partial x^{l_{1}}} \cdots \frac{\partial u^{b_{n-1}}}{\partial x^{l_{n-1}}} \alpha_{b_{1}\cdots b_{n-1}c}\right).$$

If we choose (x^1, \ldots, x^n) to be Riemannian normal coordinates for g centred at $x \in \Sigma$, and (y^1, \ldots, y^m) to be Riemannian normal coordinates for h centred at $u(x) \in M$, then the covariant derivative ∇_i evaluated at the point x is the same as the partial derivative $\frac{\partial}{\partial x^i}$ at x, since the Christoffel symbols vanish. Thus, at x, we have $(\operatorname{div} B)^a$

$$\begin{split} &=\frac{1}{(n-1)!}\mu_{i_1\cdots i_{n-1}j}g^{ij}g^{i_1l_1}\cdots g^{i_{n-1}l_{n-1}}h^{ca}\Big(\sum_{k=1}^{n-1}\frac{\partial^2 u^{b_k}}{\partial x^{l_k}\partial x^i}\frac{\partial u^b_{i_1}}{\partial x^{l_1}}\cdots\widehat{\frac{\partial u^b_{i_k}}{\partial x^{l_k}}}\cdots\frac{\partial u^{b_{n-1}}}{\partial x^{l_{n-1}}}\alpha_{b_1\cdots b_{n-1}c}\Big)\\ &+\frac{1}{(n-1)!}\mu_{i_1\cdots i_{n-1}j}g^{ij}g^{i_1l_1}\cdots g^{i_{n-1}l_{n-1}}h^{ca}\frac{\partial u^{b_1}}{\partial x^{l_1}}\cdots\frac{\partial u^{b_{n-1}}}{\partial x^{l_{n-1}}}\Big(\nabla_{\frac{\partial}{\partial x^i}}\alpha_{b_1\cdots b_{n-1}c}\Big). \end{split}$$

Since $\frac{\partial^2 u^{b_k}}{\partial x^{l_k} \partial x^i}$ is symmetric in l_k, i and $\mu_{i_1 \cdots i_{n-1} j}$ is skew in i_k, j , all the terms in the first line above vanish. For the covariant derivative of α , we use $\nabla_{\frac{\partial}{\partial x^i}} = \frac{\partial u^{b_n}}{\partial x^i} \nabla_{\frac{\partial}{\partial y^{b_n}}}$, which we write as $\frac{\partial u^{b_n}}{\partial x^i} \nabla_{b_n}$ for simplicity. We thus have that at the point x,

$$(\operatorname{div} B)^{a} = \frac{1}{(n-1)!} \mu_{i_{1} \cdots i_{n-1} j} g^{ij} g^{i_{1} l_{1}} \cdots g^{i_{n-1} l_{n-1}} h^{ca} \frac{\partial u^{b_{1}}}{\partial x^{l_{1}}} \cdots \frac{\partial u^{b_{n-1}}}{\partial x^{l_{n-1}}} \frac{\partial u^{b_{n}}}{\partial x^{i}} \Big(\nabla_{b_{n}} \alpha_{b_{1} \cdots b_{n-1} c} \Big).$$

Relabel $j \to i_n$ and $i \to l_n$. We have

$$(\operatorname{div} B)^a = \frac{1}{(n-1)!} \mu_{i_1 \cdots i_n} g^{i_1 l_1} \cdots g^{i_n l_n} h^{ca} \frac{\partial u^{b_1}}{\partial x^{l_1}} \cdots \frac{\partial u^{b_n}}{\partial x^{l_n}} \Big(\nabla_{b_n} \alpha_{b_1 \cdots b_{n-1} c} \Big).$$

It follows by the skew-symmetry of $\mu_{i_1\cdots i_n}$ that if we interchange any two of b_1,\ldots,b_n in the last factor $\nabla_{b_n}\alpha_{b_1\cdots b_{n-1}c}$, then the right hand side above will change sign. Consequently, we can write

$$(\operatorname{div} B)^{a} = \frac{1}{(n-1)!} \mu_{i_{1}\cdots i_{n}} g^{i_{1}l_{1}} \cdots g^{i_{n}l_{n}} h^{ca} \frac{\partial u^{b_{1}}}{\partial x^{l_{1}}} \cdots \frac{\partial u^{b_{n}}}{\partial x^{l_{n}}} \left(\frac{1}{n} \sum_{k=1}^{n} \nabla_{b_{k}} \alpha_{b_{1}\cdots b_{k-1}cb_{k+1}\cdots b_{n}}\right).$$
 (57)

Now we use the hypothesis that $d\alpha = 0$. Since d is the skew-symmetrization of ∇ , we have

$$0 = (d\alpha)_{cb_1 \cdots b_n} = \nabla_c \alpha_{b_1 \cdots b_n} - \sum_{k=1}^n \nabla_{b_k} \alpha_{b_1 \cdots b_{k-1} cb_{k+1} \cdots b_n}.$$

Substituting the above into (57) gives

$$(\operatorname{div} B)^{a} = \frac{1}{n!} \mu_{i_{1} \cdots i_{n}} g^{i_{1} l_{1}} \cdots g^{i_{n} l_{n}} h^{ca} \frac{\partial u^{b_{1}}}{\partial x^{l_{1}}} \cdots \frac{\partial u^{b_{n}}}{\partial x^{l_{n}}} \nabla_{c} \alpha_{b_{1} \cdots b_{n}}.$$
 (58)

We observe that Corollary 3.19 with $V = \frac{\partial}{\partial y^c}$ evaluated on W_1, \dots, W_n with $W_k = \frac{\partial}{\partial x^{l_k}}$ says that

$$0 = (u^* \nabla_c \alpha) \left(\frac{\partial}{\partial x^{l_1}}, \dots, \frac{\partial}{\partial x^{l_n}} \right) = \frac{\partial u^{b_1}}{\partial x^{l_1}} \cdots \frac{\partial u^{b_n}}{\partial x^{l_n}} \nabla_c \alpha_{b_1 \cdots b_n}.$$

Substituting the above into (58) gives div B=0, as claimed. \square

3.6. Smith maps via currents. Most of this section can be read independently from the rest of the paper, with the exception of Corollary 3.27, which is used later. We give two proofs of Corollary 3.27, one using the geometric measure theory framework of this section, and another using the energy identity of Theorem 3.14. The reader who is not interested in geometric measure theory can skip this entire section with the exception of Corollary 3.27 and its second proof.

Here we continue using the notation of §3.1, and we assume that the *n*-form α associated with the vector cross product Q on M is closed. Also, we emphasize that we allow Σ to have *nonempty boundary*.

Recall that in §3.1 and §3.4 the following result is effectively proved.

Proposition 3.21. Suppose Σ is compact, possibly with boundary, and let $u:\Sigma\to M$ be a Lipschitz map. Then

$$\int_{\Sigma} u^* \alpha \le \frac{1}{(\sqrt{n})^n} \int_{\Sigma} |du|^n \text{vol.}$$
 (59)

Moreover, we have the following two equivalent characterizations for when equality holds:

- (a) u is a Smith map.
- (b) u is conformally calibrating; that is, $u^*h = \frac{1}{n}|du|^2g$ and $u^*\alpha = \frac{1}{(\sqrt{n})^n}|du|^n \text{vol}$.

The main purpose of this section is to complete the inequality (59) by showing that it is part of a longer string of inequalities which involves the volume of the image of Σ under u, which has to be understood as a *current* when crit_u is nonempty. In addition, we show that, in the equality case, the image current is calibrated by α . Below we very briefly recall some relevant definitions from the theory of currents. For a general introduction to the subject, see for instance [44, Chapter 6].

Following standard practice in geometric measure theory, we assume that Σ and M are isometrically embedded into \mathbb{R}^l and \mathbb{R}^d , respectively, in which case the n-dimensional Hausdorff measure H^n restricts to the volume measure vol on Σ . Next, to Σ we associate an integral current $[\![\Sigma]\!]$ given by

$$\llbracket \Sigma \rrbracket(\omega) := \int_{\Sigma} \langle \omega_x, \xi_x \rangle dH^n(x) \quad \text{ for all } \omega \in \mathcal{D}^n(\mathbb{R}^l),$$

where $\mathcal{D}^n(\mathbb{R}^l)$ denotes the space of all smooth compactly supported *n*-forms on \mathbb{R}^l , and $\xi_x \in \Lambda^n \mathbb{R}^l$ is the unit simple *n*-vector giving the orientation on $T_x \Sigma$ for each $x \in \Sigma$. (Here $T_x \Sigma$ is considered a subspace of \mathbb{R}^l via the embedding $\Sigma \to \mathbb{R}^l$.)

Given a Lipschitz map $u: \Sigma \to M$, we regard it as a map into \mathbb{R}^d and define the pushforward $u_{\#} \llbracket \Sigma \rrbracket$ of $\llbracket \Sigma \rrbracket$ by u, which is an integral current supported in M, by

$$u_{\#}\llbracket\Sigma\rrbracket(\eta) := \int_{\Sigma} \langle \eta_{u(x)}, \Lambda^n(du)_x \xi_x \rangle dH^n(x) \quad \text{ for all } \eta \in \mathcal{D}^n(\mathbb{R}^d).$$

Note that the right hand side is just $\llbracket \Sigma \rrbracket (u^* \eta)$ when u is smooth. Also, strictly speaking, on the right we should write \overline{u} instead u, where $\overline{u} : \mathbb{R}^l \to \mathbb{R}^d$ is any compactly supported extension of u. (The choice of extension does not affect the definition of $u \# \llbracket \Sigma \rrbracket$.)

The mass of the current $u_{\#}[\![\Sigma]\!]$ is, by definition,

$$\mathbf{M}(u_{\#}[\![\Sigma]\!]) = \sup \left\{ u_{\#}[\![\Sigma]\!](\eta) \mid \eta \in \mathcal{D}^{n}(\mathbb{R}^{d}), \|\eta_{y}\| \leq 1 \text{ for all } y \in \mathbb{R}^{d} \right\},$$

where $\|\cdot\|$ denotes the comass norm of a covector.

To pair the current $u_{\#}[\![\Sigma]\!]$ with the calibrating *n*-form α , we need to extend the latter to a compactly supported form on \mathbb{R}^d . Specifically, take δ_0 sufficiently small so that the neighborhood

$$\mathcal{N}_{2\delta_0}(M) := \{ x \in \mathbb{R}^d \mid \operatorname{dist}_{\mathbb{R}^d}(x, M) < 2\delta_0 \}$$

is strictly contained in a tubular neighborhood of M in \mathbb{R}^d , and let $\pi: \mathcal{N}_{2\delta_0}(M) \to M$ be the nearest-point projection. In addition, fix a cutoff function ζ which is identically 1 on $\mathcal{N}_{\frac{3\delta_0}{2}}(M)$ and vanishes outside of $\mathcal{N}_{2\delta_0}(M)$.

LEMMA 3.22. The n-form $\tilde{\alpha}$ on \mathbb{R}^d defined by $\tilde{\alpha} = \zeta \pi^* \alpha$ has the following properties.

- (a) $d\tilde{\alpha} = 0$ on $\mathcal{N}_{\frac{3\delta_0}{2}}(M)$.
- (b) For $y \in \mathbb{R}^d$, the comass norm of $\tilde{\alpha}_y$ satisfies $\|\tilde{\alpha}_y\| \leq 1$.
- (c) For $y \in M$ and $\tau \in \Lambda^n T_y M$, we have $\langle \tilde{\alpha}_y, \tau \rangle = \langle \alpha_y, \tau \rangle$, where on the left hand side we view τ as an element of $\Lambda^n \mathbb{R}^d$ via the embedding $M \to \mathbb{R}^d$.

Proof. Statements (a) and (c) can be verified by direct computation. For (b), take a unit simple *n*-vector $\tau \in \bigwedge^n \mathbb{R}^d$. Then by the definition of $\tilde{\alpha}$ we have

$$\begin{aligned} |\langle \tilde{\alpha}_{y}, \tau \rangle| &= \zeta(y) \left| \langle \alpha_{\pi(y)}, \Lambda^{n}(d\pi)_{y} \tau \rangle \right| \\ &\leq \|\alpha_{\pi(y)}\| \left| \Lambda^{n}(d\pi)_{y} \tau \right| \leq |\Lambda^{n}(d\pi)_{y} \tau|, \end{aligned}$$

where we used that fact that $\|\alpha_{\pi(y)}\| \leq 1$ in the last inequality. Therefore, taking the supremum over all unit simple n-vectors τ , we obtain

$$\|\tilde{\alpha}_y\| \le |\Lambda^n (d\pi)_y| \le 1,$$

which implies the desired estimate. \Box

LEMMA 3.23. Let $u: \Sigma \to M$ be a Lipschitz map. For any $x \in \Sigma$ such that $(du)_x$ exists, and any $\eta \in \mathcal{D}^n(\mathbb{R}^d)$ with $\|\eta_y\| \leq 1$ for all $y \in M$, we have

$$\langle \eta_{u(x)}, \Lambda^n(du)_x \xi_x \rangle \le |\Lambda^n(du)_x \xi_x| \le \frac{1}{(\sqrt{n})^n} |(du)_x|^n.$$

Proof. This is immediate from the proof of the generalized calibration inequality, Theorem 2.40. \Box

We now give the precise version of the main result of this section along with two corollaries.

Proposition 3.24. Let $u: \Sigma \to M$ be a Lipschitz map.

(a) We have

$$u_{\#}\llbracket\Sigma\rrbracket(\tilde{\alpha}) \leq \mathbf{M}(u_{\#}\llbracket\Sigma\rrbracket) \leq \int_{\Sigma} |\Lambda^{n}(du)_{x}\xi_{x}| dH^{n}(x) \leq \frac{1}{(\sqrt{n})^{n}} \int_{\Sigma} |du|^{n} dH^{n}. \quad (60)$$

(b) If u is a Smith map, then all the inequalities in (60) become equalities. In particular, the integral current $u_{\#}[\![\Sigma]\!]$ is calibrated by $\tilde{\alpha}$. That is, it is a positive $\tilde{\alpha}$ -current in the sense of [23, Section II.4].

Remark 3.25. The third term in (60) coincides with the mass of the pushforward of Σ as a varifold via the map u.

Proof of Proposition 3.24. We begin with part (a). Applying Lemma 3.23 to any $\eta \in \mathcal{D}^n(\mathbb{R}^d)$ with $\|\eta_y\| \leq 1$ everywhere, integrating over Σ , and then taking the supremum over all such η , we see that

$$\mathbf{M}(u_{\#}\llbracket \Sigma \rrbracket) \leq \int_{\Sigma} |\Lambda^n(du)_x \xi_x| dH^n(x) \leq \frac{1}{(\sqrt{n})^n} \int_{\Sigma} |du|^n dH^n.$$

To finish the proof of (a), we simply note that $u_{\#}[\![\Sigma]\!](\tilde{\alpha}) \leq \mathbf{M}(u_{\#}[\![\Sigma]\!])$ by the definition of $\mathbf{M}(u_{\#}[\![\Sigma]\!])$, because $\|\tilde{\alpha}_y\| \leq 1$ for all $y \in \mathbb{R}^d$.

The first claim of part (b) follows directly from Proposition 3.21. The second part holds because any integral current T supported in M and satisfying $T(\tilde{\alpha}) = \mathbf{M}(T)$ must be a positive $\tilde{\alpha}$ -current. \square

COROLLARY 3.26. Let A, B be compact subsets of Σ with smooth boundary and let $u: A \to M$ and $v: B \to M$ be Lipschitz maps, with u being a Smith map. Suppose further that $u_{\#}[\![A]\!] + v_{\#}[\![B]\!] = \partial S$ for some (n+1)-current supported in M. Then

$$\int_{A} |du|^{n} dH^{n} \le \int_{B} |dv|^{n} dH^{n}.$$

Proof of Corollary 3.26. By Proposition 3.24(b), the current $u_{\#}[\![A]\!]$ is calibrated by $\tilde{\alpha}$ and hence is mass-minimizing among all currents homologous to it. Since $-v_{\#}[\![B]\!]$ is homologous to $u_{\#}[\![A]\!]$ by assumption, we get

$$\mathbf{M}(u_{\#}[\![A]\!]) \leq \mathbf{M}(-v_{\#}[\![B]\!]) = \mathbf{M}(v_{\#}[\![B]\!]) \leq \frac{1}{(\sqrt{n})^n} \int_{B} |dv|^n dH^n,$$

where the last inequality follows from (60). We complete the proof by noting that

$$\mathbf{M}(u_{\#}[\![A]\!]) = \frac{1}{(\sqrt{n})^n} \int_A |du|^n dH^n,$$

because of Proposition 3.24(b). \square

The following result is used in §5.3 to derive an estimate crucial to the proofs of Theorems 5.1 and 5.2. As mentioned at the beginning of this section, we give two proofs, the second of which depends only on Theorem 3.14 and does not use the geometric measure theory framework above.

Corollary 3.27. Suppose $\Sigma = \partial W$ where W is a compact oriented Riemannian (n+1)-manifold isometrically embedded in \mathbb{R}^l , and let $u:A\to M$ be a Lipschitz Smith map on a compact set $A\subseteq \Sigma$ with smooth boundary. Suppose moreover that u has a Lipschitz extension $f:\Sigma\to M$, which extends further to a Lipschitz map $F:W\to M$. Then we have

$$\frac{1}{(\sqrt{n})^n} \int_A |du|^n dH^n \le \frac{1}{(\sqrt{n})^n} \int_{\Sigma \setminus A} |df|^n dH^n.$$

Proof of Corollary 3.27. This follows at once from Corollary 3.26 with A being the same, $B = \overline{\Sigma} \setminus \overline{A}$, and $v = f|_B$, and $S = F_{\#}[W]$. (Note that $\partial S = f_{\#}[\Sigma] = u_{\#}[A] + v_{\#}[B]$.)

We also give a second proof, using the energy identity of Theorem 3.14. By the energy identity applied to the calibration $-\alpha$ and the map f with domain $\Sigma \setminus A$, we have

$$\frac{1}{(\sqrt{n})^n} \int_{\Sigma \backslash A} |df|^n dH^n \ge -\int_{\Sigma \backslash A} f^* \alpha.$$

On the other hand, recalling that $f|_A = u$ is a Smith map and using the equality case of the energy identity (with the calibration α), we get

$$\frac{1}{(\sqrt{n})^n} \int_A |df|^n dH^n = \int_A f^* \alpha = \int_{\Sigma} f^* \alpha - \int_{\Sigma \setminus A} f^* \alpha.$$

Combining the two relations above, we arrive at

$$\frac{1}{(\sqrt{n})^n} \int_A |du|^n dH^n \le \int_{\Sigma} f^* \alpha + \frac{1}{(\sqrt{n})^n} \int_{\Sigma \setminus A} |df|^n dH^n.$$

To finish, it suffices to prove that $\int_{\Sigma} f^* \alpha = 0$. For that, note that by Stokes's theorem we have

$$\int_{\Sigma} f^* \alpha = \int_{\Sigma} F^* \alpha = \int_{W} F^* (d\alpha) = 0,$$

as claimed. \square

4. Analytic aspects of Smith maps.

Note. Beginning in this section and for the remainder of the paper, for concreteness we restrict to the case of associative Smith maps $u : (\Sigma^3, g, *, \mu) \to (M^7, h, J, \varphi)$, although all of our results also apply to Cayley Smith maps after making the obvious modifications.

Thus, (Σ, g) is an oriented Riemannian 3-manifold with volume form μ and Hodge star operator *, and (M, φ) is a 7-manifold with a G₂-structure φ inducing a Riemannian metric h and an associated 2-fold vector cross product which we denote by J in analogy with almost complex structures. In addition, recall that M is compact without boundary and that it is isometrically embedded into $(\mathbb{R}^d, g_{\text{euc}})$.

The present section is divided into five parts. In §4.1 we derive an explicit useful form of the Smith equation in local coordinates. In §4.2 we prove the ε -regularity theorem, which is Theorem 4.7. In §4.3 we establish a mean value inequality, interior regularity, and a removable singularity theorem. §4.4 is devoted to the basic convergence result for a sequence of Smith maps with uniformly bounded 3-energy, which gives C^1 -subsequential convergence away from a set of isolated points and is important for the bubble tree construction in §5. Finally in §4.6 we mention two well-known results giving positive lower bounds for the 3-energy of maps from S^3 which are "nontrivial" in some sense.

4.1. The Smith equation. In this section and the next, Greek indices run from 1 to 3 and Latin indices run from 1 to d, where M is isometrically embedded into $(\mathbb{R}^d, g_{\text{euc}})$. In particular we have global coordinates y^1, \ldots, y^d on \mathbb{R}^d and the vector fields $\frac{\partial}{\partial y^1}, \ldots, \frac{\partial}{\partial y^d}$ are orthonormal. We extend the vector cross product J on M to \mathbb{R}^d as follows. Let δ_0 be chosen such that

$$\mathcal{N}_{2\delta_0}(M) := \{ x \in \mathbb{R}^d \mid \operatorname{dist}_{\mathbb{R}^d}(x, M) < 2\delta_0 \}$$

is strictly contained in a tubular neighborhood of M in \mathbb{R}^d , and let π denote the nearest-point projection onto M. Moreover, let ζ be a cutoff function which is identically equal to 1 on $\mathcal{N}_{\frac{3\delta_0}{2}}(M)$ and vanishes outside of $\mathcal{N}_{2\delta_0}(M)$. Then we define

$$\tilde{J}_y(X,Y) = \begin{cases} \zeta(y) J_{\pi(y)} \big((d\pi)_y X, (d\pi)_y Y \big) & \text{if } y \in \mathcal{N}_{2\delta_0}(M), \\ 0 & \text{otherwise.} \end{cases}$$

Note that \tilde{J}_y is still bilinear and skew-symmetric for all $y \in \mathbb{R}^d$, and moreover

$$\tilde{J}_{y}(X,Y) = J_{y}(X,Y)$$
 whenever $y \in M$ and $X,Y \in T_{y}M$.

Note that \tilde{J} is *not* a vector cross product on \mathbb{R}^d . However, this does not affect any of the results where \tilde{J} is used, as only the bilinearity and skew-symmetry are required whenever we use \tilde{J} .

We write \tilde{J} in coordinates as

$$J^{i}_{jk}(y)\frac{\partial}{\partial y^{i}} = \tilde{J}_{y}\Big(\frac{\partial}{\partial y^{j}}, \frac{\partial}{\partial y^{k}}\Big).$$

Note that we have dropped the tilde in the notation for the components.

In addition, because all of the results in §4.2 and §4.3 are local, in these two sections we assume the domain is the ball $B(2) \subseteq \mathbb{R}^3$ equipped with a Riemannian metric $g = (g_{\alpha\beta})$. Moreover we assume B(2) is given the standard orientation, so that $\partial_1, \partial_2, \partial_3$ is a positive basis everywhere. For a map $u : B(2) \to \mathbb{R}^d$ we write the components of its differential as

$$u_{\alpha}^{i} \frac{\partial}{\partial y^{i}} = \frac{\partial u}{\partial x^{\alpha}} =: u_{\alpha}.$$

We need to rewrite the Smith equation (36) in terms of the components of \tilde{J} and μ . It is more convenient to precompose both sides of (36) with *. Thus we have (recall n=3 from now on) that

$$\frac{1}{\sqrt{3}}|du|du = J(\Lambda^2 du) *. \tag{61}$$

If we write ∂_{λ} for $\frac{\partial}{\partial x^{\lambda}}$, then $*\partial_{\lambda} = \frac{1}{2}C_{\lambda}^{\alpha\beta}\partial_{\alpha} \wedge \partial_{\beta}$ for some functions $C_{\lambda}^{\alpha\beta} = -C_{\lambda}^{\beta\alpha}$. Using that * is an isometry and $*^2 = 1$ in three dimensions, we find that

$$\begin{split} \mu_{\gamma\delta\lambda} &= g(*(\partial_{\gamma} \wedge \partial_{\delta}), \partial_{\lambda}) = g(\partial_{\gamma} \wedge \partial_{\delta}, *\partial_{\lambda}) \\ &= \frac{1}{2} C_{\lambda}^{\alpha\beta} g(\partial_{\gamma} \wedge \partial_{\delta}, \partial_{\alpha} \wedge \partial_{\beta}) = \frac{1}{2} C_{\lambda}^{\alpha\beta} (g_{\gamma\alpha} g_{\delta\beta} - g_{\gamma\beta} g_{\delta\alpha}) \\ &= C_{\lambda}^{\alpha\beta} g_{\gamma\alpha} g_{\delta\beta}. \end{split}$$

Thus we conclude that $C_{\lambda}^{\alpha\beta}=\mu_{\lambda\gamma\delta}g^{\gamma\alpha}g^{\delta\beta}=\mu^{\alpha\beta\gamma}g_{\alpha\lambda}$. The second expression is preferable, since $\mu^{\alpha\beta\gamma}$ are the components of the 3-vector that is metric dual to the Riemannian volume form μ . Thus

$$\mu^{\alpha\beta\gamma} = \frac{1}{\sqrt{g}} \epsilon^{\alpha\beta\gamma},$$

where \sqrt{g} denotes $\sqrt{\det(g_{\alpha\beta})}$ and $\epsilon^{\alpha\beta\gamma}$ is the permutation symbol on three letters. That is, $\epsilon^{\sigma(1)\sigma(2)\sigma(3)} = \operatorname{sgn}(\sigma)$ for $\sigma \in S_3$. Thus

$$*\partial_{\lambda} = \frac{1}{2\sqrt{g}} \epsilon^{\alpha\beta\gamma} g_{\lambda\gamma} \partial_{\alpha} \wedge \partial_{\beta}. \tag{62}$$

Using (62) and the notation defined above, equation (61) becomes

$$\begin{split} \frac{1}{\sqrt{3}}|du|u_{\lambda}^{i}\frac{\partial}{\partial y^{i}} &= \left[J\circ\left(du\wedge du\right)\circ\ast\right]\left(\partial_{\lambda}\right)\\ &= \frac{1}{2\sqrt{g}}\left[J\circ\left(du\wedge du\right)\right]\left(\epsilon^{\alpha\beta\gamma}g_{\lambda\gamma}\partial_{\alpha}\wedge\partial_{\beta}\right)\\ &= \frac{1}{2\sqrt{g}}\epsilon^{\alpha\beta\gamma}g_{\lambda\gamma}J(u_{\alpha},u_{\beta})\\ &= \frac{1}{2\sqrt{g}}\epsilon^{\alpha\beta\gamma}g_{\lambda\gamma}(J_{jk}^{i}\circ u)u_{\alpha}^{j}u_{\beta}^{k}\frac{\partial}{\partial y^{i}}. \end{split}$$

Finally, equating the $\frac{\partial}{\partial y^i}$ components of each side and skew-symmetrizing in α , β , we conclude that

$$\frac{1}{\sqrt{3}}|du|u_{\lambda}^{i} = \frac{1}{4\sqrt{g}}\epsilon^{\alpha\beta\gamma}g_{\lambda\gamma}(J_{jk}^{i}\circ u)(u_{\alpha}^{j}u_{\beta}^{k} - u_{\beta}^{j}u_{\alpha}^{k}),\tag{63}$$

which is the form of the Smith equation that we require in the next section.

4.2. The ε -regularity theorem. The main result of this section, namely Theorem 4.7, is an ε -regularity theorem in the spirit of Sacks-Uhlenbeck [42]. The proof is accomplished in a number of stages. First, in Proposition 4.1 we derive from the Smith equation (63) a system of second order elliptic equations that resembles the n-harmonic map system. Then, combining the work of Uhlenbeck [49] and some results from harmonic analysis, which are collected in Appendix A (see also Remarks 4.2 and 4.5), we prove a decay estimate on the 3-energy under a smallness assumption in Proposition 4.3. From there and using standard theory, we derive $C^{0,\alpha}$ -regularity for any $\alpha \in (0,1)$ in Proposition 4.6. Finally, choosing α sufficiently close to 1 and adapting an argument from [11], which is itself based on a refinement [17] of Uhlenbeck's work, allows us to prove Theorem 4.7.

We point out that the system (65) we derive in Proposition 4.1 belongs to the class of n-harmonic type systems studied by Mou–Wang in [37]. Using the same harmonic analysis tools that we use below, but rather differently, they obtained a Hölder continuity result similar to Corollary 4.6 for some $\alpha \in (0,1)$. They also obtained analogues of Theorem 4.10 (the first conclusion) and Proposition 4.17 using arguments different from ours.

PROPOSITION 4.1. Let $u \in W^{1,3}(B(2); M)$ be an associative Smith map. That is, u satisfies the Smith equation (63) almost everywhere on B(2). Then the following hold:

(a) We have

$$\int_{B(2)} \langle |du|du, d\eta \rangle_g d\mu_g = -\frac{\sqrt{3}}{4} \int_{B(2)} \epsilon^{\alpha\beta\gamma} (u^j - \xi^j) \Big((J^i_{jk} \circ u)_\alpha u^k_\beta - (J^i_{jk} \circ u)_\beta u^k_\alpha \Big) \eta^i_\gamma dx, \tag{64}$$

for any constant vector $\xi \in \mathbb{R}^d$ and any $\eta \in W_0^{1,3} \cap L^{\infty}(B(2); \mathbb{R}^d)$.

(b) We have

$$\int_{B(2)} \langle |du|du, d\eta \rangle_g d\mu_g = -\frac{\sqrt{3}}{2} \int_{B(2)} \epsilon^{\alpha\beta\gamma} (J^i_{jk} \circ u)_{\gamma} u^j_{\alpha} u^k_{\beta} \eta^i dx.$$
 (65)

for any $\eta \in W_0^{1,3} \cap L^{\infty}(B(2); \mathbb{R}^d)$.

Proof. The idea is to compute $\int_{B(2)} \langle |du|du, d\eta \rangle_g d\mu_g$ using the Smith equation (63) and then integrate by parts with the help of smooth approximations. We only give the details for part (a), since part (b) is similar.

For any η as in the statement of the theorem, using (63) and that ξ is a constant

vector, we have

$$\begin{split} \int_{B(2)} \langle |du| du, d\eta \rangle_g d\mu_g &= \int_{B(2)} |du| u^i_\lambda \eta^i_\delta g^{\lambda\delta} \sqrt{g} dx \\ &= \frac{\sqrt{3}}{4} \int_{B(2)} \epsilon^{\alpha\beta\gamma} (J^i_{jk} \circ u) (u^j_\alpha u^k_\beta - u^j_\beta u^k_\alpha) \eta^i_\gamma dx \\ &= \frac{\sqrt{3}}{4} \int_{B(2)} \epsilon^{\alpha\beta\gamma} (J^i_{jk} \circ u) \left((u^j - \xi^j)_\alpha u^k_\beta - (u^j - \xi^j)_\beta u^k_\alpha \right) \eta^i_\gamma dx. \end{split}$$

We want to integrate by parts, but u does not necessarily possess weak second derivatives. Thus we take a sequence (v_n) in $C^{\infty}(\overline{B(2)}; \mathbb{R}^d)$ such that $||v_n - u||_{1,3;B(2)} \to 0$ as $n \to \infty$, and consider

$$\begin{split} &\int_{B(2)} \epsilon^{\alpha\beta\gamma} (J^i_{jk} \circ u) \left((u^j - \xi^j)_{\alpha} v^k_{n,\beta} - (u^j - \xi^j)_{\beta} v^k_{n,\alpha} \right) \eta^i_{\gamma} dx \\ &= \epsilon^{\alpha\beta\gamma} \int_{B(2)} (J^i_{jk} \circ u) (u^j - \xi^j)_{\alpha} v^k_{n,\beta} \eta^i_{\gamma} dx - \epsilon^{\alpha\beta\gamma} \int_{B(2)} (J^i_{jk} \circ u) (u^j - \xi^j)_{\beta} v^k_{n,\alpha} \eta^i_{\gamma} dx. \end{split}$$

The first term can be treated as follows.

$$\begin{split} &\int_{B(2)} (J^i_{jk} \circ u)(u^j - \xi^j)_{\alpha} v^k_{n,\beta} \eta^i_{\gamma} dx \\ &= -\int_{B(2)} (u^j - \xi^j) \left(v^k_{n,\beta\alpha} \eta^i_{\gamma} (J^i_{jk} \circ u) + v^k_{n,\beta} \eta^i_{\gamma\alpha} (J^i_{jk} \circ u) + v^k_{n,\beta} \eta^i_{\gamma} (J^i_{jk} \circ u)_{\alpha} \right). \end{split}$$

Similarly,

$$\begin{split} \int_{B(2)} (J^i_{jk} \circ u) (u^j - \xi^j)_{\beta} v^k_{n,\alpha} \eta^i_{\gamma} dx \\ &= - \int_{B(2)} (u^j - \xi^j) \left(v^k_{n,\beta\alpha} \eta^i_{\gamma} (J^i_{jk} \circ u) + v^k_{n,\alpha} \eta^i_{\beta\gamma} (J^i_{jk} \circ u) + v^k_{n,\alpha} \eta^i_{\gamma} (J^i_{jk} \circ u)_{\beta} \right). \end{split}$$

Subtracting this from the previous identity, multiplying by $\epsilon^{\alpha\beta\gamma}$, and summing over α, β, γ , we arrive at

$$\begin{split} \int_{B(2)} \epsilon^{\alpha\beta\gamma} (J^i_{jk} \circ u) \left((u^j - \xi^j)_{\alpha} v^k_{n,\beta} - (u^j - \xi^j)_{\beta} v^k_{n,\alpha} \right) \eta^i_{\gamma} dx \\ &= - \int_{B(2)} \epsilon^{\alpha\beta\gamma} (u^j - \xi^j) \left((J^i_{jk} \circ u)_{\alpha} v^k_{n,\beta} - (J^i_{jk} \circ u)_{\beta} v^k_{n,\alpha} \right) \eta^i_{\gamma} dx. \end{split}$$

Letting $n \to \infty$ completes the proof. \square

Remark 4.2. Equation (64) is useful because, fixing any i, j, the vector field X on B(2) defined by

$$X^{\gamma} = \epsilon^{\alpha\beta\gamma} \big((J^i_{jk} \circ u)_{\alpha} u^k_{\beta} - (J^i_{jk} \circ u)_{\beta} u^k_{\alpha} \big)$$

lies in $L^{\frac{3}{2}}(B(2))$ and furthermore has divergence zero in B(2) in the sense of distributions. Hence, by Proposition A.6, we know that for any $\eta \in W_0^{1,3}(B(1); \mathbb{R}^d)$ and fixed i, j, the function

$$\epsilon^{\alpha\beta\gamma} \big((J^i_{jk} \circ u)_\alpha u^k_\beta - (J^i_{jk} \circ u)_\beta u^k_\alpha \big) \eta^i_\gamma$$

lies in the Hardy space $\mathcal{H}^1(\mathbb{R}^3)$ and has \mathcal{H}^1 -norm bounded by

$$C||X||_{\frac{3}{2};B(2)}||D\eta||_{3;B(1)} \le C||Du||_{3;B(2)}^2||D\eta||_{3;B(1)}.$$

Next, we establish the regularity of $W^{1,3}$ -Smith maps on B(2).

PROPOSITION 4.3. For all $\alpha \in (0,1)$, there exist constants $\varepsilon_0, \theta \in (0,\frac{1}{9})$, depending only on α, M, J and on the embedding $M \to \mathbb{R}^d$, such that if the metric g on B(2) satisfies

$$|g - g_{\text{euc}}|_{0;B(2)} + |Dg|_{0;B(2)} \le \varepsilon_0,$$
 (66)

and if $u: B(2) \to M$ is a $W^{1,3}$ associative Smith map with

$$\int_{B(2)} |Du|^3 dx < \varepsilon_0, \tag{67}$$

then we have

$$\int_{B(\theta)} |Du|^3 dx \le \theta^{3\alpha} \int_{B(2)} |Du|^3 dx. \tag{68}$$

Remark 4.4. Note that if $\varepsilon_0 < \frac{1}{9}$ then, because the domain is 3-dimensional, condition (66) implies that

$$\frac{1}{2}|\xi|^2 \le g_{\alpha\beta}(x)\xi^{\alpha}\xi^{\beta} \le 2|\xi|^2, \text{ for all } x \in B(2), \xi \in \mathbb{R}^3.$$
(69)

Here the $|\cdot|$ denotes length measured with respect to the Euclidean metric g_{euc} . We often use relation (69) to go back and forth between g and g_{euc} .

Proof of Proposition 4.3. First note that the inequalities (69) mean that for any map $w \in W^{1,3}(B(1); \mathbb{R}^d)$ we have

$$\frac{1}{8} \int_{A} |Dw|^{3} dx \le \int_{A} |dw|_{g}^{3} d\mu_{g} \le 8 \int_{A} |Dw|^{3} dx, \quad \text{for all } A \subseteq B(1).$$
 (70)

Next we let v be the unique minimizer of $\int_{B(1)} |Dw|^3 dx$ amongst all maps $w \in W^{1,3}(B(1); \mathbb{R}^d)$ satisfying $w - u \in W_0^{1,3}(B(1); \mathbb{R}^d)$. Then v satisfies

$$\int_{B(1)} (|Dv|Dv \cdot D\eta) dx = 0 \quad \text{for all } \eta \in W_0^{1,3}(B(1); \mathbb{R}^d).$$
 (71)

Fix $i \in \{1, ..., d\}$ and choose $\eta = (v^i - \operatorname{ess\,sup}_{B(1)} u^i)_+ \frac{\partial}{\partial y^i}$, where $(\cdot)_+$ denotes the positive part. This η belongs to $W_0^{1,3}(B(1); \mathbb{R}^d)$ because v - u does. Then (71) implies

$$v^i \le \operatorname{ess\,sup}_{B(1)} u^i \text{ on } B(1), \quad \text{for all } i = 1, \dots, d.$$
 (72)

Similarly, we can prove that

$$v^i \ge \operatorname{ess\,inf}_{B(1)} u^i \text{ on } B(1), \quad \text{for all } i = 1, \dots, d.$$
 (73)

Since M is compact, the two sets of inequalities above imply that v lies in $L^{\infty}(B(1); \mathbb{R}^d)$. Moreover, by the 3-energy minimizing property of v we have

$$\int_{B(1)} |Dv|^3 dx \le \int_{B(1)} |Du|^3 dx. \tag{74}$$

The following result due to Uhlenbeck [49] is the reason we introduced the map v. Specifically, her work tells us that $v \in C^{1,\gamma}_{\text{loc}}(B(1); \mathbb{R}^d)$, and satisfies

$$\sup_{B(\frac{1}{2})} |Dv|^3 \le C \int_{B(1)} |Dv|^3 dx \quad \Big(\le C \int_{B(1)} |Du|^3 dx \Big), \tag{75}$$

where γ , C are universal constants.

To establish the asserted energy decay, we compare the 3-energy of u with that of v and use (64) to estimate the difference. To do that, we need a system of equations satisfied by v which resembles (64). Therefore we use (71) to deduce, for all $\eta \in W_0^{1,3}(B(1); \mathbb{R}^d)$, that

$$\int_{B(1)} \langle |dv|_{g} dv, d\eta \rangle \sqrt{g} dx$$

$$= \int_{B(1)} \left(\langle |dv|_{g} dv, d\eta \rangle \sqrt{g} - |Dv| Dv \cdot D\eta \right) dx$$

$$= \int_{B(1)} \left(\sqrt{g} |dv|_{g} \langle dv, d\eta \rangle - |Dv| Dv \cdot D\eta \right) dx$$

$$= \int_{B(1)} \left((\sqrt{g} |dv|_{g} - |Dv|) g^{\alpha \beta} v_{\alpha} \cdot \eta_{\beta} + |Dv| \left(g^{\alpha \beta} - g_{\text{euc}}^{\alpha \beta} \right) v_{\alpha} \cdot \eta_{\beta} \right) dx$$

$$= \int_{B(1)} (F^{\beta} \cdot \eta_{\beta}) dx, \tag{76}$$

where we defined $F^{\beta} = (\sqrt{g}|dv|_g - |Dv|)g^{\alpha\beta}v_{\alpha} + |Dv|(g^{\alpha\beta} - g_{\text{euc}}^{\alpha\beta})v_{\alpha}$. Note that by the definition of F and (74), for some universal constant C we have

$$||F||_{\frac{3}{2};B(1)} \le C|g - g_{\text{euc}}|_{0;B(1)} ||Dv||_{3;B(1)}^2 \le C|g - g_{\text{euc}}|_{0;B(1)} ||Du||_{3;B(1)}^2.$$
 (77)

Now we test both (64) and (76) against w := u - v, extended to be zero outside of B(1), and consider the differences of the resulting identities to get

$$\int_{B(1)} \langle |du|_g du - |dv|_g dv, dw \rangle d\mu_g$$

$$= -\frac{\sqrt{3}}{4} \int_{B(1)} \epsilon^{\alpha\beta\gamma} (u^j - \xi^j) \left((J^i_{jk} \circ u)_\alpha u^k_\beta - (J^i_{jk} \circ u)_\beta u^k_\alpha \right) w^i_\gamma dx - \int_{B(1)} (F^\beta \cdot w_\beta) dx$$

$$= -\frac{\sqrt{3}}{4} \int_{B(1)} \epsilon^{\alpha\beta\gamma} (u^j - \xi^j) \zeta \left((J^i_{jk} \circ u)_\alpha u^k_\beta - (J^i_{jk} \circ u)_\beta u^k_\alpha \right) w^i_\gamma dx - \int_{B(1)} (F^\beta \cdot w_\beta) dx,$$
(78)

where ζ is a cutoff function which is identically 1 on B(1) and vanishes outside of $B(\frac{3}{2})$.

We next show how the equation (78) can be used to estimate $||Dw||_{3;B(1)}$. Note that the second term on the last line of (78) can be estimated by Hölder's inequality

and (77), yielding

$$\int_{B(1)} (F^{\beta} \cdot w_{\beta}) dx \le C \|F\|_{\frac{3}{2}; B(1)} \|Dw\|_{3; B(1)} \le C \|g - g_{\text{euc}}\|_{0; B(1)} \|Du\|_{3; B(1)}^2 \|Dw\|_{3; B(1)}.$$

$$(79)$$

To estimate the first term, we choose $\xi = \int_{B(2)} u dx$ and use Proposition A.5 to deduce that

$$[(u - \xi)\zeta]_{BMO} \le C||Du||_{3;B(2)}. \tag{80}$$

On the other hand, denoting by X_j the ($\mathbb{R}^{3\times d}$ -valued) function defined by

$$(X_j)_i^{\gamma} = \epsilon^{\alpha\beta\gamma} \big((J_{jk}^i \circ u)_{\alpha} u_{\beta}^k - (J_{jk}^i \circ u)_{\beta} u_{\alpha}^k \big),$$

we see by Remark 4.2 and Proposition A.6 that

$$||X_j \cdot Dw||_{\mathcal{H}^1} \le C||Du||_{3:B(2)}^2 ||Dw||_{3;B(1)}.$$

Using this together with (80) and Theorem A.4, we obtain

$$\int_{B(1)} \epsilon^{\alpha\beta\gamma} (u^j - \xi^j) \zeta \left((J^i_{jk} \circ u)_{\alpha} u^k_{\beta} - (J^i_{jk} \circ u)_{\beta} u^k_{\alpha} \right) w^i_{\gamma} dx \le C \|Du\|^3_{3;B(2)} \|Dw\|_{3;B(1)}. \tag{81}$$

Putting (81) and (79) back into the equation (78) and rearranging, we obtain

$$\int_{B(1)} \langle |du|_g du - |dv|_g dv, dw \rangle d\mu_g \le C (\|Du\|_{3;B(2)} + |g - g_{\text{euc}}|_{0;B(1)}) \|Du\|_{3;B(2)}^2 \|Dw\|_{3;B(1)}.$$

Next, we show how the left-hand side controls $||Dw||_{3:B(1)}$. Specifically, we have

$$\begin{split} \int_{B(1)} |Dw|^3 dx &\leq C \int_{B(1)} |dw|_g^3 d\mu_g \\ &\leq C \int_{B(1)} \left(|du|_g + |dv|_g \right) |dw|_g^2 d\mu_g \\ &\leq C \int_{B(1)} \langle |du|_g du - |dv|_g dv, dw \rangle d\mu_g, \end{split}$$

where the first inequality above follows from (69), and in the last inequality we used [11, equation (10) on page 240]. To sum up, we arrive at

$$\int_{B(1)} |Dw|^3 dx \le C (\|Du\|_{3;B(2)} + |g - g_{\text{euc}}|_{0;B(1)}) \|Du\|_{3;B(2)}^2 \|Dw\|_{3;B(1)}. \tag{82}$$

Cancelling a factor of $||Dw||_{3;B(1)}$ from both sides and recalling the assumptions (66) and (67), we obtain

$$\int_{B(1)} |Dw|^3 dx \le C\varepsilon_0 ||Du||_{3;B(2)}^3.$$

On the other hand, by (75) and (74), for each $\theta \in (0, \frac{1}{9})$ we have

$$\int_{B(\theta)} |Dv|^3 dx \le C\theta^3 \int_{B(2)} |Du|^3 dx.$$

Combining the two estimates above gives

$$\int_{B(\theta)} |Du|^3 dx \le C \int_{B(\theta)} |Dv|^3 + |Dw|^3 dx \le C(\theta^3 + \varepsilon_0) \int_{B(2)} |Du|^3 dx. \tag{83}$$

Now fix $\theta \in (0, \frac{1}{9})$ such that

$$C\theta^3 < \frac{1}{2}\theta^{3\alpha},$$

which is possible because $\alpha \in (0,1)$. Then choose $\varepsilon_0 = \theta^3$. We have

$$C(\theta^3 + \varepsilon_0) < 2C\theta^3 < \theta^{3\alpha}$$
.

By the above and (83), we get (68) and the proof is complete. \square

Remark 4.5. The Hardy space \mathcal{H}^1 and the Fefferman–Stein identification of its dual space with BMO (see [14] and also our Theorem A.4) underlie a "compensation phenomenon" that shows up in many geometric PDEs to give solutions better regularity than afforded by standard theory. Examples include the H-system for CMC surfaces (see for example [2, 51]), the harmonic map system [1, 12, 26], p-harmonic maps into special targets [46, 48], and the Cauchy–Riemann equation for J-holomorphic maps [50]. Proposition 4.3 is another manifestation of this compensation phenomenon.

COROLLARY 4.6. For all $\alpha \in (0,1)$, let ε_0 be as in Proposition 4.3. If $u \in W^{1,3}(B(2);M)$ is an associative Smith map with respect to g and if both (66) and (67) hold, then for all $x_0 \in B(\frac{3}{2})$ and $r \in (0,\frac{1}{4})$ we have

$$\int_{B(x_0;r)} |Du|^3 dx \le Kr^{3\alpha} \int_{B(2)} |Du|^3 dx \text{ for all } x_0 \in B(\frac{3}{2}), r \in (0, \frac{1}{4}).$$
 (84)

In particular, the map u lies in $C^{0,\alpha}(B(\frac{3}{2});\mathbb{R}^d)$, with

$$[u]_{\alpha;B(\frac{3}{2})}^{3} \le C \int_{B(2)} |Du|^{3} dx. \tag{85}$$

The constants K and C above have the same dependence as ε_0 .

Proof. Choose any $x_0 \in B(\frac{3}{2})$ and $r < \frac{1}{4}$ and consider the rescalings

$$\tilde{u}(x) = u(x_0 + rx), \qquad \tilde{g}(x) = g(x_0 + rx).$$

Then $(B(2), \tilde{g})$ and $(B(x_0; 2r), g)$ are conformal via $x \mapsto x_0 + rx$, and by the conformal invariance of the Smith equation, we see that \tilde{u} is a Smith map with respect to \tilde{g} . Furthermore, \tilde{g} again satisfies (66) on B(2), and

$$\int_{B(2)} |D\tilde{u}|^3 dx = \int_{B(x_0;2r)} |Du|^3 dx \le \varepsilon_0.$$

Hence we may apply Proposition 4.3 on B(2) to \tilde{u} and undo the rescaling to get

$$\int_{B(x_0,\theta r)} |Du|^3 dx \le \theta^{3\alpha} \int_{B(x_0;2r)} |Du|^3 dx \quad \text{for all } x_0 \in B(\frac{3}{2}), r \in (0,\frac{1}{4}). \tag{86}$$

It is now fairly standard (compare with [18, Lemma 8.23]) to iterate (86) to obtain

$$\int_{B(x_0;r)} |Du|^3 dx \le Kr^{3\alpha} \int_{B(x_0;\frac{1}{2})} |Du|^3 dx \quad \text{ for all } x_0 \in B(\frac{3}{2}), r \in (0,\frac{1}{4}),$$

for some K depending on θ and α . This immediately gives (84), and the second conclusion follows by Morrey's embedding (compare with [18, Chapter 7]). \square

THEOREM 4.7 (ε -regularity). Fix $\alpha \in (\frac{3}{4}, 1)$ and let ε_0 be given by Proposition 4.3 with this choice of α . There exists a constant $\beta \in (0, 1)$, depending only on α, M, J and the embedding $M \to \mathbb{R}^d$, such that if $u \in W^{1,3}(B(2); M)$ is an associative Smith map with respect to g and if both (66) and (67) hold, then the following hold.

- (a) The map u belongs to $C^{1,\beta}(B(1);\mathbb{R}^d)$. Moreover $|u|_{1,\beta;B(1)}$ can be estimated in terms of M, J, the embedding $M \to \mathbb{R}^d$ and $||Du||_{3;B(2)}$.
- (b) In addition, u is smooth on the (open) set

$$\{x \in B(1) \mid Du(x) \neq 0\}.$$

Proof. Part (a) can be proven as in [11, Lemma 6], with only minor modifications, while part (b) is standard. For the sake of completeness we included proofs of both parts in Appendix B. \square

Remark 4.8. The constant ε_0 of Theorem 4.7 is called the threshold energy.

4.3. Mean value inequality, interior regularity and removable singularities. We now discuss three additional important properties of Smith maps, two of which are immediate consequences of the regularity results in $\S4.2$. These are Theorem 4.9, which is a mean value inequality for the gradient of an associative Smith map, Theorem 4.10, which gives everywhere interior regularity of $W^{1,3}$ -Smith maps, and Theorem 4.11, which is a removable singularity result.

Theorem 4.9 (Mean value inequality). Suppose that Ric_g is bounded by K, in the sense that

$$\left|\operatorname{Ric}_{g(x)}(v,v)\right| \le K|v|_{g(x)}^2$$
 for all $x \in B(2), v \in \mathbb{R}^3$.

There exists $\varepsilon_1 > 0$ such that if (66) and (67) hold with ε_1 in place of ε_0 and if $u \in W^{1,3}(B(2); M)$ is an associative Smith map with respect to g, then we have

$$\sup_{x\in B(\frac{1}{3})}|Du(x)|\leq C\Big(\int_{B(2)}|Du|^3dx\Big)^{\frac{1}{3}},$$

where both ε_1 and C depend only on M, J, the embedding $M \to \mathbb{R}^d$, and K.

Proof. This essentially follows from the arguments in [10, Theorem 2.1], which concerns weakly p-harmonic maps with respect to the Euclidean metric. Indications of the main steps of the proof along with necessary modifications can be found Appendix B. \square

THEOREM 4.10 (Interior regularity). Suppose that g is a smooth Riemannain metric on B(2), and that u is a $W^{1,3}$ -Smith map on B(2) with respect to g. Then u has Hölder continuous first derivatives on B(1). Moreover, u is C^{∞} on the open set $\{x \in B(1): du(x) \neq 0\}$.

Proof. Obviously it suffices to prove that for all $x_0 \in B(1)$ there exists $r \in (0, \frac{1}{4})$ such that both conclusions hold with $B(x_0; r)$ in place of B(1). To that end, take any $x_0 \in B(1)$ and define the rescalings \tilde{u} and \tilde{g} as in the proof of Corollary 4.6, with $r < \frac{1}{4}$ to be determined. Then by the conformal invariance of the Smith equation, \tilde{u} is a $W^{1,3}$ -Smith map with respect to \tilde{g} on B(2). Moreover, we have

$$\int_{B(2)} |D\tilde{u}|^3 dx = \int_{B(x_0; 2r)} |Du|^3 dx,$$

and

$$|\tilde{g} - g_{\text{euc}}|_{0;B(2)} + |D\tilde{g}|_{0;B(2)} = |g - g_{\text{euc}}|_{0;B(x_0;2r)} + r|Dg|_{0;B(x_0;2r)} \le 2r|Dg|_{0;B(x_0;2r)}$$

Since $u \in W^{1,3}(B(2); M)$ and g is smooth on B(2), the two relations above imply that we may choose r sufficiently small so that (66) and (67) hold with ε_0 given by Theorem 4.7. Consequently $\tilde{u} \in C^{1,\beta}(B(1); M)$ and is smooth on the open set $\{x \in B(1) \mid D\tilde{u}(x) \neq 0\}$. Since $x_0 \in B(1)$ is arbitrary, we are done upon scaling back to u and recalling the observation at the beginning of the proof. \square

Theorem 4.11 (Removable singuarity). Suppose that g is a smooth Riemannian metric on B(2), and that $u \in C^1_{loc}(B(2) \setminus \{0\}; M)$ is a Smith map with respect to g, satisfying

$$\int_{B(2)} |Du|^3 dx < \infty.$$

Then in fact u extends to a C^1 -Smith map on all of B(2).

Proof. We first note that the assumptions imply that u belongs to $W^{1,3}(B(2);M)$ and that its weak derivative, which coincides with its classical derivative away from the origin, satisfies the Smith equation (with respect to g) almost everywhere. Next, we again consider the rescalings \tilde{u} and \tilde{g} introduced in the proof of Corollary 4.6, this time choosing $x_0 = 0$. Then, as in the proof of Theorem 4.10, there exists a small enough r such that Theorem 4.7 is applicable to \tilde{u} on B(2). Hence \tilde{u} lies in $C^{1,\beta}(B(1);M)$. In other words, $u \in C^{1,\beta}(B(r);M)$, which immediately gives the desired conclusion. \square

4.4. Convergence modulo bubbling. In this section we study sequences of Smith maps with uniformly bounded 3-energy, which we may assume to be C^1 thanks to Theorem 4.10. We show that the estimates in §4.3 give C^1 -(subsequential) convergence locally away from a finite set of points to a C^1 associative Smith map. As mentioned in §1.2, the main result of the present section, Proposition 4.12, figures prominently in the construction of the bubble tree limit. In particular, we show in §5 that bubbling phenomena occur precisely at the points where C^1 -convergence fails.

We now state the main result of this section.

PROPOSITION 4.12 (Convergence modulo bubbling). Let (Σ, g) be a closed Riemannian 3-manifold and let Ω be an open subset of Σ with $\{\Omega_n\}$ an increasing sequence of open subsets exhausting Ω . Moreover, suppose that for each n we have a Riemannian metric g_n on Ω_n and an associative Smith map $u_n \in C^1(\Omega_n, M)$ with respect to g_n , such that

and

$$\int_{\Omega_n} |du_n|_{g_n}^3 d\mu_{g_n} \le E_0 \text{ for all } n.$$
(87)

Then, there exists a finite set of points $S \subseteq \Sigma$ such that, up to taking a subsequence, the following hold.

(a) (u_n) converges in $C^1_{loc}(\Omega \setminus S)$ to $u \in C^1_{loc}(\Omega; M)$ which is an associative Smith map with respect to g, satisfying

$$\int_{\Omega} |du|_g^3 d\mu_g \le E_0.$$

(b) As Radon measures on Ω , we have

$$|du_n|_{g_n}^3 d\mu_{g_n} \to |du|_g^3 d\mu_g + \sum_{x \in S} m_x \delta_x,$$

with each $m_x \geq \frac{\varepsilon_0}{2}$, where the constant ε_0 is the threshold energy from Remark 4.8. (c) If $||du_n||_{p;\Omega_n} \leq C$ for some $p \in (3,\infty]$, then $S = \varnothing$.

Proof. We first identify the set S. Let ν_n denote the Radon measure $|du_n|_{g_n}^3 d\mu_{g_n}$. Then (87) implies that, for all $m \in \mathbb{N}$, the sequence $(\nu_n)_{n>m}$ has uniformly bounded total mass on Ω_m . Hence, by standard functional analysis and a diagonal process, we obtain a subsequence, which we do not relabel, that converges in the weak-* sense to a Radon measure ν on Ω which satisfies

$$\nu(\Omega) \leq E_0$$
.

Consequently, if we define

$$S = \{ x \in \Omega \mid \nu(\{x\}) \ge \frac{1}{2}\varepsilon_0 \},\$$

where ε_0 is the constant from Theorem 4.7, then \mathcal{S} is finite.

To prove part (a), take any $x \notin \mathcal{S}$. Then by the definition of ν and \mathcal{S} , there exists a radius r > 0 such that for all large enough n we have

$$\int_{B(x;r)} |du_n|_{g_n}^3 d\mu_{g_n} < \frac{1}{2}\varepsilon_0,$$

where the geodesic ball B(x;r) is taken with respect to g. By the smooth convergence of g_n to g on $\overline{B(x;r)}$ and the above inequality, and shrinking r if necessary, we may pull u_n and g_n back to B(r) via the exponential map \exp_x with respect to g, and then dilate to B(2) to conclude that the hypotheses of Theorem 4.7 are all verified. Thus we obtain uniform $C^{1,\beta}$ -estimates on $B(x;\frac{r}{2})$ for the sequence (u_n) . Therefore, passing successively to subsequences (without relabelling), we see that (u_n) converges in C^1 on every compact subset of $\Omega \setminus \mathcal{S}$ to some map $u \in C^{1,\beta}_{\mathrm{loc}}(\Omega \setminus \mathcal{S}; M)$ which satisfies the Smith equation pointwise on $\Omega \setminus \mathcal{S}$. To see that u is in fact C^1 on all of Σ , note that for each $\Omega' \subset\subset \Omega \setminus \mathcal{S}$ we have

$$\int_{\Omega'} |du|_g^3 d\mu_g = \lim_{n \to \infty} \int_{\Omega'} |du_n|_{g_n}^3 d\mu_{g_n} \le E_0,$$

and hence $\int_{\Omega} |du|_g^3 d\mu_g \leq E_0$, since \mathcal{S} is a finite set. But then Theorem 4.11 applies to give C^1 -regularity of u on all of Σ , and we have completed the proof of (a).

For (b), we consider the so-called *defect measure*, defined by

$$\theta := \nu - |du|_g^3 d\mu_g,$$

and observe that supp $(\theta) \subset \mathcal{S}$ because u_n converges to u in $C^1_{loc}(\Omega \setminus \mathcal{S})$. Therefore θ must be of the form

$$\theta = \sum_{x \in \mathcal{S}} m_x \delta_x,$$

and it remains to show that $m_x \ge \frac{1}{2}\varepsilon_0$ for all $x \in \mathcal{S}$. Indeed, for any $x \in \mathcal{S}$ and r > 0 such that $B(x;r) \cap \mathcal{S} = \{x\}$, by the definition of θ we have

$$m_x = \theta(B(x;r)) = \nu(B(x;r)) - \int_{B(x;r)} |du|_g^3 d\mu_g.$$

Since $u \in C^1(\Sigma; M)$, letting r tend to zero yields

$$m_x = \lim_{r \to 0} \nu(B(x; r)) = \nu(\{x\}) \ge \frac{1}{2}\varepsilon_0,$$
 (88)

and we are done with part (b).

Finally we prove part (c). Since $p \in (3, \infty]$, we see by the assumption and Hölder's inequality that for all $B(x; r) \subset \Omega_n$ there holds

$$\int_{B(x;r)} |du_n|_{g_n}^3 d\mu_{g_n} \le \left(\int_{B(x;r)} |du_n|_{g_n}^p d\mu_{g_n} \right)^{\frac{3}{p}} \left(\mu_{g_n}(B(x;r)) \right)^{\frac{p-3}{p}} \\
\le C^3 \left(\mu_{g_n}(B(x;r)) \right)^{\frac{p-3}{p}}.$$

This implies, by the smooth local convergence of g_n to g, that for all $x \in \Omega$ there exists a constant C_1 independent of r and n such that

$$\int_{B(x;r)} |du_n|_{g_n}^3 d\mu_{g_n} \le C_1 r^{\frac{3(p-3)}{p}},$$

for r small enough and n large enough. Consequently for all $x \in \Omega$ and small enough r we have

$$\nu(\{x\}) \le \nu(B(x;r)) \le C_1 r^{\frac{3(p-3)}{p}},$$

which implies that $S = \emptyset$ by its definition. \square

Remark 4.13. We make three remarks about Proposition 4.12.

- (i) In practice we only apply Proposition 4.12 to the case $\Omega = \Sigma$ or $\Omega = S^3 \setminus \{p^-\}$, where p^- is the south pole. Note that in the latter case, the finite-energy property of u and Theorem 4.11 imply that u is actually C^1 on all of S^3 .
- (ii) From (88) we deduce that the numbers $\{m_x\}_{x\in\mathcal{S}}$ can be characterized by

$$m_x = \lim_{r \to 0} \lim_{n \to \infty} \nu_n(B(x; r))$$

for all $x \in \mathcal{S}$. Thus one can view m_x as the amount of energy concentrating at the point x. For this reason, the set \mathcal{S} is often referred to as the *energy* concentration set. Note that \mathcal{S} can equivalently be defined as

$$S = \left\{ x \in \Sigma \mid \liminf_{n \to \infty} \nu_n(B(x; r)) \ge \frac{1}{2} \varepsilon_0 \text{ for all } r > 0 \right\}.$$

- (iii) The defect measure θ can be thought to capture the energy that escapes the limit map u. One of the main goals of the present paper is to study what happens to this "escaped energy". We do this in §5. The defect measure and the energy concentration set have long been used in many other contexts, such as the study of harmonic maps (for example, see [34, 43]) and Yang–Mills connections (for example, see [41, 47]).
- 4.5. Non-relation between *J*-holomorphic and associative bubbling. In Proposition 4.12 we established that a sequence u_n of associative Smith maps with uniformly bounded 3-energy will (up to passing to a subsequence) converge away from a finite set of points, where bubbling occurs. These are points where the 3-energy of u_n concentrates.

As stated in item (ix) of §1.1, an entirely analogous result holds in the classical situation of holomorphic curves. That is, if v_n is a sequence of holomorphic curves with uniformly bounded 2-energy, then up to passing to a subsequence, v_n will converge away from a finite set of points where bubbling occurs. In this case, these are points where the 2-energy of v_n concentrates.

Recall from §3.3 that a complex curve Σ^2 in a Calabi-Yau 3-fold Y^6 gives rise to an associative submanifold $\Sigma^2 \times S^1$ in the G_2 manifold $Y^6 \times S^1$. Therefore it is natural to wonder, if $v_n: (\Sigma^2, [g_{2,n}]) \to (Y^6, h_6)$ is a sequence of holomorphic curves bubbling at points $\{z_1, \ldots, z_N\}$ in Σ , whether it should be the case that $u_n = v_n \times \mathrm{Id}_{S^1}: \Sigma^2 \times S^1 \to Y^6 \times S^1$ is a sequence of associative Smith maps bubbling along a finite set of circles $\{z_k\} \times S^1$, for $k=1,\ldots,N$. This would appear to contradict Proposition 4.12 which says that bubbling of associative Smith maps always occurs in codimension 3, not codimension 2.

The following proposition clarifies in what precise sense the above reasoning is faulty. We use the notation of §3.3. For simplicity we assume that the representatives $g_{2,n}$ in the conformal classes $[g_{2,n}]$ can be chosen to converge smoothly to a limit metric g_2 , since 2-energy concentration for holomorphic curves can already occur in this setting.

PROPOSITION 4.14. Let $v_n: \Sigma \to (Y^6, h_6)$ be a sequence of immersions, where each v_n is holomorphic with respect to the conformal class $[g_{2,n}]$ of metrics on Σ . Assume that the representatives $g_{2,n}$ converge smoothly to a limit metric g_2 . Define $u_n: \Sigma \times S^1 \to (Y \times S^1, h_7 = h_6 + (d\theta^2))$ by

$$u_n(x,\phi) = (v_n(x), f_n(x,\phi))$$

and suppose that the $f'_n := \frac{\partial f_n}{\partial \phi}$ are nonvanishing, so that the u_n are also immersions. Further suppose that the following three conditions all hold:

- (a) Each u_n is an associative Smith map with respect to the conformal class $[g_{3,n}]$ of metrics on $\Sigma \times S^1$.
- (b) The representatives $g_{3,n}$ can be chosen to converge smoothly to some limit metric g_3 .
- (c) The 3-energies

$$\int_{\Sigma\times S^1} |du_n|_{g_{3,n}}^3 \operatorname{vol}_{g_{3,n}} \quad are \ uniformly \ bounded.$$

Then the 3-energy of v_n with respect to $g_{2,n}$ is uniformly bounded. Hence, by item (ix)(c) in §1.1, the sequence $\{v_n\}$ does not bubble.

Proof. Suppose that (a), (b), and (c) all hold. Using the notation of (43) and (48) we write

$$\mu_n^2 = \frac{1}{2} |dv_n|_{g_{2,n}}^2, \qquad \lambda_n^2 = \frac{1}{3} |du_n|_{g_{3,n}}^2.$$
 (89)

Since the v_n are holomorphic with respect to $[g_{2,n}]$, and using assumption (a), Proposition 3.12 applies. From equation (49) in that proposition, we have

$$g_{3,n} = \lambda_n^{-2} (\mu_n^2 g_{2,n} + (df_n)^2). \tag{90}$$

From assumption (b), in local coordinates the components of $g_{3,n}$ converge smoothly to the components of the limit metric g_3 . In local coordinates (x^1, x^2) on Σ , write $g_{2,n} = (g_{2,n})_{ij} dx^i dx^j$. Then in terms of (x^1, x^2, ϕ) we can write (90) as

$$g_{3,n} = \left(\frac{\mu_n^2}{\lambda_n^2}(g_{2,n})_{ij} + \lambda_n^{-2} \frac{\partial f_n}{\partial x^i} \frac{\partial f_n}{\partial x^j}\right) dx^i dx^j + 2\left(\lambda_n^{-2} \frac{\partial f_n}{\partial x^i} \frac{\partial f_n}{\partial \phi}\right) dx^i d\phi + \lambda_n^{-2} \left(\frac{\partial f_n}{\partial \phi}\right)^2 (d\phi)^2.$$

Therefore we must have

$$\lambda_n^{-1} \frac{\partial f_n}{\partial \phi} \to k \quad \text{smoothly as } n \to \infty,$$
 (91)

where k > 0 on $\Sigma \times S^1$. Using this in the cross term $dx^i d\phi$ above tells us that for i = 1, 2 the functions

$$\lambda_n^{-1} \frac{\partial f_n}{\partial x^i} = \frac{\lambda_n^{-2} \frac{\partial f_n}{\partial x^i} \frac{\partial f_n}{\partial \phi}}{\lambda_n^{-1} \frac{\partial f_n}{\partial \phi}}$$

converge smoothly to some limit function. Consequently, looking at the $dx^i dx^j$ terms above, we conclude that

$$\lambda_n^{-1}\mu_n \to h \quad \text{smoothly as } n \to \infty$$
 (92)

where $h \geq 0$ on $\Sigma \times S^1$ because λ_n, μ_n are both positive. We may not have h > 0 everywhere, but this does not matter. Writing $f'_n := \frac{\partial f_n}{\partial \phi}$, by combining (91) and (92) we find that

$$\frac{\mu_n}{f_n'} \to \frac{h}{k}$$
 smoothly as $n \to \infty$. (93)

Thus, for n sufficiently large, we have

$$0 < \frac{\mu_n}{f_n'} < \left| \frac{h}{k} \right|_{0; \Sigma \times S^1} + 1 =: L$$

and hence

$$\frac{f'_n}{\mu_n} > L^{-1}$$
 everywhere on $\Sigma \times S^1$ for sufficiently large n . (94)

Note that L is just a positive *constant*.

Since the v_n are holomorphic with respect to $[g_{2,n}]$, and using assumption (a), Proposition 3.12 applies. Equation (50) in that proposition gives

$$\lambda_n^3 \operatorname{vol}_{g_{3,n}} = \mu_n^2 f_n' d\phi \wedge \operatorname{vol}_{g_{2,n}}. \tag{95}$$

From (95) and (89), we have that for n sufficiently large,

$$\begin{split} \frac{1}{3\sqrt{3}} \int_{\Sigma \times S^1} |du_n|_{g_{3,n}}^3 \mathrm{vol}_{g_{3,n}} &= \int_{\Sigma \times S^1} \lambda_n^3 \mathrm{vol}_{g_{3,n}} \\ &= \int_{\Sigma \times S^1} f_n' \mu_n^2 \mathrm{vol}_{g_{2,n}} d\phi \\ &= \int_{\Sigma \times S^1} \frac{f_n'}{\mu_n} \mu_n^3 \mathrm{vol}_{g_{2,n}} d\phi \\ &> L^{-1} \int_{\Sigma \times S^1} \mu_n^3 \mathrm{vol}_{g_{2,n}} d\phi \\ &= \frac{2\pi L^{-1}}{2^{\frac{3}{2}}} \int_{\Sigma} |dv_n|_{g_{2,n}}^3 \mathrm{vol}_{g_{2,n}}. \end{split}$$

Hence, assumption (c) implies that the 3-energy of v_n with respect to $g_{2,n}$ is uniformly bounded, which by item (ix)(c) in §1.1 prevents $\sup_{x\in\Sigma}|dv_n(x)|_{g_2}$ from going to infinity. \square

Remark 4.15. The above argument suggests that in order to construct a sequence of associative Smith maps u_n from a sequence of holomorphic curves v_n that "bubbles" along circles, one would have to violate assumptions (b) or (c) above. That is, one would have to allow the conformal classes $[g_{3,n}]$ to degenerate, or else allow for the 3-energies of u_n to be unbounded.

REMARK 4.16. Suppose we chose $f_n(x,\phi) = \phi$, so that $u_n = v_n \times \operatorname{Id}_{S^1}$. With this choice for f_n , if $(v_n)_*[\Sigma] \in H^2(Y,\mathbb{R})$ is constant, then $(u_n)_*[\Sigma \times S^1] \in H^3(Y \times S^1,\mathbb{R})$ is constant. But then $f'_n = 1$ for all n, and equation (94) shows that the sequence $\{\sup_{x \in \Sigma} \mu_n(x)\}$ is bounded. Thus for the choice $f_n(x,\phi) = \phi$, if (a), (b), (c) all hold, then the sequence v_n does not bubble by item (ix)(c) of §1.1.

4.6. Energy lower bound for maps from S^3 . In this section we record for later use two results, both of which give lower bounds for the 3-energy of maps from S^3 which are "nontrivial" in some sense. The first result concerns Smith maps on the standard sphere S^3 and is important for proving that the bubble tree terminates after finitely many steps.

PROPOSITION 4.17 (Energy Gap. Compare with [35, Proposition 4.1.4]). Let ε_0 be as in Theorem 4.7. If $u: S^3 \to M$ is a C^1 -associative Smith map with respect to the round metric on S^3 , and if

$$\int_{S^3} |du|^3 d\mu < \varepsilon_0,$$

then u is constant.

Proof. The argument below is inspired by the proof of [35, Proposition 4.1.4] and relies on the fact that the round S^3 is conformally flat. Let $\sigma: \mathbb{R}^3 \to S^3 \setminus \{p\}$ denote the stereographic projection. By the conformal invariance of the Smith equation and the 3-energy, the map $\bar{u} = \sigma^* u$ is a Smith map on \mathbb{R}^3 with respect to the flat metric and satisfies

$$\int_{\mathbb{R}^3} |D\bar{u}|^3 dx < \varepsilon_0.$$

In particular, Theorem 4.9 is applicable to the map $\bar{u}_R(x) := \bar{u}(Rx)$ on B(2) for any R > 0, giving us

$$|D\bar{u}|_{0;B(\frac{R}{2})} \le CR^{-3} \int_{B(2R)} |D\bar{u}|^3 dx \le CR^{-3} \varepsilon_0$$
 for all $R > 0$.

From this we easily see that \bar{u} must be constant, and hence u is constant as well. \square

The next result is a special case of a theorem due to White [52] and concerns a general Lipschitz map on S^3 with small energy and is a crucial ingredient in proving that there is no energy loss through the necks.

PROPOSITION 4.18 ([52, Theorem 2]). There exists a constant $\gamma_1 > 0$ depending only on M and the embedding $M \to \mathbb{R}^d$ such that if $u: S^3 \to M$ is a Lipschitz map with

$$\int_{S^3} |du|^3 d\mu < \gamma_1,\tag{96}$$

then u is homotopic to a constant. Here |du| and $d\mu$ are both with respect to the round metric on S^3 .

Proof. For the convenience of the reader we briefly describe what the proof in [52] becomes in our case. In short, the homotopy is constructed out of the averages of u on geodesic balls, and the assumption (96) enters the argument through the Poincaré inequality. To set things up, for a geodesic ball B(x;r), we define

$$(u)_{x,r} = \int_{B(x;r)} u d\mu.$$

Then there exist universal constants $C, r_0 > 0$ such that for all $x \in S^3$ and $r \le r_0$ the following two conditions hold:

$$\frac{1}{2}r^3 \le \mu(B(x;r)) \le 2r^3,$$

$$r^{-3} \int_{B(x;r)} |u - (u)_{x,r}|^3 d\mu \le C \int_{B(x;r)} |du|^3 d\mu.$$

Furthermore, there exists δ_0 depending on M such that

$$\mathcal{N}_{\delta_0}(M) := \{ x \in \mathbb{R}^d \mid \operatorname{dist}_{\mathbb{R}^d}(x, M) < \delta_0 \}$$

is strictly contained in a tubular neighborhood of M in \mathbb{R}^d . Let $\pi: \mathcal{N}_{\delta_0}(M) \to M$ denote the nearest-point projection.

Now since u maps into M, the inequalities above and (96) imply that if γ_1 is small enough, then

$$\operatorname{dist}_{\mathbb{R}^d}((u)_{x,r},M)^3 < 2C\gamma_1 < \delta_0^3 \quad \text{ for all } x \in S^3, r \leq r_0,$$

and thus $\pi((u)_{x,r})$ is well-defined, and we have

$$|(u)_{x,r} - \pi((u)_{x,r})| \le (2C\gamma_1)^{\frac{1}{3}}$$
 for all $x \in S^3, r \le r_0$. (97)

Consider the function $H: S^3 \times [0, r_0] \to M$ defined by

$$H(x,r) = \begin{cases} \pi((u)_{x,r}) & \text{if } r \in (0, r_0], \\ u(x) & \text{if } r = 0. \end{cases}$$

Since u is Lipschitz by assumption, the function H is continuous and gives a homotopy from u to $H(\cdot, r_0)$. It remains to show that the latter is null-homotopic. To that end we recall the following Poincaré inequality on the whole S^3 . Namely,

$$\int_{S^3} |u - (u)_{S^3}|^3 d\mu \le C \int_{S^3} |du|^3 d\mu.$$

Then for all $x \in S^3$ we compute

$$\begin{split} |(u)_{x,r_0} - (u)_{S^3}| &= \left(\int_{B(x;r_0)} |(u)_{x_0,r} - (u)_{S^3}|^3 d\mu \right)^{\frac{1}{3}} \\ &\leq \left(\int_{B(x;r_0)} |(u)_{x_0,r} - u|^3 d\mu \right)^{\frac{1}{3}} + \left(\int_{B(x;r_0)} |u - (u)_{S^3}|^3 d\mu \right)^{\frac{1}{3}} \\ &\leq C \left(\frac{r_0^3}{\mu(B(x;r_0))} \int_{B(x;r_0)} |du|^3 d\mu \right)^{\frac{1}{3}} + C \left[\mu(B(x;r_0)) \right]^{-\frac{1}{3}} \left(\int_{S^3} |du|^3 d\mu \right)^{\frac{1}{3}} \\ &\leq C (1 + r_0^{-1}) \gamma_1^{\frac{1}{3}}. \end{split}$$

Thus for all $x_1, x_2 \in S^3$ we obtain

$$|(u)_{x_1,r_0} - (u)_{x_2,r_0}| \le 2C(1+r_0^{-1})\gamma_1^{\frac{1}{3}}.$$

Combining this with (97), we arrive at

$$|H(x_1, r_0) - H(x_2, r_0)| \le C'(1 + r_0^{-1})\gamma_1^{\frac{1}{3}}$$
 for all $x_1, x_2 \in S^3$.

Thus, making a smaller choice of γ_1 if necessary, we deduce that $H(\cdot, r_0)$ has image contained in a geodesic ball in M, and hence must be null-homotopic. The proof is complete. \square

- 5. The bubble tree. As stated in §4, for the remainder of this paper, we are exclusively concerned with associative Smith maps $u: (\Sigma^3, g) \to (M^7, h)$, although our results also apply to Cayley Smith maps after making the obvious modifications. In this section we employ the following notation:
 - For a $W^{1,3}$ map $u: (\Sigma, g) \to (M, h)$ and a measurable subset $A \subset \Sigma$, we write

$$E_g(u; A) = E(u; A) = \int_A |du|^3 d\mu_g,$$

often suppressing explicit reference to the domain metric g when it is clear from context. (From now on, for convenience, we drop the $\frac{1}{3\sqrt{3}}$ factor in the 3-energy.) For brevity, we often write $E(u) := E(u; \Sigma)$.

- We use $I = (i_1, \ldots, i_k)$ to denote a multi-index of length $|I| = k \ge 1$.
- The symbol ε_0 always denotes the threshold energy constant of Theorem 4.7 and Remark 4.8.
- **5.1. Overview.** Let $u_n: (\Sigma^3, g) \to (M^7, h)$ be a sequence of associative Smith maps with bounded 3-energy. That is,

$$E(u_n) \le E_0$$
 for all $n \in \mathbb{N}$.

By Proposition 4.12, after passing to a subsequence, there exists an associative Smith map $u_{\infty} \in C^1_{loc}(\Sigma; M)$ called the **base map** for which

$$u_n \to u_\infty \text{ in } C^1_{\text{loc}}(\Sigma \setminus \mathcal{S})$$
 (98)

where $S = \{x_1, \ldots, x_q\}$ is a finite set (possibly empty) whose elements are called (zeroth level) bubble points, and also

$$|du_n|^3 d\mu_g \to |du_\infty|^3 d\mu_g + \sum_{i=1}^q m_i \delta(x_i)$$
 as Radon measures, (99)

where each $m_i \geq \frac{1}{2}\varepsilon_0$. The purpose of the whole of §5 is to understand the behavior of the sequence $\{u_n\}$ at the bubble points x_i .

Suppose that $S \neq \emptyset$. At each $x_i \in S$, we choose local coordinates centered at x_i and construct a sequence of conformal maps

$$R_n: \Omega_n \subset S^3_{x_i} \setminus \{p^-\} \to \mathbb{R}^3,$$

where $S_{x_i}^3$ is the round 3-sphere (the subscript x_i is simply for bookkeeping), $p^- \in S_{x_i}^3$ is the south pole, and $\Omega_n \subset S_{x_i}^3$ is a particular increasing sequence of open sets that exhaust $S_{x_i}^3 \setminus \{p^-\}$. The rescaled maps

$$\widetilde{u}_{n,i} := u_n \circ R_n \colon \Omega_n \to M \tag{100}$$

are a sequence of associative Smith maps (with respect to appropriate metrics h_n on Ω_n) that has bounded 3-energy.

Hence, again by Proposition 4.12, there exists an associative Smith map $\widetilde{u}_{\infty,i}\colon (S^3_{x_i},g_{\mathrm{round}})\to (M,h)$ of class C^1_{loc} , called a **first level bubble map**, and a finite set (possibly empty)

$$S_i = \{x_{i1}, \dots, x_{iq_i}\} \subset S_{x_i}^3 \setminus \{p^-\},$$

called the **first level bubble points**, for which (after passing to a subsequence) we have

$$\widetilde{u}_{n,i} \to \widetilde{u}_{\infty,i}$$
 in $C^1_{loc}(S^3_{x_i} \setminus (\{p^-\} \cup S_i))$

and

$$|d\widetilde{u}_{n,i}|^3 d\mu_{h_n} \to |d\widetilde{u}_{\infty,i}|^3 d\mu_{\text{round}} + \sum_{j=1}^{q_i} m_{ij} \delta(x_{ij})$$
 as Radon measures on $S_{x_i}^3 \setminus \{p^-\}$

where each $m_{ij} \geq \frac{1}{2}\varepsilon_0$.

If $S_i \neq \emptyset$ for some $i = 1, \ldots, q$, then this process may be repeated, yielding associative Smith maps $\widetilde{u}_{\infty,ij} \colon (S^3_{x_{ij}}, g_{\text{round}}) \to M$ of class C^1_{loc} , called **second level bubble maps**, and a finite set $S_{ij} \subset S^3_{x_{ij}} \setminus \{p^-\}$ (possibly empty) of **second level bubble points**. Evidently, we may iterate this procedure as long as the sets of bubble points remain nonempty. In §5.2, we show that, in fact, this process must eventually terminate.

The result of this iteration is a **bubble tree**. That is, one obtains a *tree* $(\{T_0, T_I\}, \{E_I\})$, meaning a connected graph without cycles, in the following way:

- The vertex T_0 corresponds to the base map u_{∞} .
- Each vertex T_i corresponds to the first-level bubble map $\widetilde{u}_{\infty,i}$. Each vertex T_i (for $i = 1, \ldots, q$) is joined to T_0 by an edge E_i , which corresponds to the bubble point x_i .

- Each vertex T_{ij} corresponds to the second-level bubble map $\widetilde{u}_{\infty,ij}$. Each vertex T_{ij} is joined to T_i by an edge E_{ij} , which corresponds to the bubble point x_{ij} .
- And so on.

As we show in §5.2, the construction of the bubble tree, which primarily amounts to a careful choice of the rescaling maps R_n and open sets Ω_n , is essentially a formal process. Indeed, in that discussion we do not need the full strength of the associative Smith condition, but only the following properties of such maps:

- The conformal invariance of the 3-energy functional E.
- The conformal invariance of the Smith equation from Proposition 3.6.
- Removal of singularities from Theorem 4.11.
- The convergence result from Proposition 4.12.
- The energy gap from Proposition 4.17.

Analogous properties hold in several other conformally invariant settings, including harmonic maps [40], Yang–Mills connections [7, 13], and holomorphic curves [35, 39, 53], leading to bubble trees in such settings.

Zero Energy Loss and Zero Neck Length. By virtue of (99), we see that energy appears to be lost in the limit. That is,

$$\lim_{n \to \infty} E(u_n) \ge E(u_\infty).$$

Ideally, we would like to say that the discrepancy $\sum m_i$ is completely accounted for by the energies of the bubble maps. More precisely, we would like to say that each zeroth-level energy concentration m_i is equal to the energy $E(\widetilde{u}_{\infty,i})$ of the first-level bubble map plus all of the the first-level energy concentrations m_{ij} . That is,

$$m_i = E(\widetilde{u}_{\infty,i}) + \sum_{i=1}^{q_i} m_{ij}. \tag{101}$$

In turn, the first-level energy concentrations m_{ij} ought to equal the sum of the energy of a second-level bubble map plus all of the second-level energy concentrations m_{ijk} , and so on. If this were the case, then we would indeed have

$$\lim_{n \to \infty} E(u_n) = E(u_\infty) + \sum_{I} E(\widetilde{u}_{\infty,I})$$
 (102)

essentially asserting that energy is preserved in the "bubble tree limit".

A priori, it is not obvious that (101) holds true, and we call the discrepancy τ_i the **energy loss**. That is, we define

$$\tau_i = m_i - \left(E(\widetilde{u}_{\infty,i}) + \sum_{j=1}^{q_i} m_{ij} \right),\,$$

and in §5.4 we prove:

THEOREM 5.1 (No energy loss). We have $\tau_i = 0$. Consequently, (101) and (102) both hold.

Now, although we have described the bubble tree in abstract terms, as a connected graph that records the various bubble maps and bubble points, it may also be regarded

as a concrete geometric object. That is, we may regard the bubble tree as the subset of M given by the union of the images of the base and bubble maps:

$$u_{\infty}(\Sigma) \cup \bigcup_{I} \widetilde{u}_{\infty,I}(S_{x_{I}}^{3}).$$
 (103)

We prove that this set is connected. In fact, in §5.5 we prove the stronger statement that:

THEOREM 5.2 (Zero neck length). We have

$$u_{\infty}(x_i) = \widetilde{u}_{\infty,i}(p^-). \tag{104}$$

By extension, we have $\widetilde{u}_{\infty,I}(x_{Ij}) = \widetilde{u}_{\infty,Ij}(p^-)$.

In fact, our proof of (104) establishes more: we prove the stronger result given in (105) below. Moreover, combining (105) and a straightforward adaptation of the proof of [35, Theorem 5.2.2(ii)], we see that for large enough k, the map u_k is homotopic to the connected sum $u_\infty \# (\#_I \widetilde{u}_{\infty,I})$. Therefore homotopy is preserved in the bubble tree limit.

The proofs of Theorems 5.1 and 5.2 are significantly less trivial than the construction of the bubble tree. Both results rely on a deeper understanding of the energies of associative Smith maps on annuli

$$A = B^{\text{outer}} \setminus B^{\text{inner}},$$

where here $B^{\text{inner}} \subset B^{\text{outer}}$ are concentric 3-balls. We devote §5.3 to this study. Note that the results in §5.3 require geometric properties of associative Smith maps beyond those used in the construction of the bubble tree, namely the energy gap of Proposition 4.18 and the energy identity of Corollary 3.27.

In Lemma 5.7, we explain how to cap off maps from the boundary 2-spheres $\partial B^{\text{outer}}$ and $\partial B^{\text{inner}}$ to obtain maps from 3-balls with controlled energy. We then patch these maps together to obtain a single map $S^3 \to M$ with small energy, which by the energy gap of Proposition 4.18 must be null-homotopic. As a result, we obtain a crucial bound (Proposition 5.8) on the energy of associative Smith maps on annuli A in terms of the energies on the boundary 2-spheres $\partial B^{\text{outer}}$ and $\partial B^{\text{inner}}$.

In §5.4, we prove Theorem 5.1. Intuitively, the idea is to "trap" τ_i in a sequence of carefully chosen annuli

$$A_k = B_k^{\text{outer}} \setminus B_k^{\text{inner}}.$$

That is, we reinterpret τ_i as a subsequential limit of energies on the annuli A_k (in Lemma 5.10(b)(i)), and observe that energies on the boundary 2-spheres $\partial B_k^{\text{outer}}$, $\partial B_k^{\text{inner}}$ approach zero (in Lemma 5.3(c) and Lemma 5.10(b)). Therefore, the bound (Proposition 5.8) described in the previous paragraph implies that the energies on A_k approach zero, whence $\tau_i = 0$.

Now, geometrically we think of the images of the outer 2-spheres $u_k(\partial B_k^{\text{outer}})$ as being close to $u_{\infty}(x_i)$, and think of the images of the inner 2-spheres $u_k(\partial B_k^{\text{inner}})$ as being close to $\widetilde{u}_{\infty,i}(p^-)$. In this way, we regard the images of the annuli $u_k(A_k) \subset M$ as "necks" of the bubble tree (103).

This intuition leads to the beginning of the proof of Theorem 5.2. Indeed, choose points $y_k \in \partial B_k^{\text{outer}}$ and $z_k \in \partial B_k^{\text{inner}}$ and let $\tilde{z}_k = R_k^{-1}(z_k) \subset S_{x_i}^3$. From (100) we get

 $u_k(z_k) = \widetilde{u}_{k,i}(\widetilde{z}_k)$. By the triangle inequality, we have

$$|u_{\infty}(x_{i}) - \widetilde{u}_{\infty,i}(p^{-})| \leq |u_{\infty}(x_{i}) - u_{k}(y_{k})| + |u_{k}(y_{k}) - u_{k}(z_{k})| + |\widetilde{u}_{k,i}(\widetilde{z}_{k}) - \widetilde{u}_{\infty,i}(p^{-})|$$

$$\leq |u_{\infty}(x_{i}) - u_{k}(y_{k})| + \operatorname{diam}(u_{k}(A_{k})) + |\widetilde{u}_{k,i}(\widetilde{z}_{k}) - \widetilde{u}_{\infty,i}(p^{-})|.$$

By our choice of A_k , it is relatively straightforward to see, using Lemma 5.3(c) and Lemma 5.10(b), that the first and third terms approach zero as $k \to \infty$. The significantly more nontrivial matter, however, is establishing that the second term also approaches zero, that is:

$$\lim_{k \to \infty} \operatorname{diam}(u_k(A_k)) = 0. \tag{105}$$

In other words, the bubble tree (103) has no necks. In §5.5, we prove (105) by making use of both the mean value inequality of Theorem 4.9 as well as the bound from Proposition 5.8.

5.2. Construction of the bubble tree. As in §5.1, we consider a sequence $u_n: (\Sigma^3, g) \to (M^7, h)$ of associative Smith maps with bounded 3-energy:

$$E(u_n) \le E_0$$
 for all $n \in \mathbb{N}$.

Let the base map $u_{\infty} \in C^1_{loc}(\Sigma; M)$, the set of bubble points \mathcal{S} , and the energy concentrations $m_i \geq \frac{1}{2}\varepsilon_0$ be as in (98) and (99).

Suppose that $S \neq \emptyset$ and fix a bubble point $x_i \in S$. Choose a geodesic ball centered at x_i that contains no other points in S, and identify it with $B(2) \subset \mathbb{R}^3$ via \exp_{x_i} and a dilation. In this setting, the conclusion of Proposition 4.12 states that

$$u_n \to u_\infty$$
 in $C^1_{loc}(B(2) \setminus \{0\}; M)$

although not in $C^1_{loc}(B(2); M)$, and that

$$|du_n|^3 d\mu_q \to |du_\infty|^3 d\mu_q + m_i \delta_0$$

as Radon measures on B(1).

Given this setup, we now describe the construction of the bubble tree as a three step process. Steps one and two describe the base case, while step three indicates the induction step and contains a proof that the procedure eventually terminates.

Step One: Choice of center points and dilation factors for rescaling. We begin by focusing our attention on x_i . To that end, we let $\varepsilon_k \in (0, \frac{1}{2})$ be a sequence of radii with $\varepsilon_k \to 0$ and

$$E(u_{\infty}; B(2\varepsilon_k)) = O(\frac{1}{k^2}). \tag{106}$$

Having made this choice, we define a nested sequence of open balls $D_1(k) \subset D_2(k) \subset D_3(k) \subset D_4(k)$ centered at 0 by:

$$D_1(k) = B(\frac{1}{2k^2}\varepsilon_k), \quad D_2(k) = B(\frac{1}{k^2}\varepsilon_k), \quad D_3(k) = B(\varepsilon_k), \quad D_4(k) = B(2\varepsilon_k).$$

We also fix, once and for all, a positive constant $\eta_0 > 0$ for which

$$\eta_0 < \frac{1}{16} \min(\frac{1}{3}\varepsilon_0, \gamma_1),$$

where γ_1 is the energy gap constant of Proposition 4.18.

Next, we choose center points and dilation factors by which to rescale. Our choice is given by the following lemma.

Lemma 5.3. The following results hold:

(a) There exists a subsequence $u_k := u_{n_k}$ such that:

$$E(u_k; D_1(k)) = m_i + O(\frac{1}{k^2}),$$
 (107)

$$E(u_k; D_4(k) \setminus D_1(k)) = O(\frac{1}{k^2}),$$
 (108)

$$|u_k - u_\infty|_{1:D_4(k)\setminus D_1(k)} = O(\frac{1}{k^2}).$$
 (109)

(b) There exist a sequence of centers $c_k \in D_2(k)$ and a sequence of radii $\lambda_k \in (0, \frac{\varepsilon_k}{2k^2}]$ such that

$$\eta_0 = E(u_k; D_4(k) \setminus B(c_k; \lambda_k)) \le E(u_k; D_4(k) \setminus B(x; r)) \tag{110}$$

for all balls B(x;r) with centers $x \in \overline{D_3(k)}$ and radii $r \le \lambda_k$.

(c) We have the following estimates:

$$E(u_k; B(c_k; \varepsilon_k)) = m_i + O(\frac{1}{k^2}), \tag{111}$$

$$\varepsilon_k \sup_{\partial B(c_k; \varepsilon_k)} |du_k| = o(1) \quad \text{as } k \to \infty,$$
 (112)

$$\lim_{k \to \infty} |u_k - u_\infty(0)|_{0; B(c_k; \varepsilon_k) \setminus B(c_k; \frac{1}{R}\varepsilon_k)} = 0, \text{ for all } R > 1.$$
 (113)

Proof. (a) Since $u_n \to u_\infty$ in $C^1_{loc}(B(2) \setminus \{0\}; M)$, and since we have the convergence of Radon measures $|du_n|^3 d\mu_g \to |du_\infty|^3 d\mu_g + m_i \delta_0$ on B(1), for each fixed k, the following hold:

$$\lim_{n \to \infty} E(u_n; D_1(k)) = m_i + E(u_\infty; D_1(k)),$$

$$\lim_{n \to \infty} E(u_n; D_4(k) \setminus D_1(k)) = E(u_\infty; D_4(k) \setminus D_1(k)),$$

$$\lim_{n \to \infty} |u_n - u_\infty|_{1; D_4(k) \setminus D_1(k)} = 0.$$

Hence, because we chose ε_k to satisfy (106), for each k, the relations (107), (108), and (109) all hold for large enough choices of n_k . In this way, we obtain a subsequence (u_k) satisfying (107), (108), (109).

(b) Consider the quantity $E(u_k; D_4(k) \setminus B(x; r)) = \int_{B(2\varepsilon_k) \setminus B(x; r)} |du_k|^3 d\mu_g$. Note that it is continuous in (r, x) on $[0, \varepsilon_k] \times \overline{D_3(k)}$. Hence, for each k, the function

$$F_k(r) := \inf_{x \in \overline{D_3(k)}} E(u_k; D_4(k) \setminus B(x; r))$$

is continuous in r. In fact, $F_k : [0, \varepsilon_k] \to \mathbb{R}$ is a decreasing function that satisfies

$$F_k(0) = E(u_k; D_4(k)) = m_i + O(\frac{1}{k^2}),$$

$$F_k(\frac{1}{2k^2}\varepsilon_k) \le E(u_k; D_4(k) \setminus D_1(k)) = O(\frac{1}{k^2}).$$

Recalling our choice of η_0 and that $m_i \geq \frac{\varepsilon_0}{2}$, for each large enough k we have

$$F_k(\frac{1}{2k^2}\varepsilon_k) < \eta_0 < F_k(0).$$

Thus, for each sufficiently large k, there exists a smallest radius $\lambda_k \in [0, \frac{1}{2k^2}\varepsilon_k]$ for which

$$F_k(\lambda_k) = \eta_0.$$

We now choose $c_k \in D_3(k)$ to be a point that attains the infimum defining $F_k(\lambda_k)$, which exists because $\overline{D_3(k)}$ is compact. That is,

$$F_k(\lambda_k) = E(u_k; D_4(k) \setminus B(c_k; \lambda_k)).$$

In fact, we claim that eventually $|c_k| < \frac{1}{k^2} \varepsilon_k$ for large enough k, meaning that $c_k \in D_2(k)$. Indeed, if we instead had $|c_k| \ge \frac{1}{k^2} \varepsilon_k$, then $\lambda_k \le \frac{1}{2k^2} \varepsilon_k$ implies that $D_1(k) \subset D_4(k) \setminus B(c_k; \lambda_k)$, whence

$$m_i + O(\frac{1}{k^2}) = E(u_k; D_1(k)) \le E(u_k; D_4(k) \setminus B(c_k; \lambda_k)) = F_k(\lambda_k) = \eta_0.$$

But this contradicts $\eta_0 < \frac{1}{2}\varepsilon_0 \le m_i$. (c) For (111), since $|c_k| < \frac{1}{k^2}\varepsilon_k$, we have $D_1(k) \subset B(c_k; \varepsilon_k)$. Hence,

$$E(u_k; B(c_k; \varepsilon_k)) = E(u_k; D_1(k)) + E(u_k; B(c_k; \varepsilon_k) \setminus D_1(k))$$

= $m_i + O(\frac{1}{k^2})$

where the last equality follows from (107) and (108) and the fact that $B(c_k; \varepsilon_k) \subset$

To prove (112), note that since $\partial B(c_k; \varepsilon_k) \subset D_4(k) \setminus D_1(k)$ for k sufficiently large, by (109), we get

$$\sup_{\partial B(c_k;\varepsilon_k)} |du_k| \le \sup_{\partial B(c_k;\varepsilon_k)} |du_\infty| + O(\frac{1}{k^2}) = O(1),$$

where the boundedness of $|du_{\infty}|$ uses the fact that u_{∞} extends to a C^1 map on all of B(1) by Theorem 4.11. It follows that

$$\varepsilon_k \sup_{\partial B(c_k; \varepsilon_k)} |du_k| = O(\varepsilon_k) = o(1)$$
 as $k \to \infty$.

Finally, to see (113), we observe that by the triangle inequality, for each $x \in$ $B(c_k; \varepsilon_k) \setminus B(c_k; \frac{1}{R}\varepsilon_k)$, we have:

$$|u_k(x) - u_{\infty}(0)| \le |u_k(x) - u_{\infty}(x)| + |u_{\infty}(x) - u_{\infty}(0)|$$

$$\le |u_k - u_{\infty}|_{0; B(c_k; \varepsilon_k) \setminus B(c_k; \frac{1}{B}\varepsilon_k)} + C\varepsilon_k |du_{\infty}|_{0; B(c_k; \varepsilon_k)}.$$

Since $c_k \in D_2(k)$, we see by the triangle inequality that, when k is so large that $k^2 > 2R$, there holds

$$B(c_k; \varepsilon_k) \setminus B(c_k; \frac{1}{R}\varepsilon_k) \subset D_4(k) \setminus D_1(k).$$

Thus we may use the estimate (109) and the fact that u_{∞} is C^1 on all of B(1) to bound the last two terms in the above string of inequalities and get

$$|u_k(x) - u_{\infty}(0)| \le O(\frac{1}{k^2}) + O(\varepsilon_k)$$

= $o(1)$ as $k \to \infty$.

This proves the lemma. \square

Step Two: The rescaled maps and first level bubble points. With the choices of center points c_k and scale factors λ_k of Lemma 5.3 in place, we may now define the desired rescalings R_k .

We introduce some notation. Let $\sigma \colon S^3 \setminus \{p^-\} \to \mathbb{R}^3$ denote stereographic projection from the south pole $p^- \in S^3$. Also let S^3_+ and S^3_- denote the upper and lower hemispheres, respectively. In particular,

$$\sigma(p^+) = 0, \qquad \sigma(S_+^3) = B(1).$$

We also let $a_k : \mathbb{R}^3 \to \mathbb{R}^3$ denote the affine function $a_k(x) = \lambda_k x + c_k$, so that

$$a_k(B(x;r)) = B(\lambda_k x + c_k; \lambda_k r).$$

In particular,

$$a_k(0) = c_k,$$
 $a_k(B(1)) = B(c_k; \lambda_k).$

We denote their composition by

$$R_k = a_k \circ \sigma \colon S^3 \setminus \{p^-\} \xrightarrow{\sigma} \mathbb{R}^3 \xrightarrow{a_k} \mathbb{R}^3$$

so that

$$R_k(p^+) = c_k, \qquad R_k(S_+^3) = B(c_k; \lambda_k) \subset D_4(k).$$

Finally, we let $\Omega_k \subset S^3$ be the open sets for which

$$R_k(\Omega_k) = D_4(k).$$

Since R_k preserves inclusions, we see that $S^3_+ \subset \Omega_k$. Note that the Ω_k are increasing and exhaust $S^3 \setminus \{p^-\}$.

Lemma 5.4. The rescaled maps

$$\widetilde{u}_{k,i} \colon \Omega_k \to M$$

$$\widetilde{u}_{k,i} \coloneqq u_k \circ R_k = u_k \circ a_k \circ \sigma$$

are associative Smith maps with respect to a sequence of metrics h_k on $\Omega_k \subset S^3 \setminus \{p^-\}$ that converge in C_{loc}^{∞} to the round metric. Moreover, $\widetilde{u}_{k,i}$ has uniformly bounded 3-energy.

Proof. By the conformal invariance of the Smith equation, the maps $u_k \circ a_k$ are associative Smith maps with respect to the metrics $\lambda_k^{-2}a_k^*g$ on $a_k^{-1}(D_4(k))$. Note that $\frac{1}{\lambda_k^2}a_k^*g \to g_{\text{euc}}$ in C_{loc}^{∞} . Again by the conformal invariance of the Smith equation, the maps $\widetilde{u}_{k,i} = u_k \circ a_k \circ \sigma$ are associative Smith with respect to the metrics $\frac{1}{\lambda_k^2}R_k^*g = \frac{1}{\lambda_k^2}\sigma^*a_k^*g$ on $\Omega_k \subset S^3$, and $\frac{1}{\lambda_k^2}R_k^*g \to \sigma^*g_{\text{euc}}$ in C_{loc}^{∞} . Finally, since σ^*g_{euc} is conformal to the round metric, say $\sigma^*g_{\text{euc}} = bg_{\text{round}}$ for an appropriate function b, we see that $h_k := \frac{1}{b}\frac{1}{\lambda_k^2}R_k^*g \to g_{\text{round}}$ in C_{loc}^{∞} . \square

Therefore, by Lemma 5.4 and Proposition 4.12 applied to $\Omega = S_{x_i}^3 \setminus \{p^-\}$ and $\Omega_k = R_k^{-1}(D_4(k))$, there exists a (possibly empty) finite set of points $S_i = \{x_{i1}, \ldots, x_{iq_i}\} \subset S_{x_i}^3 \setminus \{p^-\}$, called **first level bubble points**, and a C^1 associative Smith map $\widetilde{u}_{\infty,i} \colon (S_{x_i}^3, g_{\text{round}}) \to M$, called a **first level bubble map**, such that after passing to a subsequence

$$\widetilde{u}_{k,i} \to \widetilde{u}_{\infty,i}$$
 in $C^1_{\text{loc}}(S^3_{x_i} \setminus (\{p^-\} \cup S_i))$

and such that, as Radon measures on $S_{x_i}^3 \setminus \{p^-\}$, we have

$$|d\widetilde{u}_{k,i}|^3 d\mu_{h_k} \to |d\widetilde{u}_{\infty,i}|^3 d\mu_{\text{round}} + \sum_{j=1}^{q_i} m_{ij} \delta(x_{ij}) =: \kappa_i$$
(114)

with each $m_{ij} \geq \frac{1}{2}\varepsilon_0$. In fact, we claim that all of the first level bubble points lie in the closure of the upper hemisphere. This is part (d) of the following result.

Lemma 5.5. The following results hold:

- (a) $\kappa_i(S^3_{x_i} \setminus \{p^-\}) \le m_i$,
- (b) $\kappa_i(\overline{S_+^3}) \ge m_i \eta_0,$ (c) $\kappa_i(S_-^3 \setminus \{p^-\}) \le \eta_0,$
- (d) $S_i \subset \overline{S^3_+}$.

Proof. (a) By conformal invariance of the 3-energy and (107)–(108), we have

$$E_{h_k}(\widetilde{u}_{k,i};\Omega_k) = E_g(u_k;D_4(k)) = m_i + O(\frac{1}{k^2}).$$
 (115)

Hence, using (114) and that Ω_{ℓ} is an open set, for each ℓ we have:

$$\kappa_i(\Omega_\ell) \le \liminf_{k \to \infty} E_{h_k}(\widetilde{u}_{k,i}; \Omega_\ell) \le \liminf_{k \to \infty} E_{h_k}(\widetilde{u}_{k,i}; \Omega_k) = m_i.$$

Hence, $\kappa_i(S_{x_i}^3 \setminus \{p^-\}) \leq m_i$.

(b) By conformal invariance of the 3-energy and (110), we have

$$E_{h_k}(\widetilde{u}_{k,i};\Omega_k\setminus\overline{S_+^3}) = E_q(u_k;D_4(k)\setminus B(c_k;\lambda_k)) = \eta_0.$$
(116)

Thus, using (114) and the fact that $\overline{S_+^3}$ is compact, followed by (115) and (116), we have

$$\begin{split} \kappa_i(\overline{S^3_+}) &\geq \limsup_{k \to \infty} E_{h_k}(\widetilde{u}_{k,i}; \overline{S^3_+}) \\ &= \limsup_{k \to \infty} \left[E_{h_k}(\widetilde{u}_{k,i}; \Omega_k) - E_{h_k}(\widetilde{u}_{k,i}; \Omega_k \setminus \overline{S^3_+}) \right] \\ &= m_i - \eta_0. \end{split}$$

(c) This follows immediately from parts (a) and (b):

$$\kappa_i(S^3_- \setminus \{p^-\}) = \kappa_i(S^3_{x_i} \setminus \{p^-\}) - \kappa_i(\overline{S^3_+}) \le \eta_0.$$

(d) On the one hand,

$$\kappa_i(S_-^3 \setminus \{p^-\}) = E(\widetilde{u}_{\infty,i}; S_-^3 \setminus \{p^-\}) + \sum_{j: x_{ij} \in S_-^3} m_{ij} \ge \sum_{j: x_{ij} \in S_-^3} m_{ij}.$$
 (117)

On the other hand, by part (c), we have

$$\kappa_i(S_-^3 \setminus \{p^-\}) \le \eta_0 < \frac{1}{2}\varepsilon_0 \le m_{ij} \tag{118}$$

for all $1 \le j \le q_i$. Comparing (117) with (118), we conclude that none of the points in S_i can be in S_-^3 . That is, $S_i \subset \overline{S_+^3}$. \square

Step Three: Iteration and finite termination. If any of the sets S_i is nonempty, then Steps One and Two above may be repeated, resulting in second-level bubble maps $\widetilde{u}_{\infty,ij}:(S^3_{x_{ij}},g_{\mathrm{round}})\setminus\{p^-\}\to M$ and a (possibly empty) finite set $S_{ij}\subset S^3_{x_{ij}}\setminus\{p^-\}$ of second level bubble points. If any of the S_{ij} is nonempty, then we may repeat the process again, and so on.

As explained above in §5.1, iterating this procedure results in a tree $(\{T_0, T_I\}, \{E_I\})$ whose base vertex T_0 corresponds to the base map u_{∞} , whose higher level vertices T_I correspond to bubble maps $\widetilde{u}_{\infty,I}$, and whose edges E_I correspond to bubble points x_I . In the tree, the edge $E_{i_1 \cdots i_k i_{k+1}}$ joins a kth level vertex $T_{i_1 \cdots i_k}$ to a (k+1)st level vertex $T_{i_1 \cdots i_k i_{k+1}}$.

Thus, to conclude the construction, it remains only to show that the bubble tree is finite, that is that the sets $S_{i_1\cdots i_k}$ of kth level bubble points are all empty for k sufficiently large. To do this, we demonstrate that all of the (k+1)st level energy concentrations $m_{i_1\cdots i_{k+1}i_{k+2}}$ are a fixed constant less than the kth level energy concentrations $m_{i_1\cdots i_{k+1}}$. More precisely we have the following result.

LEMMA 5.6. We have $m_{ij} \leq m_i - \frac{1}{2}\eta_0$ for each $1 \leq j \leq q_i$.

Proof. By definition of κ_i followed by Lemma 5.5(a), we have

$$E(\widetilde{u}_{\infty,i}; S_{x_i}^3 \setminus \{p^-\}) + \sum_{i=1}^{q_i} m_{ij} = \kappa_i(S_{x_i}^3 \setminus \{p^-\}) \le m_i.$$
 (119)

Suppose for the sake of contradiction that the first-level bubble point $x_{i1} \in \mathcal{S}_i = \{x_{i1}, \dots, x_{iq_i}\}$ has energy concentration $m_{i1} > m_i - \frac{1}{2}\eta_0$. Then (119) implies

$$E(\widetilde{u}_{\infty,i}; S_{x_i}^3 \setminus \{p^-\}) + \sum_{i=2}^{q_i} m_{ij} \le m_i - m_{i1} < \frac{1}{2}\eta_0 < \frac{1}{6}\varepsilon_0.$$

By Proposition 4.17, we see that $\widetilde{u}_{\infty,i}$ is a constant map. Moreover, since each $m_{ij} \geq \frac{1}{2}\varepsilon_0$, we must have $S_i = \{x_{i1}\}$. Thus, the Radon measure $\kappa_i = |d\widetilde{u}_{\infty,i}|^3 d\mu_{\text{round}} + \sum m_{ij}\delta(x_{ij})$ is simply

$$\kappa_i = m_{i1}\delta(x_{i1}).$$

Let $B = B^{S^3}(x_{i1}; \rho)$ denote a small geodesic ball in S^3 centered at x_{i1} . Using the conformal invariance of E first, followed by (114) and $\kappa_i = m_{i1}\delta(x_{i1})$ second, and our hypothesis third, we obtain

$$\lim_{k \to \infty} E_g(u_k; R_k(B)) = \lim_{k \to \infty} E_{h_k}(\widetilde{u}_{k,i}; B) = m_{i1} > m_i - \frac{1}{2}\eta_0.$$

On the other hand, (107)–(108) give

$$E_g(u_k; D_4(k)) = m_i + O(\frac{1}{k^2}).$$

We deduce that for k sufficiently large,

$$E_q(u_k; D_4(k) \setminus R_k(B)) < \frac{1}{2}\eta_0 + O(\frac{1}{k^2}).$$
 (120)

Now, by Lemma 5.5(d), we have $x_{i1} \in \overline{S_+^3}$, and hence $R_k(x_{i1}) \in \overline{B(c_k; \lambda_k)}$. By choosing the radius $\rho > 0$ smaller if necessary, we may suppose that $R_k(B)$ is contained in a

ball B(z;r) with center $z \in \overline{B(c_k;\lambda_k)} \subset \overline{D_3(k)}$ and radius $r < \lambda_k$. But then (120) implies that

$$E_g(u_k; D_4(k) \setminus B(z; r)) \le E_g(u_k; D_4(k) \setminus R_k(B)) < \eta_0,$$

which contradicts (110). \square

Iterating Lemma 5.6 shows that the bubble tree has finitely many levels. Indeed, at a kth level bubble point $x_I = x_{i_1 \cdots i_{k+1}}$, the energy concentration $m_I = m_{i_1 \cdots i_{k+1}}$ at x_I satisfies

$$m_I \le m_{i_1} - k\left(\frac{1}{2}\eta_0\right).$$

On the other hand, the energy concentration m_I must satisfy

$$m_I \geq \frac{1}{2}\varepsilon_0$$
.

Together, these two inequalities yield $k(\frac{1}{2}\eta_0) \leq m_{i_1} - \frac{1}{2}\varepsilon_0$, an upper bound on k. That is, for k is sufficiently large, the set $\bigcup_{|I|=k} S_I$ of kth level bubble points is empty, and hence the bubble tree terminates.

5.3. Energy of Smith maps on annuli. In this section we show (Proposition 5.8) that the 3-energies of associative Smith maps on annular regions are controlled by their 3-energies on the boundary spheres. One could view this as an isoperimetric-type estimate. As mentioned in §5.1, such an estimate is crucial to the proofs of Theorems 5.1 and 5.2.

We begin by recalling the following well-known extension result (compare with [35, Remark 4.4.2]). For completeness we include a proof.

Lemma 5.7. Suppose $u: S^2 \to M$ is a C^1 -map and that

$$\operatorname{diam}_M\left(u(S^2)\right) < \operatorname{inj}_M.$$

Then there exists a Lipschitz map $v: B(1) \to M$ such that $v|_{S^2} = u$ and

$$\int_{B(1)} |Dv|^3 dx \le K \int_{S^2} |du|^3,$$

where K depends only on the geometry of M.

Proof. Fix an arbitrary $\xi_0 \in S^2$ and let $p = u(\xi_0)$. Then, by assumption, it makes sense to talk about the map $\tilde{u}: S^2 \to \mathbb{R}^7$ given by

$$\tilde{u}(\xi) = \exp_p^{-1} \left(u(\xi) \right).$$

Note that $|\tilde{u}(\xi)| = \operatorname{dist}_M(u(\xi), p)$. Next we define $v : B(1) \to M$ to be the homogeneous degree-1 extension of u. Specifically, let

$$v(r\xi) = \exp_p\left(r\tilde{u}(\xi)\right) \quad \text{ for } r \in [0,1], \xi \in S^2.$$

Then a straightforward calculation shows that

$$|Dv(r\xi)|^3 = (|\partial_r v|^2 + r^{-2}|\partial_\xi v|^2)^{\frac{3}{2}}$$

$$\leq C(|\tilde{u}(\xi)|^3 + |du|^3).$$

Therefore we have

$$\int_{B(1)} |Dv|^3 dx = \int_0^1 r^2 \int_{S^2} |Dv(r\xi)|^3 d\xi dr$$

$$\leq C \int_{S^2} (|\widetilde{u}|^3 + |du|^3) d\xi$$

$$\leq C \int_{S^2} (|d\widetilde{u}|^3 + |du|^3) d\xi$$

$$\leq C \int_{S^2} |du(\xi)|^3 d\xi,$$

where we used the Poincaré inequality for $W^{1,3}$ -functions on S^2 in the second-to-last inequality. \square

Proposition 5.8. Let $u: B(2) \to M$ be a C^1 -Smith map with respect to a metric which satisfies

$$\frac{1}{2}|\xi|^2 \le g_{\alpha\beta}(x)\xi^{\alpha}\xi^{\beta} \le 2|\xi|^2, \quad \text{for all } x \in B(2), \xi \in \mathbb{R}^3.$$
 (121)

Suppose further that for some 0 < r < 1 we have

$$\operatorname{diam}_{M}\left(u(\partial B(r))\right) + \operatorname{diam}_{M}\left(u(\partial B(1))\right) < \operatorname{inj}_{M},\tag{122}$$

and that

$$\int_{B(1)\backslash B(r)} |du|^3 d\mu_g + Kr \int_{\partial B(r)} |du|^3 dS_g + K \int_{\partial B(1)} |du|^3 dS_g < \frac{1}{8}\gamma_1,$$
 (123)

with γ_1 coming from Proposition 4.18 and K from Lemma 5.7. Then there holds

$$\int_{B(1)\backslash B(r)} |du|^3 d\mu_g \le C\left(r \int_{\partial B(r)} |du|^3 dS_g + \int_{\partial B(1)} |du|^3 dS_g\right),$$

where C depends only on the geometry of M. In particular C is independent of r.

Proof. By assumption (122), we may apply Lemma 5.7 to $u|_{\partial B(1)}$ and (a suitable rescaling of) $u|_{\partial B(r)}$ to obtain Lipschitz maps $v:B(1)\to M$ and $w:B(r)\to M$ such that

$$v|_{\partial B(1)} = u|_{\partial B(1)}, \qquad w|_{\partial B(r)} = u|_{\partial B(r)},$$

$$\int_{B(1)} |Dv|^3 dx \le K \int_{\partial B(1)} |du|^3,$$

$$\int_{B(r)} |Dw|^3 dx \le Kr \int_{\partial B(r)} |du|^3.$$

Note that $u|_{B(1)\backslash B(r)}$ and w together form a Lipschitz map on B(1) into M, which can be pulled back to a Lipschitz map from S^3_+ into M via the stereographic projection. Similarly, v gives rise to a Lipschitz map from S^3_- into M. The two maps agree on

the equator, and we thus obtain a Lipschitz map $f: S^3 \to M$. Moreover, by the conformal invariance of the 3-energy, we see that

$$\int_{S^{3}} |Df|^{3} dx = \int_{B(1)\backslash B(r)} |Du|^{3} dx + \int_{B(r)} |Dw|^{3} dx + \int_{B(1)} |Dv|^{3} dx
\leq \int_{B(1)\backslash B(r)} |Du|^{3} dx + Kr \int_{\partial B(r)} |du|^{3} + K \int_{\partial B(1)} |du|^{3}
\leq 8 \left(\int_{B(1)\backslash B(r)} |du|^{3} d\mu_{g} + Kr \int_{\partial B(r)} |du|^{3} dS_{g} + K \int_{\partial B(1)} |du|^{3} dS_{g} \right)
< \gamma_{1}.$$

where we used (121) in the second to last inequality, and (123) in the last one. But then by Proposition 4.18, the map f is null-homotopic and hence extends to a Lipschitz map $F: B^4(1) \to M$. We may then invoke Corollary 3.27 with $\Sigma = S^3$ and $W = B^4(1)$ to obtain

$$\int_{B(1)\backslash B(r)} |du|^3 d\mu_g \le \int_{B(r)} |dw|^3 d\mu_g + \int_{B(1)} |dv|^3 d\mu_g$$

$$\le C \left(r \int_{\partial B(r)} |du|^3 dS_g + \int_{\partial B(1)} |du|^3 dS_g \right),$$

and the proof is complete. \square

5.4. No energy loss. This section is devoted to the proof of Theorem 5.1 (No energy loss). Recall that, by the definition of τ_i and the measure κ_i , we have $\tau_i = m - \kappa_i(S^3 \setminus \{p^-\})$. To streamline notation, in this section we omit the subscript i, writing $m := m_i$ and $\tau := \tau_i$ and $\kappa := \kappa_i$. We also write $v_k := \widetilde{u}_{k,i}$ and $v := \widetilde{u}_{\infty,i}$. Finally we write $S_1 = S_i$.

As a preliminary step, we note the following consequence of Lemma 5.5, which says that the energy loss τ is nonnegative, and lies below the threshold η_0 we chose in Lemma 5.3.

LEMMA 5.9. We have $0 < \tau < \eta_0$.

Proof. The lower bound $\tau \geq 0$ is exactly the statement of Lemma 5.5(a). For the upper bound, observe that Lemma 5.5(b) gives

$$\tau = m - \kappa(S^3 \setminus \{p^-\}) \le m - \kappa(\overline{S_+^3}) \le \eta_0$$

as desired. \Box

Next, as explained in $\S5.1$, we want to identify (after passing to a further subsequence if necessary) suitable annuli A_l for which

$$\tau = \lim_{l \to \infty} E(u_l; A_l).$$

The outer boundaries of A_l are a subsequence $\partial B(c_{k_l}; \varepsilon_{k_l})$ of the spheres $\partial B(c_k; \varepsilon_k)$ constructed in §5.2. Choosing this subsequence and finding the inner boundaries is the content of Lemma 5.10 below. The proof of Theorem 5.1 is given after that of Lemma 5.10.

We introduce the following notation: For r > 0 we define the open sets

$$G(r) = \sigma^{-1}(B(r)) \subset S^3,$$

so that

$$R_k(G(r)) = B(c_k; r\lambda_k).$$

Then, since the sets $G(\sqrt{k})$ exhaust $S^3 \setminus \{p^-\}$, by the definition of τ we know that $\kappa(G(\sqrt{l}))$ increases to $\kappa(S^3 \setminus \{p^-\}) = m - \tau$ as $l \to \infty$. In other words,

$$\kappa(G(\sqrt{l})) = m - \tau + o(1) \quad \text{as } l \to \infty.$$
 (124)

Moreover, since v is a C^1 -map on S^3 , we have

$$E(v; S^3 \setminus G(\sqrt{l})) = o(1)$$
 as $l \to \infty$. (125)

Lemma 5.10. We have:

(a) There exists a subsequence (v_{k_l}) of (v_k) such that the following hold:

$$E_{h_{k_l}}(v_{k_l}; G(\sqrt{l})) = m - \tau + o(1), \tag{126}$$

$$E_{h_{k_l}}(v_{k_l}; G(l^2) \setminus G(\sqrt{l})) = o(1),$$
 (127)

$$|v_{k_l} - v|_{1:G(l^2)\setminus G(\sqrt{l})} = o(1).$$
 (128)

(b) Scaling back to (u_{k_l}) , writing $u_l := u_{k_l}$ and $c_l := c_{k_l}$, and setting

$$\rho_l = \varepsilon_{k_l}, \ \sigma_l = l\lambda_{k_l},$$

we have

$$E(u_l; B(c_l; \rho_l) \setminus B(c_l; \sigma_l)) = \tau + o(1), \tag{129}$$

$$l\lambda_{k_l} \sup_{\partial B(c_{k_l}; l\lambda_{k_l})} |du_l| = o(1), \tag{130}$$

and

$$\lim_{l \to \infty} |u_l - v(p^-)|_{0; B(c_{k_l}; Rl\lambda_{k_l}) \setminus B(c_{k_l}; l\lambda_{k_l})} = 0 \quad \text{for all } R > 1,$$
or equivalently,
$$\lim_{l \to \infty} |v_{k_l} - v(p^-)|_{0; G(Rl) \setminus G(l)} = 0 \quad \text{for all } R > 1.$$
(131)

Proof. (a) Recall that, as Radon measures on $S^3 \setminus \{p^-\}$ we have

$$|dv_k|^3 d\mu_{h_k} \to |dv|^3 d\mu + \sum_{x_{ij} \in \mathcal{S}_1} m_{ij} \delta(x_{ij}) = \kappa,$$

where $S_1 \subseteq \overline{S_+^3}$. This has the following two consequences. First, for all l > 1, we have $\kappa(\partial G(\sqrt{l})) = 0$, which implies that

$$\kappa(G(\sqrt{l})) = \lim_{k \to \infty} E_{h_k}(v_k; G(\sqrt{l})).$$

Secondly, since v_k converges in C^1 to v on compact subsets of $S^3 \setminus (S_1 \cup \{p^-\})$, we infer that, for all l > 1,

$$\lim_{k \to \infty} |v_k - v|_{1; G(l^2) \setminus G(\sqrt{l})} = 0.$$

From these facts and (124), (125), we easily see that there is a subsequence (v_{k_l}) such that (126), (127) and (128) all hold.

(b) To prove (129), recall from Lemma 5.3(c) that

$$E(u_l; B(c_l; \rho_l)) = m + o(1).$$

On the other hand, (126) and (127) imply that

$$E(u_l; B(c_l; \sigma_l)) = m - \tau + o(1).$$

Subtracting this from the previous equality gives (129).

For (130), we note that if $x \in \partial B(c_{k_l}; l\lambda_{k_l})$, then eventually $B(x; \frac{1}{2}l\lambda_{k_l}) \subseteq B(c_{k_l}; l^2\lambda_{k_l}) \setminus B(c_{k_l}; \sqrt{l}\lambda_{k_l})$. Hence, by (127),

$$E(u_l; B(x; \frac{1}{2}l\lambda_{k_l})) \le E(u_l; B(c_{k_l}; l^2\lambda_{k_l}) \setminus B(c_{k_l}; \sqrt{l\lambda_{k_l}})) = o(1),$$

which means for large enough l we may apply Theorem 4.9 to get

$$(l\lambda_{k_l})^3 |du_l(x)|^3 \le E(u_l; B(c_{k_l}; l^2\lambda_{k_l}) \setminus B(c_{k_l}; \sqrt{l\lambda_{k_l}})).$$

Since $x \in \partial B(c_{k_l}; l\lambda_{k_l})$ is arbitrary, we conclude that

$$(l\lambda_{k_l})^3 \sup_{\partial B(c_{k_l};l\lambda_{k_l})} |du_l|^3 \le E(u_l; B(c_{k_l}; l^2\lambda_{k_l}) \setminus B(c_{k_l}; \sqrt{l\lambda_{k_l}})) = o(1).$$

To prove (131), we take an arbitrary $y \in G(Rl) \setminus G(l)$ and estimate

$$|v_l(y) - v(p^-)| \le |v_l(y) - v(y)| + |v(y) - v(p^-)|$$

$$\le |v_l - v|_{0;G(Rl) \setminus G(l)} + |v - v(p^-)|_{0;S^3 \setminus G(l)}.$$

Thus we obtain, for all l > R, that

$$|v_l - v(p^-)|_{0;G(Rl)\setminus G(l)} \le |v_l - v|_{0;G(l^2)\setminus G(\sqrt{l})} + |v - v(p^-)|_{0;S^3\setminus G(l)}.$$

To finish the proof, we recall (128) and also note that $|v-v(p^-)|_{0;S^3\setminus G(l)}=o(1)$ as $l\to\infty$, since $v\in C^1(S^3;M)$ and $S^3\setminus G(l)$ converges to $\{p^-\}$ in Hausdorff distance as $l\to\infty$. \square

We are now ready for the proof of Theorem 5.1.

Proof of Theorem 5.1. We maintain the notation of Lemma 5.10. In particular, we continue to write

$$\rho_l = \varepsilon_{k_l}, \qquad \sigma_l = l\lambda_{k_l}.$$

(Note that $\rho_l, \sigma_l \to 0$ while $\frac{\rho_l}{\sigma_l} \to \infty$ as $l \to \infty$.) Furthermore, we define the following annular regions

$$A_l = B(c_l; \rho_l) \setminus B(c_l; \sigma_l).$$

Then, by Lemma 5.3(c) and Lemma 5.10(b), we have

$$\rho_l \int_{\partial B(c_l;\rho_l)} |du_l|^3 dS_g + \sigma_l \int_{\partial B(c_l;\sigma_l)} |du_l|^3 dS_g = o(1) \text{ as } l \to \infty.$$
 (132)

This estimate then allows us to apply Proposition 5.8 to suitable rescalings of the maps u_l . Specifically, we show that the rescaled maps

$$\tilde{u}_l(x) = u_l(c_l + \rho_l x)$$

eventually satisfy the hypotheses (122) and (123).

To see this, note that we have by equation (112) and equations (129)–(130) that,

$$\lim_{l \to \infty} \left(E(u_l; A_l) + K \rho_l \int_{\partial B(c_l; \rho_l)} |du_l|^3 dS_g + K \sigma_l \int_{\partial B(c_l; \sigma_l)} |du_l|^3 dS_g \right) = \tau,$$

where K is as in Lemma 5.7. Since $\tau \leq \eta_0 < \frac{1}{16}\gamma_1$, we see that there exists l_0 such that for all $l \geq l_0$, the rescaled maps \tilde{u}_l satisfy (123) with $r = \frac{\sigma_l}{\rho_l}$. Moreover, increasing l_0 if necessary, we see that (113) and (131) yield (122) for all $l \geq l_0$. Therefore we may invoke Proposition 5.8 to conclude that

$$E(u_l; A_l) \le C\Big(\rho_l \int_{\partial B(c_l; \rho_l)} |du_l|^3 dS_g + \sigma_l \int_{\partial B(c_l; \sigma_l)} |du_l|^3 dS_g\Big),$$

with C independent of l. Letting $l \to \infty$ and recalling (129) and (132), we get $\tau = 0$ as desired. \square

5.5. Zero neck length. In this section we continue to use notation from §5.3 and §5.4. Our goal here is to prove Theorem 5.2 (Zero neck length), which in the present notation reads

$$u_{\infty}(0) = v(p^{-}).$$

In fact, it is enough to show that

$$\lim_{l \to \infty} \operatorname{diam}\left(u_l\left(B(c_l; \frac{1}{4}\rho_l) \setminus B(c_l; 4\sigma_l)\right)\right) = 0. \tag{133}$$

To see this, note that in view of (113) and (131) with R = 4, the above implies that

$$\lim_{l \to \infty} \operatorname{diam}\left(u_l\left(A_l\right)\right) = 0,$$

which is (105). Then as indicated in §5.1, with the help of (113) and (131) again, we may finish the proof of Theorem 5.2. That is, we choose points $y_l \in \partial B(c_l; \rho_l)$ and $z_l \in \partial B(c_l; \sigma_l)$ and estimate

$$|u_{\infty}(0) - v(p^{-})| \le |u_{\infty}(0) - u_{l}(y_{l})| + |u_{l}(y_{l}) - u_{l}(z_{l})| + |u_{l}(z_{l}) - v(p^{-})|$$

$$\le |u_{\infty}(0) - u_{l}|_{0:\partial B(c_{l}:\sigma_{l})} + \operatorname{diam}(u_{l}(A_{l})) + |u_{l} - v(p^{-})|_{0:\partial B(c_{l}:\sigma_{l})}.$$

By (113), (105) and (131), all three terms in the last line tend to zero as $l \to \infty$, and we are done. Therefore, it remains to prove (133).

As a first step towards proving (133), we establish the following pointwise gradient estimates.

Lemma 5.11. For l sufficiently large, we have

$$\sup_{x \in B(c_l; r_2) \setminus B(c_l; r_1)} |x - c_l|^3 |Du_l(x)|^3 \le CE\left(u_l; B(c_l; 2r_2) \setminus B(c_l; \frac{1}{2}r_1)\right), \tag{134}$$

whenever $2\sigma_l \leq r_1 < r_2 \leq \frac{1}{2}\rho_l$. In particular,

$$\lim_{l \to \infty} \left(\sup_{x \in B(c_l; \frac{1}{2}\rho_l) \setminus B(c_l; 2\sigma_l)} |x - c_l| |Du_l(x)| \right) = 0.$$
 (135)

Proof. To begin, we note two rather obvious facts. First of all, whenever $r_1 \leq r \leq r_2$, we have

$$\min\{2r_2 - r, \ r - \frac{1}{2}r_1\} \ge \frac{1}{2}r.$$

Secondly, if $2\sigma_l \leq r_1 < r_2 \leq \frac{1}{2}\rho_l$, then

$$B(c_l; 2r_2) \setminus B(c_l; \frac{1}{2}r_1) \subseteq A_l$$
.

It follows that, for any l, if $2\sigma_l \leq r_1 < r_2 \leq \frac{1}{2}\rho_l$, then for all $r \in [r_1, r_2]$ and $x \in \partial B(c_l; r)$ we have

$$E(u_l; B(x; \frac{1}{2}r)) \le E(u_l; B(c_l; 2r_2) \setminus B(c_l; \frac{1}{2}r_1)) \le E(u_l; A_l).$$

Since we have shown in §5.4 that

$$\lim_{l \to \infty} E(u_l; A_l) = 0, \tag{136}$$

we infer that there exists an L_0 such that, for all $l \ge L_0$, and r_1, r_2, r, x as above, we may apply the mean value inequality (Theorem 4.9) on $B(x; \frac{1}{2}r)$ to get

$$|r^3|Du_l(x)|^3 \le CE(u_l; B(x; \frac{1}{2}r)) \le CE(u_l; B(c_l; 2r_2) \setminus B(c_l; \frac{1}{2}r_1)).$$

Taking the supremum over $x \in \partial B(c_l; r)$ and $r \in [r_1, r_2]$ yields (134). The second conclusion follows by taking $r_1 = 2\sigma_l$ and $r_2 = \frac{1}{2}\rho_l$ in (134) and recalling (136). \square

We introduce some additional notation before we continue. Below, we let $T_0 = \log 2$. For all l such that $\frac{\rho_l}{\sigma_l} > 4$ (recall that $\frac{\rho_l}{\sigma_l} \to \infty$), we let

$$T_l = \log \sqrt{\frac{\rho_l}{\sigma_l}}.$$

Moreover, we define $f_l: [T_0, T_l] \to \mathbb{R}$ by

$$f_l(t) = E(u_l; B(c_l; e^{-t}\rho_l) \setminus B(c_l; e^t\sigma_l)).$$

The following estimate is motivated by the proofs of [35, Lemma 4.7.3] and [27, Lemma A.4] and is the key to establishing (133).

Proposition 5.12. There exists a constant b > 0 such that

$$f_l(t) \le e^{-3b(t-T_0)} f_l(T_0),$$
 (137)

for all l sufficiently large and $t \in [T_0, T_l]$.

Proof. As the conclusion suggests, we prove the proposition by showing that f_l satisfies some first order differential inequality. To begin, suppose $2\sigma_l \leq r_1 < r_2 \leq \frac{1}{2}\rho_l$. Then by (135) and (136) we see that, for l sufficiently large, the rescaled maps

$$x \mapsto u_l(c_l + r_2 x)$$

satisfy both (122) and (123), with $r = \frac{r_1}{r_2}$. Therefore we apply Proposition 5.8 to deduce that

$$E(u_l; B(c_l; r_2) \setminus B(c_l; r_1)) \le C\left(r_2 \int_{\partial B(c_l; r_2)} |du_l|^3 dS_g + r_1 \int_{\partial B(c_l; r_1)} |du_l|^3 dS_g\right).$$

From this it is not hard to see that there exists some b independent of l such that

$$f'_l(t) \le -3bf_l(t)$$
 for all l sufficiently large and $t \in [T_0, T_l]$.

Integrating from T_0 to t gives (137). \square

At last we come to the proof of Theorem 5.2.

Proof of Theorem 5.2. As remarked in the beginning of this section it suffices to prove (133). Since $\frac{\rho_l}{\sigma_l} \to \infty$ as $l \to \infty$, we may without loss of generality assume that

$$\rho_l > 16\sigma_l$$
 for all l .

Now suppose $4\sigma_l \leq r \leq (\rho_l \sigma_l)^{\frac{1}{2}}$ and define t by the equation

$$e^t \sigma_l = \frac{1}{2} r.$$

Then $t \in [T_0, T_l]$ and moreover $r \leq \frac{1}{2}e^{-t}\rho_l$. Hence by (134) (with $r_2 = \frac{1}{2}e^{-t}\rho_l$ and $r_1 = 2e^t\sigma_l = r$) and (137), we have

$$r^{3} \sup_{\partial B(c_{l};r)} |Du_{l}|^{3} \leq Cf_{l}(t) \leq Ce^{3bT_{0}}e^{-3bt}f_{l}(T_{0})$$

$$\leq Ce^{3bT_{0}} \left(\frac{r}{2\sigma_{l}}\right)^{-3b}f_{l}(T_{0}). \tag{138}$$

Therefore, in terms of polar coordinates around c_l , we deduce that for any $4\sigma_l \leq s \leq r \leq (\rho_l \sigma_l)^{\frac{1}{2}}$ and $\xi, \eta \in S^2$, there holds

$$|u_l(s\xi) - u_l(r\eta)| \le |u_l(s\xi) - u_l(r\xi)| + |u_l(r\xi) - u_l(r\eta)|$$

$$\le \int_{4\sigma_l}^r |Du_l(t\xi)| dt + Cr \sup_{\partial B(c_l;r)} |Du_l|.$$

We use (138) to estimate the first term on the last line by

$$\int_{4\sigma_{l}}^{r} |Du_{l}(t\xi)| dt \leq C(2\sigma_{l})^{b} \Big(\int_{4\sigma_{l}}^{r} t^{-b-1} dt \Big) f_{l}(T_{0})^{\frac{1}{3}}$$

$$\leq C f_{l}(T_{0})^{\frac{1}{3}} \leq C E(u_{l}; A_{l})^{\frac{1}{3}}.$$

Combining this with the previous string of inequalities, we infer that

$$\operatorname{diam}\left(u_{l}\left(B(c_{l};\sqrt{\rho_{l}\sigma_{l}})\setminus B(c_{l};4\sigma_{l})\right)\right) \\ \leq CE\left(u_{l};A_{l}\right)^{\frac{1}{3}} + C \sup_{x\in B\left(c_{l};\frac{1}{2}\rho_{l}\right)\setminus B\left(c_{l};2\sigma_{l}\right)} |x-c_{l}| |Du_{l}(x)|,$$

which tends to zero as $l \to \infty$. A similar argument shows that

diam
$$\left(u_l\left(B(c_l; \frac{1}{4}\rho_l) \setminus B(c_l; \sqrt{\rho_l\sigma_l})\right)\right) \to 0 \text{ as } l \to \infty,$$

and hence we have proven (133). \square

Appendices.

Appendix A. Harmonic Analysis. In this appendix we gather some results from harmonic analysis that are used in §4. We first introduce the function spaces before stating the main estimates.

DEFINITION A.1. The Hardy space $\mathcal{H}^1(\mathbb{R}^n)$ is defined as follows. Consider the class

$$\mathcal{T} = \{ \zeta \in C_c^{\infty}(B(1)) \mid ||D\zeta||_{\infty} \le 1 \}.$$

For $\zeta \in \mathcal{T}$ and $\varepsilon > 0$, we write $\zeta_{\varepsilon}(x) = \varepsilon^{-n} \zeta(\frac{x}{\varepsilon})$. Then, for any $g \in L^1(\mathbb{R}^n)$, we define

$$g^*(x) = \sup_{\zeta \in \mathcal{T}} \sup_{\varepsilon > 0} \left| \int_{\mathbb{R}^n} \zeta_{\varepsilon}(x - y) g(y) dy \right|.$$

By definition, g belongs to \mathcal{H}^1 if g^* belongs to $L^1(\mathbb{R}^n)$, in which case we define $\|g\|_{\mathcal{H}^1} = \|g^*\|_1$.

DEFINITION A.2. A function $f \in L^1_{loc}(\mathbb{R}^n)$ is in $BMO(\mathbb{R}^n)$, which is called the space of functions with bounded mean oscillation, if

$$[f]_{BMO} := \sup_{x \in \mathbb{R}^n, r > 0} \int_{B(x;r)} |f - (f)_{B(x;r)}| dx < \infty.$$

Remark A.3. We make two remarks concerning these definitions.

- (i) The Hardy space is a Banach space and is strictly contained in $L^1(\mathbb{R}^n)$.
- (ii) Note that $[f]_{BMO} = [g]_{BMO}$ when f, g differ by a constant. Thus $[\cdot]_{BMO}$ descends to the quotient of $BMO(\mathbb{R}^n)$ by the constant functions and in fact makes it a Banach space.

As a consequence of a deep result due to Fefferman–Stein [14], which identifies BMO with the dual space of \mathcal{H}^1 , we have the following inequality.

THEOREM A.4 ([14, Theorem 2]). Suppose $f \in L^{\infty}(\mathbb{R}^n)$ and $g \in \mathcal{H}^1(\mathbb{R}^n)$. Then

$$\int_{\mathbb{R}^n} fg dx \le C \|g\|_{\mathcal{H}^1} [f]_{BMO},$$

where C depends only on the dimension n.

The following two propositions can be found in [12] and contain estimates which, together with the Fefferman–Stein theorem above, are key to the regularity results in $\S 4$.

PROPOSITION A.5. Suppose $f \in W^{1,n}(B(2))$ and that $\int_{B(2)} f dx = 0$. Let ζ be a cutoff function which is identically 1 on B(1) and vanishes outside of $B(\frac{3}{2})$. Then $\zeta f \in BMO(\mathbb{R}^n)$ with

$$[\zeta f]_{BMO} \le C \|Df\|_{n;B(2)}.$$

PROPOSITION A.6. Let $f \in W_0^{1,n}(B(1))$ and let $X \in L^{\frac{n}{n-1}}(B(2);\mathbb{R}^n)$ be a vector field which is divergence-free on B(2) in the sense of distributions. Then $X \cdot Df \in \mathcal{H}^1(\mathbb{R}^n)$, and

$$||X \cdot Df||_{\mathcal{H}^1} \le C||Df||_n ||X||_{\frac{n}{n-1}},$$

where C depends only on the dimension n.

Proof. For the sake of completeness we include a proof of Proposition A.6. We would like to estimate the quantity

$$\left| \int_{\mathbb{R}^n} \zeta_{\varepsilon}(x-y) X(y) \cdot Df(y) dy \right|,$$

for any given $\zeta \in \mathcal{T}$ and $\varepsilon > 0$. To do that we need to treat a few different cases depending on where x is and how large ε is.

Case 1. Suppose that $x \in B(\frac{3}{2})$ and $\varepsilon < \frac{1}{2}$. Then $B(x;\varepsilon) \subseteq B(2)$, so for any $\lambda \in \mathbb{R}$, the function

$$y \mapsto (f(y) - \lambda)\zeta_{\varepsilon}(x - y)$$

has compact support in B(2). Since div X vanishes on B(2) in the sense of distributions, this implies

$$0 = \int_{\mathbb{R}^n} X(y) \cdot D\left[(f(y) - \lambda) \zeta_{\varepsilon}(x - y) \right] dy$$
$$= \int_{\mathbb{R}^n} \zeta_{\varepsilon}(x - y) X(y) \cdot Df(y) dy - \int_{\mathbb{R}^n} (f(y) - \lambda) X(y) \cdot D\zeta_{\varepsilon}(x - y) dy.$$

In other words, recalling the definition of ζ_{ε}

$$\int_{\mathbb{R}^n} \zeta_{\varepsilon}(x-y)X(y) \cdot Df(y)dy = \varepsilon^{-1} \int_{B(x;\varepsilon)} D\zeta \left(\varepsilon^{-1}(x-y)\right) \cdot (f(y)-\lambda)X(y)dy. \tag{139}$$

Next we choose an exponent p > n and define

$$q = \frac{p}{p-1}, \qquad p_* = \frac{np}{p+n},$$

so that W^{1,p_*} controls L^p by the Sobolev embedding theorem, and $\frac{1}{p} + \frac{1}{q} = 1$. For later use, we note that $q < \frac{n}{n-1}$ and $p_* < n$. Now we choose $\lambda = \int_{B(x;\varepsilon)} f(y) dy$ and use the Hölder inequality, the fact that $\|D\zeta\|_{\infty} \leq 1$, and the Poincaré inequality to estimate the right hand side of (139) as follows:

$$\left| \varepsilon^{-1} \oint_{B(x;\varepsilon)} D\zeta(\varepsilon^{-1}(x-y)) \cdot (f(y)-\lambda)X(y)dy \right|$$

$$\leq \varepsilon^{-1} \left(\oint_{B(x;\varepsilon)} |X|^q \right)^{\frac{1}{q}} \left(\oint_{B(x;\varepsilon)} |f-\lambda|^p \right)^{\frac{1}{p}}$$

$$\leq \left(\oint_{B(x;\varepsilon)} |X|^q \right)^{\frac{1}{q}} \left(\oint_{B(x;\varepsilon)} |Df|^{p_*} \right)^{\frac{1}{p_*}}$$

$$\leq [M(|X|^q)(x)]^{\frac{1}{q}} [M(|Df|^{p_*})(x)]^{\frac{1}{p_*}},$$

where $M(|X|^q)$ and $M(|Df|^{p_*})$ in the last line denote maximal functions. This means that

$$M(F)(x) = \sup_{\varepsilon > 0} \int_{B(x;\varepsilon)} F.$$

Case 2. Suppose that $x \in B(\frac{3}{2})$ and $\varepsilon > \frac{1}{2}$. We observe that since

$$\int_{\mathbb{R}^n} X(y) \cdot Df(y) dy = 0,$$

we have

$$\int_{\mathbb{R}^n} \zeta_{\varepsilon}(x-y)X(y) \cdot Df(y)dy = \int_{B(1)} (\zeta_{\varepsilon}(x-y) - \zeta_{\varepsilon}(x))X(y) \cdot Df(y)dy,$$

and hence, recalling that $||D\zeta_{\varepsilon}||_{\infty} \leq \varepsilon^{-n-1}$, we deduce that

$$\left| \int_{\mathbb{R}^{n}} \zeta_{\varepsilon}(x - y) X(y) \cdot Df(y) dy \right|$$

$$\leq \varepsilon^{-n-1} \int_{B(1)} |X(y)| |Df(y)| dy \leq 2^{n+1} ||X||_{\frac{n}{n-1}} ||Df||_{n}.$$
(140)

Combining Cases 1 and 2, we get that for $x \in B(\frac{3}{2})$, there holds

$$(X \cdot Df)^*(x) \le \left[M(|X|^q)(x) \right]^{\frac{1}{q}} \left[M(|Df|^{p_*})(x) \right]^{\frac{1}{p_*}} + 2^{n+1} ||X||_{\frac{n}{n-1}} ||Df||_n.$$

Consequently

$$\|(X \cdot Df)^*\|_{1;B(\frac{3}{2})} \le \|M(|X|^q)^{\frac{1}{q}}M(|Df|^{p_*})^{\frac{1}{p_*}}\|_1 + C\|X\|_{\frac{n}{n-1}}\|Df\|_n. \tag{141}$$

To continue, note that the function $|X|^q$ lies in $L^{\alpha}(\mathbb{R}^n)$ for $\alpha = \frac{n}{q(n-1)} > 1$, and that the function $|Df|^{p_*}$ lies in $L^{\beta}(\mathbb{R}^n)$ for $\beta = \frac{n}{p_*} > 1$. Hence

$$||M(|X|^q)||_{\alpha} \le C||X|^q||_{\alpha} = C||X||_{\frac{n}{n-1}}^q,$$

$$||M(|Df|^{p_*})||_{\beta} \le C||Df|^{p_*}||_{\beta} = C||Df||_n^{p_*}.$$

Combining these estimates with the Hölder inequality and the fact that $\frac{1}{\alpha q} + \frac{1}{\beta p_*} = 1$, we get

$$\begin{split} \|M(|X|^q)^{\frac{1}{q}}M(|Df|^{p_*})^{\frac{1}{p_*}}\|_1 &\leq \|M(|X|^q)^{\frac{1}{q}}\|_{q\alpha}\|M(|Df|^{p_*})^{\frac{1}{p_*}}\|_{p_*\beta} \\ &= \|M(|X|^q)\|_{q}^{\frac{1}{q}}\|M(|Df|^{p_*})\|_{\beta}^{\frac{1}{p_*}} \\ &\leq C\|X\|_{\frac{n}{n-1}}\|Df\|_n. \end{split}$$

Putting this back into (141), we get

$$\|(X \cdot Df)^*\|_{1;B(\frac{3}{2})} \le C\|X\|_{\frac{n}{n-1}}\|Df\|_n. \tag{142}$$

Case 3. Suppose that $x \in \mathbb{R}^n \setminus B(\frac{3}{2})$. Then since Df is nonzero only in B(1) and since $B(x;\varepsilon) \cap B(1) = \emptyset$ if $\varepsilon < \frac{1}{3}|x|$, we have in this case that

$$(X \cdot Df)^*(x) = \sup_{\zeta} \sup_{\varepsilon \ge \frac{1}{3}|x|} \varepsilon^{-n} \bigg| \int_{B(1)} \zeta(\varepsilon^{-1}(x-y))X(y) \cdot Df(y)dy \bigg|.$$

Then, the same observation that lead to (140) allows us to conclude that in fact

$$(X \cdot Df)^*(x) \le 3^{n+1}|x|^{-n-1}||X||_{\frac{n}{n-1}}||Df||_n.$$

Integrating this over $\mathbb{R}^n \setminus B(\frac{3}{2})$ gives

$$\|(X \cdot Df)^*\|_{1:\mathbb{R}^n \setminus B(\frac{3}{n})} \le C\|X\|_{\frac{n}{n-1}}\|Df\|_n. \tag{143}$$

We complete the proof upon combining (142) and (143), and recalling the definition of \mathcal{H}^1 . \square

Appendix B. Higher regularity and the mean value inequality. In this appendix we give proofs of Theorems 4.7 and 4.9. The arguments are largely based on the work of Duzaar–Fuchs [10] and Duzaar–Mingione [11] on p-harmonic maps, with some necessary modifications due to the fact that we have a different system of equations, and because we are working with a potentially nonflat metric on the domain.

Proof of Theorem 4.7. We begin with part (a), whose proof actually resembles the proof of Proposition 4.3, but we estimate things slightly differently. Take any $x_0 \in B(1)$ and $r < \frac{1}{8}$, and again introduce the rescalings

$$\tilde{u}(x) = u(x_0 + rx), \qquad \tilde{g}(x) = g(x_0 + rx).$$

Then Corollary 4.6 is applicable to \tilde{u} , so we have (84) and (85) at our disposal. Next, we let g^0 denote the constant metric given by

$$g_{ij}^0(x) := \tilde{g}_{ij}(0) = g_{ij}(x_0) \text{ for all } x \in B(2).$$

Similarly to the proof of Proposition 4.3, we consider the unique function $\tilde{v} \in W^{1,3}(B(1); \mathbb{R}^d)$ which minimizes

$$\int_{B(1)} |dw|_{g^0}^3 d\mu_{g^0}$$

amongst all functions $w \in W^{1,3}(B(1); \mathbb{R}^d)$ with $w - \tilde{u} \in W_0^{1,3}(B(1); \mathbb{R}^d)$. Then \tilde{v} satisfies

$$\int_{B(1)} \langle |d\tilde{v}|_{g^0} d\tilde{v}, d\eta \rangle_{g^0} d\mu_{g^0} = 0 \quad \text{ for all } \eta \in W^{1,3}_0(B(1); \mathbb{R}^d),$$

and

$$\int_{B(1)} |d\tilde{v}|_{g^0}^3 d\mu_{g^0} \le \int_{B(1)} |d\tilde{u}|_{g^0}^3 d\mu_{g^0}. \tag{144}$$

Following [11], we introduce the function $V: \mathbb{R}^{3 \times d} \to \mathbb{R}^{3 \times d}$ defined by

$$V(\xi) = |\xi|^{\frac{1}{2}} \xi.$$

Then by a refinement of Uhlenbeck's work due to Giaquinta–Modica [17, Theorem, 3.1], we know that \tilde{v} has Hölder continuous first derivatives, and furthermore there exist universal constants C and γ such that

$$\int_{B(\lambda)} \left| V(D\tilde{v}) - (V(D\tilde{v}))_{B(\lambda)} \right|^2 dx$$

$$\leq C\lambda^{3+2\gamma} \int_{B(1)} \left| V(D\tilde{v}) - (V(D\tilde{v}))_{B(1)} \right|^2 dx \quad \text{for all } \lambda \in (0, \frac{1}{4}). \tag{145}$$

Also, the argument leading to (72) and (73) is still valid, and for all $x \in B(1)$ and i = 1, ..., d, we get

$$\min_{B(1)} \tilde{u}^i - \tilde{u}^i(x) \le \tilde{v}^i(x) - \tilde{u}^i(x) \le \max_{B(1)} \tilde{u}^i - u^i(x).$$

In particular, by (85), we have

$$|\tilde{v} - \tilde{u}|_{0;B(1)} \le C[\tilde{u}]_{\alpha;B(1)} \le C||D\tilde{u}||_{3;B(2)}.$$

Therefore, using $\tilde{u} - \tilde{v}$ as a test function in (65), we get

$$\int_{B(1)} \langle |d\tilde{u}| d\tilde{u}, d(\tilde{u} - \tilde{v}) \rangle d\mu_g \le C \|D\tilde{u}\|_{3;B(2)}^4$$

$$\le C \|Du\|_{3;B(x_0;2r)}^4 \le C r^{4\alpha} \|Du\|_{3;B(2)}^4,$$

where we used (84) in the last inequality. On the other hand, repeating the calculation in (76) and (77), and using (144), we deduce that

$$\int_{B(1)} \langle |d\tilde{v}|_g d\tilde{v}, d(\tilde{u} - \tilde{v}) \rangle_g d\mu_g \leq C |\tilde{g} - g^0|_{0;B(1)} ||D\tilde{v}||_{3;B(1)}^2 ||D\tilde{u} - D\tilde{v}||_{3;B(1)} \\
\leq C |g - g(x_0)|_{0;B(x_0;r)} ||D\tilde{u}||_{3;B(1)}^3 \\
\leq C r |Dg|_{0;B(2)} r^{3\alpha} ||Du||_{3;B(2)}^3 \\
\leq C \varepsilon_0 r^{1+3\alpha} ||Du||_{3;B(2)}^3.$$

Letting $v(x) = \tilde{v}(r^{-1}(x - x_0))$, the above gives

$$\int_{B(x_0;r)} |V(Du) - V(Dv)|^2 dx = \int_{B(1)} |V(D\tilde{u}) - V(D\tilde{v})|^2 dx$$

$$\leq C \int_{B(1)} \langle |d\tilde{u}|_g d\tilde{u} - |d\tilde{v}|_g d\tilde{v}, d\tilde{u} - d\tilde{v} \rangle d\mu_g$$

$$\leq Cr^{4\alpha} (1 + ||Du||_{3:B(2)}) ||Du||_{3:B(2)}^3.$$

Note that we used [11, equation (11) on page 240] in the second line above.

Scaling \tilde{v} to v in (145) and combining it with estimate just obtained, we find that for all $x_0 \in B(1)$, $r < \frac{1}{8}$, and $\lambda \in (0, \frac{1}{4})$, we have

$$\int_{B(x_0;\lambda r)} \left| V(Du) - (V(Du))_{B(x_0;\lambda r)} \right|^2 dx \le C\lambda^{3+2\gamma} \int_{B(x_0;r)} \left| V(Du) - (V(Du))_{B(x_0;r)} \right|^2 dx + Cr^{4\alpha} \left(1 + \|Du\|_{3;B(2)} \right) \|Du\|_{3;B(2)}^3.$$

Letting $K = C (1 + ||Du||_{3;B(2)}) ||Du||_{3;B(2)}^3$ and

$$\Phi(x_0, t) = \int_{B(x_0; t)} |V(Du) - (V(Du))_{B(x_0; t)}|^2 dx,$$

we see that for all $x_0 \in B(1)$ and $s \le r < \frac{1}{8}$, we have

$$\Phi(x_0, s) \le C \left(\frac{s}{r}\right)^{3+2\gamma} \Phi(x_0, r) + Kr^{4\alpha}.$$

Recalling that $4\alpha > 3$, we obtain by [21, Lemma 3.4] that

$$\Phi(x_0, s) \le C s^{3+2\gamma'}$$

for all $x_0 \in B(1)$ and $s \in (0, \frac{1}{9})$, where $\gamma' = \min\{\gamma, \frac{1}{2}(4\alpha - 3)\}$. Thus $V(Du) \in C^{0,\gamma'}(B(1); \mathbb{R}^{3\times d})$ with $[V(Du)]_{\gamma';B(1)}$ depending on M, J, the embedding $M \to \mathbb{R}^d$, and $\|Du\|_{3:B(2)}$. Then [11, Lemma 3] yields the assertion in part (a).

For (b) we only indicate the main steps. Fix any ball B with compact closure in the set

$$\{x \in B(1) \mid Du(x) \neq 0\} \quad (= B(1) \setminus \operatorname{crit}_u).$$

Via a difference quotient argument (compare with [10, Lemma 2.2]) using (65), we can show that

$$\int_{B} |Du| |D^2u|^2 dx < \infty,$$

so that $u \in W^{2,2}(B)$, because |Du| is bounded away from zero on \overline{B} . This $W^{2,2}$ -regularity then allows us to integrate by parts on the left hand side of (65) to see that u satisfies an elliptic system of equations of the form

$$A_{ij}^{\gamma\lambda}u_{\gamma\lambda}^j + B_{ij}^{\lambda}u_{\lambda}^j = F_i,$$

almost everywhere in B, where $A_{ij}^{\gamma\lambda}$, B_{ij}^{λ} and F_i are rational expressions involving $g, Dg, J \circ u$ and Du and hence all lie in $C^{0,\beta}(B)$ because $u \in C^{1,\beta}(B)$, and because g is smooth by assumption. The regularity theory for $W^{2,2}$ -solutions to elliptic systems of the above type (see [16, Theorem 5.22]) tells us that $u \in C^{2,\beta}_{loc}(B)$, and a bootstrapping argument yields smoothness locally in B. Since B is an arbitrary ball with compact closure in $\{x \in B(1) \mid Du(x) \neq 0\}$, we are done. \square

Proof of Theorem 4.9. Throughout this proof, we use d^* to denote the dual, with respect to g, of the exterior derivative. The operators Δ and ∇ , as well as the volume form μ , are with respect to the metric g, and ∇du refers to the covariant derivative of du as a \mathbb{R}^d -valued 1-form, as opposed to a section of $\mathbb{R}^3 \otimes u^*TM$.

We first require $\varepsilon_1 < \varepsilon_0$, so that by Theorem 4.7 we have $u \in C^{1,\beta}(B(1); M)$ and u is smooth on the set $\{x \in B(1) \mid Du(x) \neq 0\}$, which we denote by U_+ for this proof.

Next, we note that the difference quotient argument in the proof of [10, Lemma 2.2] carries over to our case to give $|du|^{\frac{1}{2}}du \in W_{\text{loc}}^{1,2}(B(1))$. Since $u \in C^1(B(1))$, we deduce that $|du|^3 \in W_{\text{loc}}^{1,2}(B(1))$. Next we show, as in [10, Lemma 2.3], that the function $w := |du|^3$ satisfies a second order, uniformly elliptic equation in U_+ .

The computation in the proof of [10, Lemma 2.3] can then be modified as follows. (See also [49, pages 222–223].) At any point in U_+ we have

```
\begin{split} \Delta w &= \Delta \langle |du|du, du \rangle \\ &= \langle \Delta \left( |du|du \right), du \rangle + 2 \langle \nabla \left( |du|du \right), \nabla du \rangle + \langle |du|du, \Delta du \rangle \\ &= -\langle (d^*d + dd^*) \left( |du|du \right), du \rangle + \text{Ric}(|du|du, du) + 2 \langle \nabla \left( |du|du \right), \nabla du \rangle + \langle |du|du, \Delta du \rangle \\ &= -\langle (d^*d + dd^*) \left( |du|du \right), du \rangle + \text{Ric}(|du|du, du) + \langle \nabla \left( |du|du \right), \nabla du \rangle + \frac{1}{3} \Delta w. \end{split}
```

(Note that we used the Weitzenböck formula for differential 1-forms in going from the second line to the third.) Hence we get

$$\frac{2}{3}\Delta w + \langle d^*d(|du|du), du \rangle
= -\langle dd^*(|du|du), du \rangle + \text{Ric}(|du|du, du) + \langle \nabla(|du|du), \nabla du \rangle
\geq -\langle dd^*(|du|du), du \rangle - K|du|^3 + |du|^{-1}\langle du, \nabla du \rangle^2 + |du||\nabla du|^2
\geq -\langle dd^*(|du|du), du \rangle - K|du|^3 + |du||\nabla du|^2.$$
(146)

To continue, note that we have

$$\langle d^*d(|du|du), du \rangle = -\frac{1}{3}\operatorname{div}(dw - |du|^{-2}\langle du, dw \rangle du),$$

and thus the left hand side of (146) becomes

$$\frac{2}{3}\Delta w + \langle d^*d(|du|du), du \rangle = \frac{1}{3}\operatorname{div}(dw + |du|^{-2}\langle du, dw \rangle du), \tag{147}$$

where the right hand side can be understood as a uniformly elliptic operator of second order acting on w. For the right hand side of (146), the first term can be estimated with the help of (65) as follows:

$$-\langle dd^* \left(|du|du \right), du \rangle = \frac{\sqrt{3}}{2} \langle d\left(\frac{1}{\sqrt{g}} \epsilon^{\alpha\beta\gamma} (J^i_{jk} \circ u)_{\gamma} u^j_{\alpha} u^k_{\beta} \right), du^i \rangle$$

$$\geq -C|du|^4 - C|du|^5 - C|du|^3 |D^2 u|$$

$$\geq -C|du|^3 - C|du|^5 - C|du|^3 |D^2 u|, \tag{148}$$

where we used Young's inequality to estimate $|du|^4 \leq C(|du|^3 + |du|^5)$ in the last inequality above.

Next, by the coordinate expression for ∇du , we have

$$|D^2u| \le C(|\nabla du| + |du|).$$

Using this and Young's inequality again, we bound the last term in (148) as follows.

$$|du|^{3}|D^{2}u| \leq C|du|^{3}(|\nabla du| + |du|)$$

$$\leq C\left(\frac{1}{4\varepsilon}|du|^{5} + \varepsilon|du||\nabla du|^{2}\right) + C\left(\frac{1}{2}|du|^{3} + \frac{1}{2}|du|^{5}\right).$$
(149)

Combining this with (148) and choosing a small enough ε , we arrive at

$$-\langle dd^*(|du|du), du \rangle \ge -C|du|^3 - C|du|^5 - \frac{1}{2}|du||\nabla du|^2.$$

Putting this back into the right hand side of (146) and using (147), we obtain

$$\frac{1}{3}\operatorname{div}(dw + |du|^{-2}\langle du, dw\rangle du) \ge -Cw^{\frac{5}{3}} - Cw,\tag{150}$$

pointwise everywhere on U_+ . Hence, for all $\zeta \in C_c^1(U_+)$ with $\zeta \geq 0$, we have

$$\int_{U_{\perp}} \left(\langle dw, d\zeta \rangle + |du|^{-2} \langle du, dw \rangle \langle du, d\zeta \rangle \right) d\mu \leq C \int_{U_{\perp}} (w^{\frac{5}{3}} + w) \zeta d\mu.$$

Now the proof of [10, Lemma 2.4] carries over to show that the above inequality actually holds for all $\zeta \in C_0^1(B(1))$ with $\zeta \geq 0$. Consequently $w \in W_{\text{loc}}^{1,2}(B(1))$ is a weak solution to (150) on all of B(1). We may then follow [10, pages 394–395] to complete the proof, shrinking ε_1 if necessary. \square

REFERENCES

- F. BETHUEL, On the singular set of stationary harmonic maps, Manuscripta Math., 78 (1993), pp. 417–443. MR1208652
- [2] H. Brezis and J.-M. Coron, Multiple solutions of H-systems and Rellich's conjecture, Comm. Pure Appl. Math., 37 (1984), pp. 149–187. MR0733715
- [3] R. B. Brown and A. Gray, Vector cross products, Comment. Math. Helv., 42 (1967), pp. 222– 236. MR0222105
- [4] R. BRYANT AND R. HARVEY, Submanifolds in hyper-Kähler geometry, J. Amer. Math. Soc., 2 (1989), pp. 1–31. MR0953169
- [5] R. L. BRYANT AND S. M. SALAMON, On the construction of some complete metrics with exceptional holonomy, Duke Math. J., 58 (1989), pp. 829–850. MR1016448
- [6] A. DOAN AND T. WALPUSKI, On counting associative submanifolds and Seiberg-Witten monopoles, Pure Appl. Math. Q., 15 (2019), pp. 1047-1133. MR4085667
- [7] S. K. DONALDSON AND P. B. KRONHEIMER, The geometry of four-manifolds, Oxford Mathematical Monographs, The Clarendon Press, Oxford University Press, New York, 1990. MR1079726
- [8] S. DONALDSON AND E. SEGAL, Gauge theory in higher dimensions, II, in "Surveys in differential geometry. Volume XVI. Geometry of special holonomy and related topics", pp. 1–41, Surv. Differ. Geom., 16, Int. Press, Somerville, MA. MR2893675
- [9] S. K. DONALDSON AND R. P. THOMAS, Gauge theory in higher dimensions, in "The geometric universe (Oxford, 1996)", pp. 31–47, Oxford Univ. Press, Oxford. MR1634503
- [10] F. DUZAAR AND M. FUCHS, On removable singularities of p-harmonic maps, Ann. Inst. H. Poincaré Anal. Non Linéaire, 7 (1990), pp. 385–405. MR1138529
- [11] F. DUZAAR AND G. MINGIONE, The p-harmonic approximation and the regularity of p-harmonic maps, Calc. Var. Partial Differential Equations, 20 (2004), pp. 235–256. MR2062943
- [12] L. C. Evans, Partial regularity for stationary harmonic maps into spheres, Arch. Rational Mech. Anal., 116 (1991), pp. 101–113. MR1143435
- [13] P. M. N. FEEHAN, Geometry of the ends of the moduli space of anti-self-dual connections, J. Differential Geom., 42 (1995), pp. 465-553. MR1367401
- [14] C. Fefferman and E. M. Stein, H^p spaces of several variables, Acta Math., 129 (1972), pp. 137–193. MR0447953
- [15] M. Fuchs, The blow-up of p-harmonic maps, Manuscripta Math., 81 (1993), pp. 89–94. MR1247590
- [16] M. GIAQUINTA AND L. MARTINAZZI, An introduction to the regularity theory for elliptic systems, harmonic maps and minimal graphs, second edition, Appunti. Scuola Normale Superiore di Pisa (Nuova Serie), 11, Edizioni della Normale, Pisa, 2012. MR3099262
- [17] M. GIAQUINTA AND G. MODICA, Remarks on the regularity of the minimizers of certain degenerate functionals, Manuscripta Math., 57 (1986), pp. 55–99. MR0866406
- [18] D. GILBARG AND N. S. TRUDINGER, Elliptic partial differential equations of second order, reprint of the 1998 edition, Classics in Mathematics, Springer-Verlag, Berlin, 2001. MR1814364
- [19] A. Gray, Vector cross products, harmonic maps and the Cauchy-Riemann equations, in "Harmonic maps (New Orleans, La., 1980)", pp. 57–74, Lecture Notes in Math., 949, Springer, Berlin. MR0673583
- [20] M. GROMOV, Pseudo holomorphic curves in symplectic manifolds, Invent. Math., 82 (1985), pp. 307–347. MR0809718
- [21] Q. HAN AND F. LIN, Elliptic partial differential equations, second edition, Courant Lecture Notes in Mathematics, 1, Courant Institute of Mathematical Sciences, New York, 2011. MR2777537
- [22] R. HARDT AND F.-H. LIN, Mappings minimizing the L^p norm of the gradient, Comm. Pure Appl. Math., 40 (1987), pp. 555–588. MR0896767
- [23] R. HARVEY AND H. B. LAWSON, JR., Calibrated geometries, Acta Math., 148 (1982), pp. 47–157. MR0666108
- [24] A. HAYDYS, G₂ instantons and the Seiberg-Witten monopoles, in "Gromov-Witten theory, gauge theory and dualities", 12 pp, Proc. Centre Math. Appl. Austral. Nat. Univ., 48, Austral. Nat. Univ., Canberra. MR3951401
- [25] A. HAYDYS AND T. WALPUSKI, A compactness theorem for the Seiberg-Witten equation with multiple spinors in dimension three, Geom. Funct. Anal., 25 (2015), pp. 1799–1821. MR3432158
- [26] F. HÉLEIN, Régularité des applications faiblement harmoniques entre une surface et une variété riemannienne, C. R. Acad. Sci. Paris Sér. I Math., 312 (1991), pp. 591–596. MR1101039

- [27] H. HOFER AND D. A. SALAMON, Floer homology and Novikov rings, in "The Floer memorial volume", pp. 483–524, Progr. Math., 133, Birkhäuser, Basel. MR1362838
- [28] D. JOYCE, Compact manifolds with special holonomy, Oxford Mathematical Monographs, Oxford University Press, Oxford, 2000. MR1787733
- [29] D. JOYCE, Conjectures on counting associative 3-folds in G₂-manifolds, in "Modern geometry: a celebration of the work of Simon Donaldson", pp. 97–160, Proc. Sympos. Pure Math., 99, Amer. Math. Soc., Providence, RI. MR3838881
- [30] S. Karigiannis, Deformations of G_2 and Spin(7) structures, Canad. J. Math., 57 (2005), pp. 1012–1055. MR2164593
- [31] J.-H. LEE AND N. C. LEUNG, Instantons and branes in manifolds with vector cross products, Asian J. Math., 12 (2008), pp. 121–143. MR2415016
- [32] N. C. LEUNG, X. WANG AND K. ZHU, Thin instantons in G₂-manifolds and Seiberg-Witten invariants, J. Differential Geom., 95 (2013), pp. 419–481. MR3128991
- [33] N. C. LEUNG, X. WANG AND K. ZHU, Instantons in G₂ manifolds from J-holomorphic curves in coassociative submanifolds, in "Proceedings of the Gökova Geometry-Topology Conference 2012", pp. 89–110, Int. Press, Somerville, MA. MR3203358
- [34] F.-H. Lin, Gradient estimates and blow-up analysis for stationary harmonic maps, Ann. of Math. (2), 149 (1999), pp. 785–829. MR1709303
- [35] D. McDuff and D. Salamon, J-holomorphic curves and symplectic topology, second edition, American Mathematical Society Colloquium Publications, 52, American Mathematical Society, Providence, RI, 2012. MR2954391
- [36] L. Mou and P. Yang, Regularity for n-harmonic maps, J. Geom. Anal., 6 (1996), pp. 91–112. MR1402388
- [37] L. MOU AND C. Y. WANG, Bubbling phenomena of Palais-Smale-like sequences of m-harmonic type systems, Calc. Var., 4 (1996), pp. 341–367. MR1393269
- [38] A. Naber, D. Valtorta and G. Veronelli, Quantitative regularity for p-harmonic maps, Comm. Anal. Geom., 27 (2019), pp. 111–159. MR3951022
- [39] T. H. PARKER, Bubble tree convergence for harmonic maps, J. Differential Geom., 44 (1996), pp. 595–633. MR1431008
- [40] T. H. PARKER AND J. G. WOLFSON, Pseudo-holomorphic maps and bubble trees, J. Geom. Anal., 3 (1993), pp. 63–98. MR1197017
- [41] T. RIVIÈRE, Interpolation spaces and energy quantization for Yang-Mills fields, Comm. Anal. Geom., 10 (2002), pp. 683–708. MR1925499
- [42] J. SACKS AND K. UHLENBECK, The existence of minimal immersions of 2-spheres, Ann. of Math. (2), 113 (1981), pp. 1–24. MR0604040
- [43] R. SCHOEN, Analytic aspects of the harmonic map problem, in "Seminar on nonlinear partial differential equations (Berkeley, Calif., 1983)", pp. 321–358, Math. Sci. Res. Inst. Publ., 2, Springer, New York. MR0765241
- [44] L. Simon, Lectures on geometric measure theory, Proceedings of the Centre for Mathematical Analysis, Australian National University, 3, Australian National University, Centre for Mathematical Analysis, Canberra, 1983. MR0756417
- [45] A. M. SMITH, A theory of multiholomorphic maps, https://arxiv.org/abs/1112.1471.
- [46] H. Takeuchi, Some conformal properties of p-harmonic maps and a regularity for spherevalued p-harmonic maps, J. Math. Soc. Japan, 46 (1994), pp. 217–234. MR1264939
- [47] G. Tian, Gauge theory and calibrated geometry I, Ann. of Math. (2), 151 (2000), pp. 193–268. MR1745014
- [48] T. TORO AND C. WANG, Compactness properties of weakly p-harmonic maps into homogeneous spaces, Indiana Univ. Math. J., 44 (1995), pp. 87–113. MR1336433
- [49] K. Uhlenbeck, Regularity for a class of non-linear elliptic systems, Acta Math., 138 (1977), pp. 219–240. MR0474389
- [50] C. Wang, Regularity and blow-up analysis for J-holomorphic maps, Commun. Contemp. Math., 5 (2003), pp. 671–704. MR2003213
- [51] H. C. Wente, An existence theorem for surfaces of constant mean curvature, J. Math. Anal. Appl., 26 (1969). MR0243467
- [52] B. WHITE, Infima of energy functionals in homotopy classes of mappings, J. Differential Geom., 23 (1986), pp. 127–142. MR0845702
- [53] R. Ye, Gromov's compactness theorem for pseudo holomorphic curves, Trans. Amer. Math. Soc., 342 (1994), pp. 671–694. MR1176088